DYNAMICS OF TRACE METALS, RARE EARTH ELEMENTS AND NUTRIENTS IN COCHIN ESTUARY, SOUTH WEST COAST OF INDIA

Thesis submitted to Cochin University of Science and Technology in Partial Fulfillment of the Requirements for the Degree of



DOCTOR OF PHILOSOPHY

In

MARINE CHEMISTRY UNDER THE FACULTY OF MARINE SCIENCES

Bу

LALLU K. R (Reg. No. 4113)



NATIONAL INSTITUTE OF OCEANOGRAPHY (COUNCIL OF SCIENTIFIC AND INDUSTRIAL RESEARCH) Regional Centre, Kochi - 682018, INDIA JUNE, 2017



This is to certify that the thesis entitled "DYNAMICS OF TRACE METALS, RARE EARTH ELEMENTS AND NUTRIENTS IN COCHIN ESTUARY, SOUTH WEST COAST OF INDIA" is an authentic record of the research carried out by Mrs. LALLU K R (Reg: No: 4113), under my supervision in CSIR-National Institute of Oceanography, Regional centre, Cochin- 18, in partial fulfillment of the requirement for the degree of Doctor of Philosophy in Faculty of Marine Sciences, Cochin University of science and Technology and that no part of this thesis has previously formed the basis for the award of any other degree, diploma or associateship in any university.

I also certify that all the relevant corrections and modifications suggested by the audience during the pre-synopsis seminar and recommended by the doctoral committee of the candidate have been incorporated in this thesis.

Kochi -18 June, 2017

Dr. P. S. Parameswaran (Supervising Guide)

<u>DECLARATION</u>

I hereby declare that the thesis entitled " DYNAMICS OF TRACE METALS, RARE EARTH ELEMENTS AND NUTRIENTS IN COCHIN ESTUARY, SOUTH WEST COAST OF INDIA" submitted by me is an authentic record of the research carried out by me, under the supervision of *Dr. P. S. Parameswaran*, Chief Scientist, CSIR - National Institute of Oceanography, Regional Centre, Kochi- 18, in partial fulfillment of the requirement for the degree of Doctor of Philosophy in Faculty of Marine Sciences, Cochin University of science and Technology and that no part of this thesis has been presented before for any other degree diploma or associateship in any university.

Kochi-18 June, 2017 **LALLUKR** (Reg. No: 4113) The work presented in this thesis would not have been possible without my close association with many people who were always there when I needed them the most. I take this opportunity to acknowledge them and extend my sincere gratitude for helping me make this Ph.D. thesis a possibility.

First and foremost, I would like to express my heartfelt gratitude to my supervising guide Dr. P. S. Parameswaran, Chief Scientist (retired), CSIR-National Institute of Oceanography, for accepting me as his research student. I gratefully acknowledge his valuable guidance, intellectual inputs, constant support and encouragement and freedom of work extended to me throughout the period of my work.

I express my sincere thanks to Dr. V. S. N. Murthy, Acting Director, CSIR-National Institute of Oceanography, Goa and Dr T. Pankajakshan, Scientist in charge, CSIR-NIO, RC- Kochi, for providing me with the necessary facilities to carry out my research work.

I wish to pronounce my gratitude to Dr. K, K, Balachanndran, Senior technical officer, for his constant guidance, support and encouragement, and I am indebted to him for offering expertise, understanding and untiring efforts during the course of my study.

I wish to express my gratitude to Dr. B. N. Nath, Chief Scientist, NIO, Goa, Dr. C. Revichandran, Chief Scientist, Dr. Maheswari Nair, Senior technical officer and Dr. Gireeshkumar T. R, Scientist, NIO, RC- Kochi for the valuable guidance, advice, motivation and blessings extended towards me.

My special thanks to Mr. Muraleedharan K R for the help rendered in field study and sample collection. I am thankful to Dr. G. Parthiban, Senior Technical Officer, NIO, Goa for ICP-MS analysis.

I am thankful to Dr. S. Muraleedharan Nair, HOD, Department of Chemical Oceanography, School of Marine Sciences, Dr. C. H Sujatha, Department of Chemical Oceanography, School of Marine Sciences and Dr. A. N. Balchand, Dean & Head of the Department, Department of Physical Oceanography, School of Marine Sciences, CUSAT for their valuable suggestions in the Doctoral and Research committee meetings.

I also extend my sincere thanks to Dr. C. Revichandran, Dr. P. K, Dineshkumar, Dr. T. V. Raveendran, Dr. R .Jyothibabu, Dr. Parvathi A, Dr. Anas A. and Dr. K, K, Balachandran, Research committee members of CSIR-NIO, RC, Kochi, for their constructive suggestions, guidance and constant encouragement during the course of my work.

My special thanks to Dr. Shyju. P and Mr. Shameem, K for ICP-MS analysis.

Words cannot express my feelings and bonding I share with my dear friends of "Chemistry lab" for always being there and bearing with me the good and bad times during my wonderful days of Ph.D. I take this opportunity to thank one and all Research fellows of NIO, Regional centre, Kochi who has actively participated in the simultaneous time series sampling in Cochin backwaters without whose help and co-operation, would have been difficult for me to complete my work. I owe my gratitude to my dear friends Mr. Vipindas P. V, Mr. Udayakrishnan P. B, Mr. Simson Sunny C, Mr. Sebin John, Mr. Sachin Mathew, Mr. Srijith B, MrArun P.V, Mr. Deepak M.P, Mr. Ullas, Mr. Naveen Kumar. K, R, Ms. Dayana Mathew, Ms. Vinitha. P. V, Ms. Anjusha A, Ms Rehitha. T. V for the encouragement and support I received through my work. My special thanks to friends at NIO, Goa, Kazib, Pavan Miriyala, Arindam Sarkar, Prachi, Subina Narayan and Treasa for their help and support at NIO, Goa.

I would like to acknowledge all the teachers I learnt from since my childhood, I would not have been here without their guidance, blessing and support.

Words are not enough to express my gratitude to my beloved parents and my dear sister, brother whose unconditional love, caring and shower of blessings have always motivated and guided me in all my endeavours and I am deeply indebted to them for all my achievements in life.

My husband, Mr. Rakesh, has always stood by me during the entire period of my work and without his optimistic support and continued encouragement; my work would have not been successfully completed.

Last but not the least; I thank GOD, Almighty, for being with me unfailingly, for his constant grace, mercy, love and umpteen blessings rendered through various hands during the course of my work.

Lallu K R



Dedicated to My Dear Father, whose advice urged me to grasp his dream of my Doctoralship

PREFACE

Estuarine and coastal areas are complex and dynamic aquatic environment. Physical, chemical, geological and biological interactions between freshwater and salt water systems can have profound influence on the transport and fate of materials (nutrients, trace metals and rare earth elements) in the estuary. The distribution of nutrients and elements through complex processes of material exchange within these aquatic environments are also influenced by anthropogenic inputs. Now-a-days, such influences are more apparent in estuaries and surrounding coastal areas, than in open ocean, due to intense human activities.

Over the past several decades, there has been rapid increase in human settlement along coastal regions / estuaries of the world. Consequently, many estuarine systems have been affected by the serious environmental deterioration and increased encroachment. Estuaries serve as important interfaces between rivers and the coastal sea, where inputs of terrestrial materials can be significantly modified and recycled before entering the coastal zone. A greater knowledge of the biogeochemical cycling of metals and nutrients in estuaries, such as transport, fate and transformation of chemical substances, is important to understand the intensity of the perturbations in these regions.

Understanding of the input/export fluxes of nutrients and trace metals on coastal seas requires monitoring of their inputs through rivers and measuring their transport across the inlet to the coastal seas. However, while it is relatively straightforward to monitor a river's material flux to its receiving estuary, estimation of the subsequent material flux to the coastal zone is much more difficult, owing to the complex biogeochemical processes involved and varied intensity. Freshwater input in to a tidal system can bring about substantial changes in the water quality (salinity, nutrients, trace metals and plankton) even on short time scales. Temporal (intertidal and intra-tidal), spatial and seasonal variations in nutrients and trace metals are found to be appropriate to explain the fate and transport of nutrients and trace metals in an ecosystem. Moreover, fractionation (physical and chemical) of sediments for their different heavy metal species will be an appropriate step to estimate their dynamics in the estuarine and coastal region. Likewise, changes in salinity, freshwater input, currents and tides are very useful to explain the physical behavior and transport properties of these materials in an estuary. Redox potential can successfully provide the oxidizing or reducing conditions prevailing in the estuary. The mineralogy and texture of the estuary adds to the geochemical processes in the sediments of both estuarine and coastal areas.

Cochin estuary is one of the largest tropical estuaries along the southwest coast of India sustaining rich bio-resources. There are several studies delineating the environmental changes in the Cochin estuary caused by anthropogenic activities. However, no studies have so far accomplished quantification of material transport across the estuary. This is the first attempt to quantify the fluxes of nutrients and heavy metals across the Cochin estuary.

Scheme of Presentation

Following an introductory chapter, the thesis is divided into the following six chapters.

Chapter II. Hydrography and nutrient fluxes

Chapter III. Fractionation of trace metals in sediment

Chapter IV. Fractionation of rare earths in sediment

Chapter V. Input and export fluxes of trace metals in the Cochin Estuary

Chapter VI. Summary and Conclusion.

Chapter I provides information on the importance of fractionation and flux studies in coastal marine regions. It also elaborates on the objectives and scope of the study. Chapter 2 deals with the dynamics and fluxes of nutrients through Cochin estuary. It also explains the inter-tidal, intra-tidal and seasonal changes in nutrients at different locations in the Cochin estuary.

Chapter 3 discusses the fractionation of heavy metals in exchangeable, carbonate bound, reducible, organic bound and residual phases and the metal availability to biota. Chapter 4 covers the fractionation of rare earth elements and its depositional characteristics.

Chapter 5 explains the partitioning of heavy metals between dissolved and particulate phases in from 6 major rivers and Cochin estuary and it also discusses the heavy metal fluxes through rivers into Cochin estuary and the net transport to the adjacent coastal sea. Chapter 6 provides the summary and conclusion of the thesis.

1. Introd	fuction1		
1.1	Introduction		
1.2	Aim and Scope of the study		
1.3	Prime Objectives		
2. Hydro	ography and nutrient fluxes		
2.1	Introduction		
2.2	Nutrient fluxes and its importance 13		
2.3	Study region and Methodology13		
	2.3.1 Regional setting 13		
	2.3.2 Data collection and analysis 17		
	2.3.3 Nutrient flux computation 20		
2.4	General hydrography 21		
	2.4.1 Salinity-Intra-tidal, diurnal and seasonal variations 21		
	2.4.2 Dissolved Oxygen		
	2.4.3 Nitrate-N		
	2.4.4 Nitrite-N		
	2.4.5 Ammonia-N		
	2.4.6 Silicate-Si		
	2.4.7 Phosphate-P		
2.5	Nutrient Fluxes		
	2.5.1 Nutrient fluxes- a comparison with other		
	estuaries55		
2.6	Summary and Conclusion		

<u>CONTENTS</u>

3. Fractionation of trace metals

in	sediment	63
3.	.1 Introduction	63
3.	.2 Sampling Strategy	68

3.3	Analy	tical Methodology		
	3.3.1	General sediment characteristics		
	3.3.2	Trace meta	l analysis 70	
	3.3.3	Metal fract	ionation71	
	3.3.4	Clay miner	alogy	
	3.3.5.	Data proce	ssing74	
		3.3.5.1	Normalisation factor74	
		3.3.5.2	Risk Assessment code 74	
		3.3.5.3	Canadian sediment quality guidelines	
			for the protection of aquatic life75	
3.4	Resul	ts		
	3.4.1	General ch	aracteristics of sediment	
		in the study	y domain 76	
		3.4.1.1	Sediment texture	
		3.4.1.2	Clay minerals 77	
		3.4.1.3	Sediment organic carbon 79	
	3.4.2	Distributio	n of trace metals and transition elements 80	
	3.4.3	Sediment Quality Guidelines and evaluation		
		of pollution criteria		
	3.4.4	Fractionati	on of metals	
		3.4.4.1	Zinc	
		3.4.4.2	Chromium	
		3.4.4.3	Cadmium	
		3.4.4.4	Lead	
		3.4.4.5	Nickel 90	
		3.4.4.6	Copper	
		3.4.4.7	Manganese	
		3.4.4.8	Vanadium	
		3.4.4.9	Tin	

5. Input and export fluxes of trace metals			
in the Cochin Es	stuary.		
5.1 Introductio	5.1 Introduction 1		
5.2 Study regio	5.2 Study region and Methodology		
5.2.1 Regi	ional S	etting 167	
5.2.2 Sam	pling a	and analysis170	
5.3 Results and	l Discu	ussion 172	
5.3.1 Gene	eral Hy	ydrography172	
5.3.2 Susp	ended	particulate matter	
5.	.3.2.1	Suspended particulate matter in the	
		upper and lower reaches of rivers 174	
5.	3.2.2	Suspended particulate matter	
		at the Cochin inlet 174	
5.3.4 Distr	ibution	n of trace metals 175	
5.	3.4.1	Concentration of trace metals	
		in the six rivers 175	
5.	.3.4.2	Distribution of trace metals in the lower	
		reaches of rivers 182	
5.	.3.4.3	Distribution and variation of trace metals	
		at the Cochin inlet 185	
5.	3.4.4	Discussion189	
5.3.5 Inpu	t-expo	rt fluxes of trace metals 196	
5.	.3.5.1	Input Fluxes 196	
5.	.3.5.2	Residual fluxes at the Cochin inlet 203	
5.	.3.5.3	Net Fluxes of trace metals 215	
5.4 Summary a	and Co	nclusion	
6. Summary and com	ıclusio	n 224	
References			
Appendix			
Publications			

<u>Abbreviations</u>

ADP	-	Acoustic Doppler Profiler
ADCP	-	Acoustic Doppler Current Profiler
APDC	-	Ammonium Pyrrolidine
		dithiocarbamate
Al	-	Aluminium
BIS	-	Bureau of Indian Standard
CE	-	Cochin Estuary
CH ₃ COOH	-	Acetic acid
CCME	-	Canadian Council of Ministers of the
		Environment
Cr	-	Chromium
Cd	-	Cadmium
Co	-	Cobalt
Cu	-	Copper
Ce	-	Cerium
CeO ₂	-	Cerium dioxide
C/g	-	Carbon per gram
cm	-	Centimeter
CTD	-	Conductivity Temperature Depth
		Profiler
DO	-	Dissolved oxygen
DIN	-	Dissolved Inorganic Nitrogen
Dy	-	Dysprosium
DDDC	-	Diethyl ammonium
		diethyl dithiocarbamate
E	-	East
Eu	-	Europium

Er	-	Erbium
Fe	-	Iron
Fr	-	Fraction
Gd	-	Gadolinium
HC1	-	Hydrochloric acid
H_2O_2	-	Hydrogen peroxide
HNO ₃	-	Nitric acid
HF	-	Hydrofluoric acid
HClO ₄	-	Perchloric acid
Hf	-	Hafnium
HREE	-	Heavy Rare Earth Elements
Но	-	Holmium
Hg	-	Mercury
hr	-	Hour
HHW	-	Highest High Water
IUPAC	-	International Union of Pure and
		Applied Chemistry
ICP-MS	-	Inductively Coupled Plasma-Mass
		Spectrometer
ICP-OES	-	Inductively Coupled Plasma Optical
		Emission Spectrometer
kg	-	Kilogram
kg/day	-	Kilogram per day
kg/m ³	-	Kilogram per meter cube
km	-	Kilometer
KC	-	Kerala Coast
km ²	-	Kilometre square
LREE	-	Light Rare Earth Elements
La	-	Lanthanum

Lu	-	Lutetium
MREE	-	Middle Rare Earth Elements
ml	-	Millilitre
m/kg	-	Milligram per kilogram
mg/g	-	Milligram per gram
m^2	-	Meter square
m ³	-	Meter cube
mm	-	Millimeter
m ³ /s	-	Meter cube per second
m/s	-	Meter per second
$m^{3}s^{-1}$	-	Meter cube per second
$m^3 d^{-1}$	-	Meter cube per day
mg	-	Milligram
mg/m ³	-	Milligram per meter cube
Mg	-	Magnesium
mgC/g	-	Milligram carbon per gram
mg/l	-	Milligram per litre
ml/l	-	Millilitre per litre
Mn	-	Manganese
NF	-	Normalisation Factor
Ν	-	Nitrogen
NH ₄ -N	-	Ammonia- nitrogen
Ν	-	North
Ni	-	Nickel
Nb	-	Niobium
Nd	-	Neodymium
nm	-	Nanometer
NO ₂ ⁻	-	Nitrite
NO_3^-	-	Nitrate

°C	-	Degree Celsius
OC	-	Organic Carbon
OMZ	-	Oxygen Minimum Zone
OH	-	Hydroxide
Pb	-	Lead
Pr	-	Praseodymium
Р	-	Phosphorus
PO4 ³⁻	-	Phosphate
PELs	-	Probable Effect Levels
ppm	-	Parts per million
REE	-	Rare Earth Elements
rpm	-	Rotation per minute
RAC	-	Risk Assessment Code
Si	-	Silicon
SiO4 ⁴⁻	-	Silicate
Sn	-	Tin
SQG	-	Standard Sediment Quality
		Guidelines
SEP	-	Sequential Extraction Procedure
Sc	-	Scandium
Sm	-	Samarium
SPM	-	Suspended Particulate Matter
S	-	South
TOM	-	Total Organic Matter
TEL	-	Threshold Effect Levels
Tb	-	Terbium
Tm	-	Thulium
V	-	Vanadium
W	-	West

WHO	-	World Health Organization
WRSPM	-	World Rivers Average
		Suspended Particulate Matter
wt.	-	Weight
XRD	-	X -ray Diffraction
Yb	-	Ytterbium
Y	-	Yttrium
Zn	-	Zinc
Zr	-	Zirconium
µg/g	-	Micro gram per gram
µg/l	-	Micro gram per litre
µg/kg	-	Micro gram per kilogram
μl	-	Micro litre
μm	-	Micro meter
µmol/g	-	Micro mole per gram
µmol/l	-	Micro mole per litre
β	-	Beta
r	-	Gamma
μ	-	Micron
%	-	Percentage

Lallu, K. R. Fausia, K. H., Vinita, J., Balachandran, K. K., Naveen Kumar, K. R., Rehitha, T. V., 2014. Transport of dissolved nutrients and chlorophyll *a* in a tropical estuary, south west coast of India. *Environ Monit Assess.* 186, 4829-4839.

Vinita, J., Lallu, K. R., Revichandran, C., Muraleedharan, K R., Jineesh, v. K., Shivaprasad, A., 2015. Residual fluxes of water and nutrient transport through the main inlet of a tropical estuary, Cochin estuary, west coast, India. *Environ Monit Assess.* 187, 675.

Jyothibabu, R., Jagadeesan, L., Lallu, K. R., 2015. Copepod carcasses in a tropical estuary during different hydrographical settings. *Environ Monit Assess*. 188, 559.

Vipindas, P. V., Anas, A., Jasmin, C., **Lallu, K. R.**, Fausia, K. H., Balachandran, K. K., Muraleedharan, K. R., Nair, S., 2015. Bacterial domination over Archaea in ammonia oxidation in a monsoon-driven tropical estuary. *Microb. Ecol.* 69, 544-553.

Arya, P. M., Jyothibabu, R., Jagadeesan, L., **Lallu, K. R**., Karnan, C., 2016. Summer monsoon onset-induced changes of autotrophic pico-and nanoplankton in the largest monsoonal estuary along the west coast of India. *Environ Monit Assess.* 188, 93.

Parvathi, A., Jasna, V., Jina, S., Jayalakshmy, K. V., **Lallu, K. R.**, Madhu, N. V., Muraleedharan, K. R., Naveen Kumar, K. R., Balachandran, K. K., 2015. Effects of hydrography on the distribution of bacteria and virus in Cochin estuary, India. *Ecol. Res.* 30, 85-92.

Sooria, P. M., Jyothibabu, R., Anjusha, A., Vineetha, G., Vinita, J., **Lallu**, **K. R**., Paul, M., Jagadeesan, L., 2015. Plankton food web and its seasonal dynamics in a large monsoonal estuary (Cochin backwaters, India) - significance of mesohaline region. *Environ Monit Assess*. 187, 427.

Madhu, N.V., Ullas, N., Ashwini, R., Meenu Paul, Rehitha, T. V. Lallu, K. R., 2014. Characterization of phytoplankton pigments and functional community structure in the Gulf of Mannar and the palk bay using HPLC-CHEMTAX analysis. *Cont Shelf Res.* 80. 79-90.

Jyothibabu, K., Arya P. Mohan., Jagadeesan, L.,Anjusha, A.,Muraleedharan,K R., **Lallu K R**, Kiran Krishna, Ullas, N., 2013. Ecology and trophic preference of picoplankton and nanoplankton in the Gulf of Mannar and the Palk Bay, Southeast coast of India. *J Mar Syst.* 111-112, 29-44.

****\$\$****

Chapter] **GENERAL INTRODUCTION**

- **1.1 Introduction**
- 1.2 Aim and Scope of the study 1.3 Prime Objectives

1.1 **INTRODUCTION**

An estuary is a partially enclosed coastal body of brackish waters, with one or more rivers or streams flowing into it and having a free connection to the open sea. Estuaries serve as important interfaces between rivers and the coastal ocean, where inputs of terrestrial materials get significantly modified and recycled before entering the coastal waters. The dominant feature controlling the distribution, speciation and reactivity of chemical components within estuaries is the mixing of fresh and saline waters. Coastal waters form a multi-dimensional system wherein the dynamic processes are rarely in equilibrium. Unlike the open ocean where nutrients are mainly supplied from subsurface waters, estuarine and coastal systems receive nutrient load from river and land run-off, atmospheric deposition and benthic remineralization. This makes coastal systems highly productive as compared to open oceans. Although the ocean marginal area comprising of estuaries, coastal and continental shelves occupy only about 10% of the global oceanic area, between 30% and 50% of the marine primary production and almost 90% of



the fishery production are reported from these regions (Romankevich, 1984; Walsh, 1988, 1991).



The exponential development of human activities during the last century has caused a negative impact on all environmental compartments,

including aquatic environments. However, the impact of these changes in the estuarine system on the coastal waters, are still not fully known. Nutrient enrichment (eutrophication) is perhaps the



most widespread problem in estuaries around the world (Howarth et al., 2000, 2002).

Nutrients, especially nitrogen, have increased in rivers and estuaries resulting in harmful algal blooms with consequent reduction in dissolved oxygen in those waters. Understanding the behavior of nutrients and hydrographic conditions in estuaries has important implications for global nutrient budgets and for controlling eutrophication of these systems (Pereira et al., 2001; Shen et al., 2003). Estuarine systems are capable of significantly modifying the flux of nutrients from land to sea (Falco et al., 2010). The effectiveness of the estuarine retention/transport depends at least in part on the geography of the estuary (Jickells et al., 2014) and internal nutrient cycling (Longphuirt et al., 2015).

Industrially important trace metals and their compounds find a variety of uses in different fields of human endeavour, including agriculture, engineering, electronic equipments, health, hygiene, etc. The increasing use of trace metals in everyday life is generally viewed with great concern as it has been established that most of these trace metals are potentially toxic to aquatic organisms.

Metals are introduced into the water column under different forms: particulate, colloidal and dissolved (Kennish, 2002). Upon entering the environment, many of these pollutants are subjected to a variety of physical, chemical, geological and biological processes, bringing about their disintegration or sometimes, even ultimate removal. Persistent chemicals that do not breakdown under above conditions remain in the water columns, leading to serious environmental and human health problems (Shibu, 1992).

In estuaries, trace metals generally exist in two phases: *i.e.*, in the dissolved phase in the water column and also in the particulate phase adsorbed on to sediments. The behaviour of trace metals in the aquatic environment is strongly influenced by adsorption to organic and inorganic particles. The dissolved fraction of the trace metals may be transported through the water column *via* the processes of advection and dispersion, while the adsorbed

particulate fraction may be transported with the sediments, which are governed by sediment dynamics.

The partition of a trace metal between its dissolved and adsorbed particulate fractions depends on the physical and chemical characteristics of the suspended particles, together with various ambient conditions, such as salinity, pH, type and concentration of dissolved organic matter (Turner et al., 2001; Turner and Millward, 2002). Resuspension of contaminated bed sediments caused by strong tidal currents or dredging operations may release a significant amount of trace metals into the water column, and this desorption of toxic metal contaminants from their particulate phase will have pronounced impact on the aquatic environments and ecosystems.



The rare earth elements (REE) of sediments and sedimentary rocks naturally reflect the mineral content of these deposits and reveal the process of their formation/incorporation. REE contents of most sediments and sedimentary rocks are similar in the relative abundance of individual elements although they differ in absolute concentration (Balashov et. al., 1964 and Spirn, 1965). REE are considered as useful tracers of various geological and oceanographic processes (Piper, 1974, Murray and Leinen, 1993). Concentration of these elements in sediments are influenced by (1) particulate supply from the adjacent land masses (Piper, 1974, McLennan, 1989; Nath et al., 2000; Hannigan et al., 2010), (2) biogenic sedimentation from overlying seawater (Murphy and Dymond, 1984) and (3) oxygenation conditions in the water column (Liu et al., 1988; Pattan et al., 2005).

Physical, chemical and biological processes within an estuary vary over a broad spectrum, both in space and time (Morris, 1990; Dyer, 1991). Spatial variations depend on the topography of the estuary, on the gradients in salinities and on the fine sediment concentration (Nair et al., 1992). While annual and seasonal variations depend on the climatological factors, fortnightly and diurnal variations are caused by tidal actions (Holmes, 1986; Vale, 1990).

An estuary provides a particularly unique chemical environment for element cycling (Burton and Liss, 1976; Elderfield, 1978; Morris et al., 1982; Martin and Whitfield, 1983). Within an estuary, the suspended matter carried by rivers is subjected to varying salinity and pH, often causing (1) precipitation of the suspended particles by flocculation (Sholkovitz, 1976, 1978; Sholkovitz and Copland, 1981) and (2) chemical exchange between the suspended sediment and estuarine waters that changes the chemistry of both (Vander Weijden et al., 1977). Gradients in the chemical environment, e.g., in the salt or oxygen content, may change the partitioning of trace metals and their transport mechanisms (Carey, 1990). Estuarine physicochemical gradients, biogeochemical processes, and organism physiology are jointly coordinating the fate and potential effects of metals in estuaries (Machado et al., 2016).

The Cochin estuary is one of the largest tropical estuarine systems along the west coast of India, spreading over 250km² (10°10' N, 76°15'E in the north to 09° 30'N, 76°25'E in the south), and providing a good nursery ground for a wide variety of flora and fauna. Cochin estuary has unique topography; oriented parallel to the coastline and connected to the sea through 2 inlets at Munambam and Kochi (Madhupratap et al., 1987; Srinivas et al., 2004). The Cochin barmouth is 450 m wide and is dredged to a depth of 10-13 m through which tides act within the estuary. Major portion of the estuary has a depth in the range of 2-7 m. The condition of the substrata is predominantly muddy (Balachandran, 2001). The freshwater runoff into the estuary is primarily contributed by the rivers Periyar, Pampa, Achankovil, Manimala, Meenachil, and Muvattupuzha. The river runoff is high during wet season (June - September) characterized by Indian Summer Monsoon (ISM). This period is followed by a moderate runoff period (October - December), and the rest of the year is relatively dry period. The hydrography of the Cochin estuary is greatly influenced by tides, freshwater water flow and estuarine circulation (Revichandran et al., 2011; Shivaprasad et al., 2013). The flushing timescale of the estuary ranged from 1 to 2.5 days during the wet season and were 8.7 days during the dry season (Vinita et al., 2015a). The flow characteristics and the mixing pattern of Cochin estuary are very peculiar and have great influence in the bio-resources, transportation, waste disposal and plankton dynamics (Balachandran et al., 2005; Sooria et al., 2015).

A number of multidisciplinary studies have been undertaken in the Cochin estuary during past several decades. The hydrography of the estuary was investigated by several investigators including Sankaranarayanan and Qasim, 1969; Qasim and Gopinathan, 1969; Sankaranarayanan et al., 1986; Madhupratap et al., 1987; Udayavarma et al., 1981; Gopalan et al., 1983; Chandramohan, 1989; Balachandran, 2001; Srinivas et al., 2003, 2005; Revichandran et al., 2011. The information available includes spatial and seasonal distributions of temperature, salinity, dissolved oxygen, pH, alkalinity, and suspended solids of surface and bottom waters of the region. Recent studies reported the seasonal stratification, tidal intrusion length and spatial tidal asymmetry in Cochin estuary (Shivaprasad et al., 2013; Bincy et al., 2013; Vinita et al., 2015b).

Qasim and Madhupratap, 1979; Nair et al., 1975; Madhupratap et al., 1977; Saraladevi, 1979; Pillai, 1977; Kurup, 1983; Madhu et al., 2007, 2010; Jyothibabu et al., 2006; Vineetha et al., 2015 and several others have investigated the biological characteristics of the estuary in detail. Recent studies reveal the seasonal dynamics of plankton food web, trophic status and the response of autotrophic pico- and nano plankton to summer monsoon (Sooria et al., 2015; Rajaneesh et al., 2015 Arya et al., 2016). The diurnal variations in bacterial and viral abundance in Cochin estuary has been studied (Parvathy et al., 2015). The microbial diversity in the Cochin estuary and their effect on nitrogen fixation and nitrification has been reported recently (Vipin et al., 2015; Jabir et al., 2017).

The greater awareness on the potential health hazards due to the industrial and domestic effluents added to this estuary prompted several researchers to study on the pollution aspects, its carrying capacity, disruption in the life cycle of organisms (Unnithan et al., 1977; Azis and Nair, 1981, Maheswari et al., 2006; Harikumar et al., 2009; Laluraj 2009; Sreekanth et al., 2015; Ramaswamy et al., 2017), morphological changes (Dineshkumar et al., 2014), destruction of natural habitat (Gopalan et al., 1983; Kurup, 1992; Shibu et al., 1990; Sujatha, 1999; Balachandran et al., 2003, 2006), diversity of bioresources (Jiya et al., 2011; Priju et al., 2007), salt water intrusion (Shivaprasad et al., 2013; Vinita et al., 2015a), flow restrictions (Balachandran et al., 2005), environmental impact of dredging (Balchand et al., 2000), eutrophication and changes in biogenic properties (Gopinathan et al., 1984; Balachandran, 2001; Nair et al., 2010; Gupta et al., 2009; Martin et al., 2011, 2013), changes in the quality of organic matter (Renjith et al., 2013; Gireeshkumar et al., 2013a), over exploitation and declining resources (Balchand, 2001; Menon et al., 2000; Balchandran et al., 2003, 2005, 2008b; Martin et al., 2011) etc. The above studies clearly stressed the need for the conservation of the ecosystem for better aquatic life. Rapid industrialization along the estuarine/riverine coasts during the past few decades and the resultant increase in human settlements have led to emergence of several large cities in these areas. These developments and the consequent high pollution load released into the estuary necessitate a critical evaluation of the ecosystem with a view to suggest appropriate remedial measures for its mitigation.

1.2 AIM AND SCOPE OF THE STUDY

A greater knowledge of the biogeochemical process in estuary, which involves the transformation and transport of chemical substances, is critical in understanding the effects of these environmental alterations. The effective management of the input of nutrients and trace metals on coastal seas require continuous monitoring of their riverine inputs and their effective transmission

8

through estuaries, thereby providing a realistic estimate of inputs to the coastal sea. Although several studies, delineating the environmental changes of the Cochin estuary caused by anthropogenic activities have been studied, only Naik et al., 2000 has reported on quantification of the material transport across the estuary. In the present study, we measured the nutrient fluxes through different locations in the Cochin estuary over successive spring and neap tidal cycles of a post-monsoon, monsoon and pre-monsoon seasons. This is the first attempt to quantify the fluxes of nutrients from the Cochin estuary into the sea. Fractionation of rare earths was performed to access the geochemical variability and depositional condition affecting their fate and transport in the estuarine and coastal environment. For unraveling the trace metal dynamics, the metal fluxes through 6 major rivers discharging into estuary and from different locations in the Cochin estuary over a complete tidal cycle were computed. Environmental impact of metal pollution in sediment depends not only on the total concentration of the metals, but also on their mobility and availability, which influence their release / interaction with other components of the ecosystem such as water, plankton and air. Different sequential extraction methods were employed for estimation of the mobility of trace metal in sediments.

1.3 PRIME OBJECTIVES

- To understand the intra-tidal, inter-tidal, spatial and seasonal variations in the dissolved nutrients in the Cochin estuary and their transport to the sea (input and export fluxes of nutrients in the Cochin estuary).
- ✤ To study the present status of trace metals in the sediments of the Cochin estuary and adjacent coastal sea and their chemical

fractionation to identify the mechanism of trace metal accumulation and their mobility.

- To study the distribution and fractionation of rare earths and radioactive uranium and thorium in the sediments of Cochin estuary and adjacent coastal regions to infer their provenance, behaviour and depositional conditions.
- To study the input and export fluxes of trace metals in the Cochin estuary.

****\$\$****

Dynamics of trace metals, rare earth elements and nutrients in the CE



Chapter **Z**

HYDROGRAPHY AND NUTRIENT FLUXES

- 2.1 Introduction
- 2.2 Nutrient flux and its importance
- 2.3 Study region and Methodology
- 2.4 General hydrography
- 2.5 Nutrient fluxes
- 2.6 Summary and Conclusion

2.1. INTRODUCTION

It has been estimated that 61% of the world population lives along the coastal margin (Alongi, 1998). These demographic changes have detrimental effects on the overall biogeochemical cycling in estuaries. Nutrient enrichment is perhaps the most widespread problem in estuaries around the world (Howarth et al., 2000, 2002). Physical processes play an important role in regulating the chemistry and biology of estuaries (Bianchi et al., 1999; Hobbie, 2000). An estuary is a semi-enclosed body of water, where seawater is measurably diluted by freshwater (Hobbie, 2000). The biogeochemical cycles of estuaries are mainly controlled by the river and groundwater discharge, tides, re-suspension and exchange flow characteristics (Leonard and Luther, 1995).

The physical and chemical environment is commonly expressed in terms of water quality parameters such as temperature, salinity, dissolved oxygen, pH and nutrients. The mixing of river water and seawater can be quite



varied in different estuarine systems. The tidal mixing and salinity gradients can significantly affect concentrations of both dissolved and particulate constituents in the adsorption/desorption, flocculation and biological processes. In short, estuarine processes are highly variable with respect to time and space that also influence the water quality.

Nutrients are the raw materials for the marine food chain, as they are essential to the primary production. The elements required in relatively large amounts are generally referred as macro nutrients that constitute carbon, oxygen, hydrogen, nitrogen, and phosphorus. Macronutrients constitute 0.2 to 1% of dry organic weight which also includes sulfur, chlorine, potassium, sodium, calcium, magnesium, iron, and copper. Nutrients needed in trace amounts are generally called micronutrients. These elements often constitute less than 0.2% of dry organic matter. Some common micronutrients required by living organisms include aluminum, boron, bromine, chromium, cobalt, fluorine, gallium, iodine, manganese, molybdenum, selenium, silicon, strontium, tin, titanium, vanadium, and zinc.

Nutrient inputs to estuarine waters come from land, rivers, groundwater, atmospheric deposition, sewage and other anthropogenic activities. Nutrient concentrations have been significantly increased in global rivers by human activity through waste discharge and agriculture (Seitzinger et al., 2002; Statham, 2011; Moore et al., 2003), leading to eutrophication of estuarine and coastal ecosystems (Jickells, 1998; Diaz and Rosenberg, 1995). Anthropogenic inputs have caused many estuarine-coastal systems to shift to N surplus leading to eutrophication (Bricker et al., 1999; Hoch and Kirchman, 1995; Gattuso et al., 1998; Middelburg and Nieuwenhuize, 2000). Increased heterotrophic activities have been observed in such ecosystems due to

anaerobic processes such as denitrification, anammox and methanogenesis (Fasham, 2003). The effects include impacts on species diversity, appearance of harmful algal blooms and development of hypoxia (Jickells, 1998). Though water quality management emphasizes control on the point sources such as wastewater treatment, most of these inputs are "nonpoint sources".

2.2 NUTRIENT FLUX AND ITS IMPORTANCE

In transport phenomena (heat transfer, mass transfer and fluid dynamics), flux is defined as the rate of flow of a property per unit area, which has the dimensions [quantity] x [time]⁻¹x [area]⁻¹. Both natural and anthropogenic activities have significant influence on the transport of nutrients to estuaries and to coastal waters. Estuaries are the intermediate zones of landsea interactions contributing significantly to the nutrient fluxes in to the ocean on varying time scales (Simpson et al., 2001). The nutrient delivery is influenced by catchment hydrology, freshwater flow and tides (Pennock et al., 1994; Justic et al., 1995; Doval et al., 1997; Mackas and Harrison, 1997). Anthropogenic activities enhance nutrient levels lead to eutrophication and oxygen depletion (Beukema, 1991; Parker and O'Reilly, 1991) and often upset the coastal water ecology. Hence, transport measurements are important to study the nutrient economy and productivity patterns of coastal marine systems (Dehairs et al., 2000; McManus et al., 2001).

2.3 STUDY REGION AND METHODOLOGY

2.3.1 Regional Setting

Cochin estuary is the second largest estuary along the west-coast of India that spreads over 250km^2 (10°10′ N, 76° 15′ E in the north to 09° 30′ N, 76° 25′ E in the south) and length of ~80 km providing a nursery ground for a wide variety of aquatic plants, shell and fin-fishes.



Fig 2.1: Cochin backwaters and the stations for the time-series observations

The annual rainfall over the region is about 3200mm of which, nearly 75% occurs during the summer monsoon (June to September). The predominant winds are south-westerly (10 to 12km/h) from June to September and north-easterly from October to January. For the rest of the year, the winds generally remain weak. Cochin estuary has unique topography; oriented parallel to the coastline and connected to the sea through 2 inlets, Munambam and Kochi (Madhupratap et al., 1987; Srinivas et al., 2003). The tides entering

the backwaters are predominantly of mixed semidiurnal (~ 1 m). The depth of the backwater varies considerably from place to place. Along the main channel it is maintained at about 10 to 12 m for navigational purposes while at other places it ranges from 2 to 6 m. The condition of the substrata is predominantly muddy (Balachandran et al., 2001).

The estuary receives nutrients from 6 major rivers namely Periyar, Muvattupuzha, Pamba, Manimala, Achankoil and Meenachil. The increased flushing considerably reduces the salinity of the estuary during monsoon (Shivaprasad et al., 2013). The erratic fluctuations in river run off and rhythmic tidal cycle make the sampling of such estuaries very difficult to study estuarine processes and quantification of fluxes. Cochin estuary has distinct zones based on the tidal excursions a dynamic central zone sandwiched between two relatively weak zones on the north and south (Balachandran et al., 2005). The estuary changes from stratified to partially mixed and completely mixed type while progressing from monsoon, postmonsoon and pre-monsoon seasons (Bincy et al., 2013).

In recent years, increased human interference has led to severe ecological disturbance to the estuary affecting its hydrographical functioning. As a consequence, the estuary has shrunken its area from 315km² in 1912 to 256km² in 1980 (Gopalan et al., 1983). The inter-basin transfer of water from river Periyar to river Muvattupuzha has changed the estuarine circulation. The construction of Thanneermukkam bund has virtually cut off the estuary from the south Vembanad Lake (Kuttanad). Closure of the bund (mid-December to March) arrests tidal propagation further to the south estuary. Consequently, the south estuary became a trap of agricultural wastes to support the growth of water hyacinth (*African payal*).
The ecological studies in the Cochin Estuary were pioneered by Sankaranarayanan and Qasim, 1969 and Sankaranarayanan and Panampunnayil, 1979. They observed a marked seasonal rhythm in the distribution of nutrients induced by precipitation and runoff. The studies of Balachandran et al., 1999 identified the control of freshwater on the dissolved silicate levels in Cochin estuary. Nitrogen loading to this estuary has increased due to increased industrial and agricultural practices, apart from aqua-tourism (Martin et al., 2010, 2013). High nitrification rate has been noticed recently in the intermediate salinities of the Cochin estuary (Miranda et al., 2008). The enrichment of nutrients during pre-monsoon is due to point sources while nonpoint sources are dominant during monsoon (Joseph et al., 2010). Though nutrients are not limiting the chlorophyll production, their physiological responses to nutrients, pH, temperature and salinity are crucial for the biological production in the coastal waters (Balachandran et al., 2008b). To sum up, over the past 5 decade, this estuary has evolved from an autotrophic to a heterotrophic system due to urbanization causing organic enrichment (Shoji et al., 2008; Gupta et al., 2009). The estuarine sediments are found to accumulate organic matter to act as a potential source of P (Gireesh et al, 2013a, b). This has also increased the sediment oxygen demand in the estuary (Abhilash et al., 2012).

The effluents from unprocessed industrial units and domestic sectors have been estimated as 0.104×10^6 and $0.26 \times 10^3 \text{m}^3 \text{d}^{-1}$ respectively, besides the seepage from agricultural, aquaculture fields, port activities, coconut husk retting yards and aqua tourism (Saraladevi et al., 1983, 1991; KSPCB, 1982, 1986; Dineshkumar, et al., 1994). The hydrography of Cochin estuary has deteriorated due to the loading of pollutants exceeding its carrying capacity (Menon et al., 2000; Balachandran et al., 2001, 2005; Miranda et al., 2008; Martin et al., 2008; Robin et al., 2012). This has caused changes in microbial activity affecting the biota (Martin et al., 2008; Miranda et al., 2008; Shoji et al., 2008). The influence of anthropogenic activities on soft benthic community and frequent fish kills are common in recent time (Martin et al., 2011). Accordingly, the major objective of present study was as follows.

To understand the inter-tidal, intra-tidal, spatial and seasonal variations in the dissolved nutrients in the Cochin estuary and their transports to the sea.

2.3.2 Data collection and analysis

In general, there are three distinct seasons: monsoon (June-September), post monsoon (October-January) and pre-monsoon (February-May). Monsoon season is characterized by high precipitation and river discharge, while post and pre-monsoons are considered as moderate and least discharge periods respectively.

The mean daily river discharge during the study period was obtained from Central Water Commission (CWC). During summer monsoon of 2010, the river discharge was 702 and 652.4m³/s on 26th and 27th July (spring phase). During neap phase, the discharges on 2nd and 3rdAugust were 605.4 and 525.3m³/s respectively. During post monsoon (spring), the daily discharges were 207 and 151m³/s on 6th and 7th October 2009, while during neap phase (13th and 14th October), the discharges were very low (80 and 94m³/s). Similarly for the pre-monsoon spring (22nd and 23rd February) and neap (1st and 2nd March), the discharges were very low (14.79, 14.84, 20.59 and 20.21m³/s respectively).

A total of six stations were selected for the time series measurements based on the circulation and input characteristics of the estuary (Fig 2.1). They were S₁ (minor inlet in the north), S₃ (major inlet at Cochin), S₂ (between S₁ and S₃), S₄ (22 km south of S₃), S₅ (43 km south near Thanneermukkom bund) and S₆ (21 km NE in the Periyar River entrance). The stations were selected in such a way that S₅ receives freshwater from 4 rivers, whereas S₆ receives freshwater from the River Periyar. It is also relevant to note that the region between S₃ and S₄ is quite dynamic due to the presence of strong rectilinear currents whereas the water movement is weak between S₄ and S₅ (Balachandran et al., 2008c). Similarly, the position of S₂ is very important, as it represents the stagnant north estuary due to the synchronous tides entering from the two inlets at S₁ and S₃ (Ramamirtham and Muthusamy, 1986).

Six Aanderaa current meters (RCM 9) were deployed at mid-depth of the channel at each location continuously for 30 days for all seasons to record the current speed and direction at 10-min intervals. Considering the deep channel at Cochin inlet, two current meters were moored at the center of the channel at 2 m and near bottom during post-monsoon. During other seasons (pre-monsoon and peak summer monsoon), the current velocity and total cross-sectional discharge through the Cochin inlet were measured using an Acoustic Doppler Profiler (ADP) with bottom-tracking. Water level variation at each location was recorded at 10-min intervals using SBE 26 plus Tide Recorder for the entire period (30 d) of observation. Cochin estuary, being a micro tidal estuary, the maximum tidal amplitude is approximately one meter. The phase of tide with maximum tidal amplitude is the spring period and that with minimum amplitude is the neap period. The selection of spring and neap phase observations are made based on the monthly variations in the water level at the Cochin inlet obtained from Indian Tide Tables. These are again checked during the simultaneous tidal measurements made at this location during the entire observational period (Appendix 1b). It can be seen that the low/high tides of spring/neap phases are different at different locations in the estuary and also during different seasons (though the difference is marginal).

Hourly profiles of salinity and temperature were measured using a conductivity-temperature-depth profiler (CTD19 plus, Sea-Bird Electronics). From all the 6 locations, water samples were collected from 0.5 m below the surface and 0.5m above the bottom at 3-h intervals using 5 L Niskin sampler (Hydro-Bios, Kiel-Holtenau, Germany). The sampling was continued for 24 hours during spring and neap tides and this scheme was adopted for all the three seasons. The sample for dissolved oxygen (DO) was collected without air bubbles in a glass bottle, fixed onboard and later analyzed according to Winkler's method (Grasshoff et al., 1983). The water samples for the determination of various dissolved inorganic nutrients were collected in 500 mL PVC bottles kept in ice boxes and brought to the laboratory within 4 h for analysis. The samples were passed through Whatman GF/F filter paper (pore size $0.7 \,\mu\text{m}$) prior to analysis. Nitrite-N was determined through the formation of a reddish purple azo dye following the method of Bendschneider and Robinson, 1952. Nitrate-N was measured by reduction to nitrite in a coppercadmium reduction column, APHA, 2000. Ammonia-N and silicate-Si were estimated by phenate method, APHA, 2000 and by ascorbic acid method Koroleff, 1983, respectively. Phosphate-P was measured using ascorbic acid as reducing agent, APHA, 2000. All the dissolved nutrients were analyzed using a UV-Vis Spectrophotometer (Shimadzu 1650PC) calibrated with respective standard solutions.

2.3.3 Nutrient flux computation

The cross-section transport Q (kg/day), for each variable can be obtained by integrating the instantaneous flux as

$$Q = A' \int_{t1}^{t2} c\bar{u}dt$$

Where A' is the cross-section area (m^2) , c is the concentration of the parameter (kg/m³), and \bar{u} is the current speed (m/s). The over bar denotes the average speed of water column. The net transport was calculated by integrating the three hourly transports over one tidal cycle. The positive values denote export fluxes (seaward fluxes) and negative values denote import fluxes (landward fluxes). The cross-sectional area of each location was calculated from the respective bathymetry data using Triangulated Irregular Network Programme (Priyakant et al., 2003). A constant value was taken for each cross-sectional area neglecting the tidal variations and the recorded current velocity was corrected with respect to the channel orientation (Appendix 1a).

The nutrient fluxes across the four sampling locations, viz S_1 (minor inlet), S_3 (Cochin inlet), S_5 and S_6 (river boundaries) were calculated (Table 2.1 - Table 2.3) for monsoon, post monsoon and pre-monsoon seasons. The significance of this study is that it is the first attempt to quantify the fluxes of nutrients in the Cochin estuary. We used the current meter data for the post monsoon period, while ADP data was used during pre-monsoon and peak summer monsoon seasons to compute the fluxes of nutrients. The instantaneous discharge (m³s⁻¹) obtained from ADP was multiplied by the corresponding nutrient concentration (μ M converted to kgm⁻³) to obtain the instantaneous nutrient flux. The input fluxes were calculated as the sum of

fluxes across S_5 and S_6 , whereas the export fluxes were calculated as the sum of fluxes across S_1 and S_3 . The net flux was obtained as the difference between input and export fluxes. The positive values for net flux denote a net addition and negative values denotes a net removal of nutrients in the estuary.

2.4 GENERAL HYDROGRAPHY

Estuary is a region where elements are recycled at a faster rate to support living organisms. These elements are to be kept in balance to maintain the health of an estuary. Characterizing the role of estuaries as a nutrient trap or a source requires information regarding the input and flushing qualities of estuaries with their exchange capacities. The general hydrography of Cochin estuary is specified here to evaluate the status of the estuary during the time series observations.

2.4.1 Salinity:-Intra-tidal, diurnal and seasonal variations

Salinity distribution in an estuary is influenced by fresh water discharge, tides, precipitation, evaporation and estuarine circulation (Jassby et al., 1995). Moreover, it is also governed by the morphologic, bathymetric and hydrodynamic characteristics of an estuary. The variation in salinity at the Cochin inlet is found to follow tidal rhythm (Balachandran et al., 2001; Srinivas et al., 2003; Shivaparasad et al., 2013; Vinita et al., 2015b). Fig 2.2 depicts the diurnal variation of surface and bottom salinity along the channel S_1 - S_5 in the Cochin estuary during three different seasons. The diurnal variation of salinity in S_6 , northern upstream is represented in Fig 2.4. It is evident that during peak summer monsoon, heavy rainfall results in high fresh water discharge which eventually reaches estuarine mouth. Freshwater flow during spring tide ($58.5 \times 10^6 \text{ m}^3 \text{d}^{-1}$) was relatively high as compared to that ($48.9 \times 10^6 \text{ m}^3 \text{d}^{-1}$) during neap tide (CWC). The maximum tidal range at the

CHAPTER 2

Cochin inlet was 0.39m during spring tide and 0.26m during neap tide (Appendix 1b). The surface (0.52m/s) and bottom (0.53m/s) currents at the inlet were stronger during spring flood and directed landwards. The advection of saltier water towards surface during spring flood probably resulted in an increase in salinity in the surface. Thus, there was salinity stratification at the Cochin inlet during spring tide with surface fresh water (salinity 2.29-11.36) and saltier bottom (19.16- 33.22).

Fresh water discharge was less during neap period, but inlet showed greater stratification during neap with a salinity gradient 16.5-31.8. The salinity at the surface and bottom varied as (1.93 to 4.76) and (20.17 to 33.73) respectively. During neap flood, relatively weaker surface currents (0.29m/s) and bottom currents (0.34m/s) weakens the advection of bottom water towards the surface and favoured intense stratification. Similar observation was reported in earlier studies (Balachandran, 2001). At the minor inlet S₁ (Maliankara) also, salinity remained higher during spring tide as compared to neap tide due to increased tidal activity. All other stations (S₂, S₄, S₅ and S₆) remained fresh water dominant during both spring and neap tides.

During the post monsoon period, there was heavy rain (~126mm) in the study area during the spring tide, while there was no rain (~1 mm) during the successive neap observation (Data obtained from website of Indian Meteorological Department, http://www.imd.gov.in/). This was well reflected in the daily river discharge data as well as in the salinity distribution in the study region.



Fig 2.2: Diurnal salinity variations in the Cochin estuary during different seasons. The stations S_1 to S_5 are represented by number 1 to 5

The fresh water flow during spring $(15.5 \times 10^6 \text{m}^3 \text{d}^{-1})$ was much higher than during neap tide $(7.5 \times 10^6 \text{m}^3 \text{d}^{-1})$. As a consequence, the water column at the Cochin inlet was well stratified during spring with freshwater (0.63-5.96) stratified over the saltier (3.40-34.31) water. However, during neap the stratification at the Cochin inlet was limited only for a short period (flood), as well-mixed conditions prevailed over for the next 12 h and then regained the stratification for the next 6 h. During neap tide, the seawater intruded into the estuary up to Arookutty S₄ (22km away from inlet) when the fresh water flow was low (Fig 2.3) which supports the view that salinity intrusion mostly

Dynamics of trace metals, rare earth elements and nutrients in the CE

depends on fresh water discharge rather than spring-neap tidal oscillations (Bincy et al., 2013).



Fig 2.3: Diurnal variation in salinity across the Cochin inlet (S_3) and Arookutty (S_4) during a spring and neap tide of summer monsoon.

Intrusion of seawater up to 22km towards the south estuary is an interesting feature to be noted, as indicated by the increase in the bottom salinity at S_4 (salinity gradient > 26) especially during neap tide. During post monsoon however, there was a contrasting behavior at the minor inlet, S_1 as it showed high salinity during neap tide due to very low fresh water flow. Other estuarine stations S_2 , S_5 and S_6 remained fresh water dominated during both tides.

During pre-monsoon observation, the river discharge was at its minimum (spring: $1.763 \times 10^{6} \text{m}^{3} \text{d}^{-1}$; neap: $1.28 \times 10^{6} \text{m}^{3} \text{d}^{-1}$) while the tidal influence was the maximum, enabling a well-mixed and homogeneous condition in the entire estuary (except at S₆). At the inlet, surface salinity (12.23-33.09) varied in accordance with the bottom salinity (26.33-33.20)

during spring tide, while the ranges were 22.19-32.3 and 26.67-32.93 during neap tide. There was stratification at S_6 (Varapuzha) due to the moderate fresh water flow from the River Periyar. Consequently, the northern upstream (Varapuzha) contained freshwater in the surface (neap: 0.99-1.77: spring: 0.66-1.38) and salt water in the bottom (neap: 7.98-12.25; spring: 0.52-11.98). The seawater intrusion into the south estuary resulted in the maximum salinity (9.56-10.35) at S_5 (Thanneermukkam) during this period. The salinity variation generally followed the tidal amplitude.



Fig 2.4: Diurnal variation in salinity at S₆ (northern upstream) during three different seasons: (a) Post Monsoon, (b) Pre Monsoon, (c) Monsoon

2.4.2 Dissolved Oxygen

The dissolved oxygen (DO) is the oxygen that dissolved in water due to diffusion from atmosphere, aeration and photosynthesis. Saturated dissolved oxygen levels are necessary for good water quality. Oxygen is an essential element for all forms of life. Aquatic life is put under stress when the dissolved oxygen levels drop below 5.0mg/l. Dissolved oxygen below 1-2mg/l can result in fish mortality. Numerous scientific studies suggest that 4-5ppm DO is the minimum level that will support a large, diverse fish population. The DO level in good fishing environment generally averages about 9.0ppm. Moreover, oxygen affects a number of esthetic factors like the odor, clarity and taste.

Fig 2.5 depicts the diurnal variation of dissolved oxygen (DO) along S_1 - S_5 channel in Cochin estuary during three seasons. Similar variation for S_6 (northern upstream) is represented in Fig 2.6. Dissolved oxygen was generally high in the surface and low in the bottom throughout the study region. Higher values were observed during ebb compared to flood periods. DO showed an increasing trend from pre-monsoon to post monsoon and to monsoon season in the upstream. Due to bidirectional characteristics, the bottom saltier layer at Cochin inlet showed a lower DO during monsoon and post monsoon periods.

During the peak summer monsoon period, there was substantial reduction in dissolved oxygen (0.61-1.82mg/L) at the inlet during 2^{nd} phase of diurnal tidal cycle during spring phase. There was hypoxic condition in the bottom layer for a short period during spring phase (flood), possibly due to the coastal upwelling. It was more perceptible during neap phase, as the dissolved oxygen further decreased to 0.56-1.16mg/L during flood tide. The intrusion of saline water into the estuary was least during this period, as evidenced from the low oxygen levels limited to the inlet. The DO levels generally increased towards northern and southern estuarine regions with peak values at S₅ (Thannermukkam), especially during neap tide in the surface (7.29-8.37mg/L)

and bottom (6.55-8.82mg/L). During spring tide, there was slight reduction in the DO at surface (6.63-7.94mg/L) and bottom (6.42-7.66mg/L).



Fig 2.5: Diurnal variations in DO (mg/l) in the Cochin estuary during different seasons. The stations S_1 to S_5 are represented by number 1 to 5

During post monsoon, DO showed saturated levels in southern upstream throughout spring and neap tides, especially during neap tides, The average DO in northern upstream during spring tide was 5.83mg/L (surface) and 5.95mg/L (bottom), while during neap tide, it was 6.15mg/L (surface) and 6.27mg/L (bottom). Similarly, in southern upstream, the DO during spring tide was 6.04mg/L (surface) and 5.91mg/L (bottom), while during neap, it was

6.32mg/L (surface) and 6.04mg/L (bottom). The stagnant north estuary (S₂) exhibited lower DO than S₆ during this period, probably due to the flow restrictions (Balachandran et al., 2008c).



Fig 2.6:Diurnal variations in DO (mg/L) at S6 (north upstream) during
(a) Post Monsoon (b) Pre Monsoon (c) Monsoon

The Cochin inlet (S_3) was relatively cooler $(24-26^{\circ}C)$ in the sub surface during spring flood period of the post monsoon. The sub surface waters were also high saline (34.9) and oxygen-deficient (0.53-0.55mg/l), which continued during neap tide to spread further up to S₄ (Arookutty) owing to low fresh water flow. It is an established fact that during the monsoon upwelling, the cool, oxygen-deficient and nutrient-rich waters rise to the surface along the southwest coast of India (Naqvi and Jayakumar, 2000; Habeebrehman et al., 2008). As the upwelled waters advances into the coastal zone, it is further depleted in oxygen due to the increased demand for the oxidation of organic matter (Naqvi and Jayakumar, 2000).

The presence of hypoxic waters (DO < 1.51mg/l) in the bottom was observed in the Cochin estuary during peak summer monsoon (July) of 2009 (Shivaprasad et al., 2013). The present study during post monsoon could further delineate the spatial extent of these hypoxic waters up to 22km inside the estuary. Sub surface hypoxia under prolonged stratification can even result in the development of anoxic conditions (Rabalais and Gilbert, 2009), which will severely affect the biological organisms (Zhang et al., 2010). Another factor that has contributed to the oxygen deficiency in Cochin estuary is the intense bacterial respiration (Shoji et al., 2008). Dissolved oxygen concentration < 2mg/L can induce stress on many aquatic organisms. However, it can be seen from Fig. 2.5 that the oxygen saturation was quickly regained due to the dynamic nature of the estuary.

The estuary remained well mixed during pre-monsoon, when the estuarine inlet recorded maximum oxygen saturation. It was slightly higher during neap phase (surface: 5.10 - 6.37 mg/L; bottom: 4.34 - 5.58 mg/L) than during spring phase (surface: 3.41-5.81 mg/L; bottom: 2.98 - 4.89 mg/L). However, the DO at northern inlet (Munambam, S₁) was < 4mg/L during both spring and neap tides.

During pre-monsoon period, the oxygen deficiency observed in the bottom (spring avg 2.99mg/L; neap avg 2.93mg/L) in the northern upstream (S_6) has not been previously reported (Sankaranarayanan and Qasim, 1969; Saraladevi et al., 1983; Anirudhan et al., 1988; Nair et al., 1988; Kunjikrishna

Pillai, 1991; Sheeba et al., 2000; Robin et al., 2012). While the surface layer at S_6 was oxygenated during spring (3.63 - 5.35mg/L) and neap tides with slightly higher values during neap (4.50 - 7.55mg/L) tide. The oxygen regained saturation at S_6 (Varapuzha) during wet period (monsoon and post monsoon) due to increased fresh water discharge.

During pre-monsoon, the north estuary (S_2) was well oxygenated (surface 5.82-7.65mg/L and bottom 3.16-7.55 mg/L) during neap tide, which was reduced during spring tide (surface 3.77-6.78 and bottom 3.64-5.44mg/L). Although the south estuary remained vertically mixed during this period, (S_4 and S_5), oxygen deficiency was felt in the S_4 bottom, possibly due to increased oxygen demand. The oxygen levels at S_4 in the surface (spring: 3.59-6.57; neap: 4.44-7.26mg/L) during pre-monsoon was slightly lower than that during monsoon. The oxygen deficiency in the bottom at S_4 during neap (3.69-6.64mg/L) and spring (2.86-5.54mg/L) phase was not intense as that in postmonsoon period. However it was well saturated at southern upstream (S_5) during both spring (surface 6.17, bottom 5.47mg/L) and neap (surface 5.96; bottom 5.27mg/L) observations.

2.4.3 Nitrate-N

The dominant dissolved nitrogen species are nitrate, nitrite, and ammonia. In well oxygenated water body, the most abundant and stable inorganic form of nitrogen is nitrate. It is chemically stable, but can be reduced by microbial action. Nitrate enters the water body through surface flow and groundwater as a consequence of agricultural activity (fertilizers and manures), wastewater and oxidation of nitrogenous wastes including septic tanks and other activities. Nitrate stimulates the growth of plankton and water weeds that may increase the fish population. However, excess growth of algae due to high nutrients cause eutrophication, which will deplete the oxygen levels and cause fish mortality.

Fig 2.7 depicts the diurnal variation of nitrate along the channel S_1 - S_5 during three seasons. The variation at S_6 (northern upstream) is represented in Fig 2.8. The results show that apart from the freshwater discharge and tidal activity, nonpoint sources and internal processes are also contributing to the spatial distribution of nitrate in the estuary. Periyar River is a major source of nitrate in to Cochin estuary, as indicated by its high concentration in the northern upstream (S_6) compared to southern upstream (S_5) even though freshwater discharge (from 4 rivers) provided a well oxygenated conditions in the latter. This evidenced the additional supply of nitrate from the industrial activities at S_6 .

During monsoon (spring) period, increased freshwater discharge resulted in high nitrate levels over the entire study region (except $S_1 \& S_4$). Nitrate concentration at S_6 ranged from 28.86-38.89µM in the surface and 9.88 25.01µM in the bottom. The low nitrate levels at S_4 (Arookutty) during spring tide (surface 8.17-17.14µM; bottom 9.24-19.67µM) were concomitant to high NH₄ contents, indicating the dominant internal processes in the estuary. During neap tide, nitrate was maximum (surface 27.26-40.17µM; bottom 31.20-39.12µM) at the minor inlet (S_1) due to nonpoint sources, while it was minimum at southern upstream S_5 (surface 9.69-16.82µM; bottom 9.92-16.93 µM) following its uptake by plankton (high photosynthesis), indicated by saturated water column.



Fig 2.7: Diurnal variations in nitrate (μ M) in the Cochin estuary during different seasons. The stations S₁ to S₅ are represented by number 1 to 5

Nitrate concentrations at the Cochin inlet were higher during ebb and lower during flood periods. The stratification was responsible for the differences in the vertical distribution of nitrate at the inlet, with bottom concentrations (spring 20; neap 17.19μ M) lower than the surface concentrations (spring 23.19; neap 19.63 μ M).



Fig 2.8: Diurnal variations in $NO_3(\mu M)$ at S_6 (northern upstream) during three seasons (a) Post Monsoon (b) Pre Monsoon (c) Monsoon.

Irrespective of spring neap tidal oscillations, the northern upstream (S₆) was enriched with nitrate during post monsoon following its low uptake. However, S₂ the stagnant north estuary was depleted in nitrate due to the increased productivity of the region. The chlorophyll *a* was consistently high at S₂ (spring: 18.6; neap: 9.4mg/m³), whereas it remained $<5mg/m^3$ in other regions.

Even though river discharge was more during spring phase, the nitrate concentration was generally high in the estuary during neap phase of post monsoon period. The average nitrate at S_6 during spring was 24.34 μ M (surface) and 24.8 μ M (bottom) while during neap, it was 34.32 μ M (surface) and 25.92 μ M (bottom). Nitrate at the Cochin inlet varied inversely with

salinity and was lower than that monsoon season. Nitrate levels were low in the bottom (spring 17.4; neap 12.3 μ M) than at the surface (spring 22.7; neap 31.4 μ M).

As rivers are the major source of nitrate, the concentration was very low in the estuary during low discharge period (pre-monsoon). However, high nitrate (15.41µM) was observed towards the northern upstream (S₆) in the surface compared to the bottom (10.77µM). During pre-monsoon spring, the nitrate at the inlet was 1.25-22µM (surface) and 1.89-11.7µM (bottom). During low tidal amplitudes (neap tide), more freshwater reached the inlet, when there was an increase in nitrate levels (surface 6.83-25.10µM and bottom 5.65-17.68 µM). Nitrate was consistently low at S₂ (surface 4.09- 8.56; bottom 3.61-9.6µM) during neap tide when oxygen was high due to increased productivity.

2.4.4 Nitrite-N

The nitrite (NO_2) is a relatively unstable oxidation state of N. Chemical and biological reactions can reduce nitrate or oxidize ammonia to nitrite. Nitrite is an intermediate species formed during nitrification and denitrification processes. Nitrite can cause serious illness (brown blood disease) in fish, even though they don't exist for very long in the environment. Fig 2.9 depicts the diurnal variation of nitrite along the channel S_1 - S_5 in Cochin estuary during three different seasons. The variation of nitrite at S_6 (northern upstream) is represented in Fig 2.10. Nitrite during peak summer monsoon was higher in saline water and lower in freshwater. Accordingly the inlet region showed maximum nitrite levels, whereas the upstream regions S_5 and S_6 showed minimum nitrite during both spring and neap tides. At the Cochin inlet, nitrite levels were $0.45-0.96\mu$ M (surface) and $0.65-2.15\mu$ M (bottom) during spring tide and $0.37-0.46\mu$ M (surface) and $0.30-1.30\mu$ M (bottom) during neap tide. Due to high salinity during spring tide, the nitrite at the minor inlet too showed higher concentration during spring (surface 0.84; bottom 0.93μ M) than neap (surface 0.48; bottom 0.42μ M) tide. High surface nitrite at the upstream S₆ (0.48 μ M) and S₄ (0.65 μ M) especially during spring indicated their riverine origin.

During post monsoon, the distribution of nitrite was dependent on the dissolved oxygen levels in the water column, when the southern upstream (S_5) had relatively low nitrite than northern upstream (S_6) owing to saturated dissolved oxygen at S_5 . The nitrite concentration at S_5 during spring tide was 0.34-0.52µM (surface) and 0.32-0.41µM (bottom), while during neap tide the nitrite was 0.21-0.36µM (surface) and 0.22-0.30µM (bottom). During spring, there was an increase in nitrite (surface 0.41-0.64; bottom 0.52-1.46µM) at the Cochin inlet especially in the bottom (similar to monsoon observations), which was high saline, cool and oxygen deficient, indicating the presence of upwelled waters. However, such an increase was not observed during neap tide (surface 0.23-0.48; bottom 0.08-0.29µM).

During pre-monsoon, even though the water column was homogenous with respect to salinity, there exist vertical gradient in the distribution of nitrite. The surface layer showed higher nitrite than bottom layer in all the stations (except S_4 and S_5) and the lowest nitrite (surface 0.10-0.45; bottom 0.12-0.35µM) was noticed at S_2 , where the water column was well oxygenated.

Spring Surface Spring Bottom Neap Surface Neap Bottom (d) Post- Monsoon 2 <u>n</u> 5 3 4 5 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 9 0.5 0.5 2 Pre- Monsoon Stations 5 0.5 4 0.5 5 9 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 ۱u 2 Monsoon 3 3 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 9 12 15 18 21 24 27 30 33 9



 $\label{eq:Fig2.9:Diurnal variation in nitrite (μM$) in Cochin estuary ($S_1$-$S_5) during three seasons The stations S_1 to S_5 are represented by number 1 to 5.$

Increase in nitrite levels during monsoon and post monsoon at Cochin inlet was associated with coastal upwelling, whereas it is generally low during pre-monsoon period. The nitrite at Cochin inlet during spring phase varied as $0.12-1.01\mu$ M (surface); $0.20-0.64\mu$ M (bottom) and during neap phase as $0.14-1.5\mu$ M (surface) and $0.17-0.92\mu$ M (bottom) throughout pre-monsoon.

Hydrography and nutrient fluxes



Fig 2.10: Diurnal variation of nitrite at S_6 (northern upstream) during different seasons (a) Post Monsoon, (b) Pre Monsoon, (c) Monsoon

During pre-monsoon, there was a remarkable increase in nitrite towards the southern estuary (S_4) up to 1.47µM during spring tide and 1.84µM during neap tide. This was not due to the changes in salinity. Instead, there are food processing industries surrounding S_4 , the sodium nitrite and sodium nitrate used as preservatives (Cammack et al., 1999) may result in this high nitrite. The presence of certain bacterial communities like Ammonia oxidizing bacteria (AOB) and Archea can also enhance ammonia oxidation to increase the nitrite levels (James et al., 2008). Vipindas et al., 2015 observed such bacterial communities and their significant role on ammonia oxidation at S_4 during monsoon. The proximity of station S_4 to the Muvattupuzha river entrance can also be a reason for the nitrite enrichment. An increase in nitrite

Dynamics of trace metals, rare earth elements and nutrients in the CE

CHAPTER 2

(1.58 and 1.93 μ M) was also noticed at the S₅ during pre-monsoon (neap) period. As the Thannermukkam bund is closed during this period, the increase in nitrite can be due to its upward intrusion from S₄ to S₅.

2.4.5 Ammonia-N

Ammonia-N can be the next most abundant form of inorganic nitrogen for phytoplankton growth after nitrate (Stephens et al., 1976). Fig 2.11 depicts the diurnal variation of ammonia along the channel S₁-S₅ in Cochin estuary during three different seasons. The diurnal variation of ammonia in S_6 , northern upstream is represented in Fig 2.12. During monsoon, ammonia concentration showed a decreasing trend towards upstream with the lowest ammonia level at S₅, southern upstream with an inverse relation with dissolved oxygen. The variation in ammonia at S5 during monsoon spring tide was 4.75 (S) and 3.98μ M (B) respectively and neap tide was 3.79 (S) and 2.34μ M (B) respectively (time series average). The ammonia recorded at northern upstream, S₆ was high, possibly due to nonpoint sources of discharge. The ammonia concentration at S_6 during spring was 6.46 μ M (S) and 5.76 μ M (B) and during neap was 5.9 μ M (S) and 5.3 μ M (B). The ammonia was maximum near the minor inlet (S_1) during spring tide and at the major inlet (S_3) during neap tide, coinciding with low DO. The ammonia at S₃ was 6.46µM (S) and 5.76 μ M (B) during spring and 6.9 μ M (S) and 11.0 μ M (B) during neap tide. The surplus ammonia in the surface waters of Cochin estuary (except S₃ & S₄ during neap tide) indicates its estuarine origin. Recent studies have shown that nutrient concentrations in the Cochin estuary are often controlled by organic matter decomposition, nitrification, and nitrogen fixation apart from sediment re-suspension (Miranda et al., 2008; Vipindas et al., 2015). In addition to the

riverine inputs, there are additional sources of ammonia in to the estuary from several tributaries, agricultural drains and sewage treatment plants.

Similar to monsoon observation, nitrate was the major nitrogen fraction during post monsoon period also, while ammonia was relatively low. During post monsoon spring, minor inlet showed enhanced level of ammonia together with lowest dissolved oxygen while during neap tide the highest ammonia was observed at S_4 when the dissolve oxygen decreased with intrusion of high saline water. Ammonia showed moderately higher values in northern upstream when compared to southern upstream S_5 due to the highly oxygenated condition at S_5 . The ammonia at S_4 during spring was 4.38µM (S) and 3.44µM (B), while during neap, it was 15.14µM (S) and 11.23µM (B).

Ammonia concentration was noticeably high in the Cochin estuary during pre-monsoon, with a decreasing trend towards upstream. The maximum concentration was noticed at inlet and Arookutty (south estuary). At the inlet, it was averaged to 22.86 μ M (S) and 15.49 μ M (B) during spring tide and 16.7 μ M (S) and 16.1 μ M (B) during neap tide. At S₄ ammonia averaged to 18.03 μ M (S) and 11.74 μ M (B) during spring tide and varied as 3.9-87.6 μ M (S) and 2.8-23.9 μ M (B) during neap tide. Ammonia was generally high at the surface in the entire estuary except at S₆, where it was stratified during premonsoon, where the average concentration was 6.9 μ M (S) and 20.2 μ M (B) during spring and 10.3 μ M (S) and 16.5 μ M (B) during neap tide. High ammonia along with oxygen deficiency in bottom of S₆ is probably due to remineralization during dry season (Pratihary et al., 2014).

CHAPTER 2



Fig 2.11: Diurnal variation in ammonia (μ M) in the Cochin estuary (S₁-S₅) during three seasons. The stations S₁ to S₅ are represented by number 1 to 5.

During pre-monsoon, the southern upstream S_5 showed higher ammonia in surface than northern upstream especially during neap tide. The average ammonia at S_5 during spring tide was 16.4µM (S) and 8.8µM (B) and during neap tide, it was 21.1µM (S) and 18.2µM (B) respectively. Bhavya et al., 2015 have shown that despite a high nitrification and biological assimilation, ammonia levels are consistently high in the Cochin estuary.



Fig 2.12: Diurnal variation of ammonia at S_6 (northern upstream) during different seasons (a) Post Monsoon, (b) Pre Monsoon, (c) Monsoon.

2.4.6 Silicate-Si

The Si is primarily derived from rock weathering and hydrothermal vents, while its removal is through biogenic processes (Treguer et al., 1995). Silicon is required as a macronutrient by certain groups of plants such as diatoms, which account for 25% of the world's net primary production (Willen, 1991). The cycling of Si is thus, strongly dependent on its assimilation and dissolution of diatoms frustules. Fig 2.13 depicts the diurnal variation of silicate along the channel S_1 - S_5 in Cochin estuary during three different seasons. The diurnal variation of silicate in S_6 , northern upstream was represented in Fig 2.14. During peak summer monsoon, increased freshwater discharge increased the silicate concentration in the northern

estuary (S_1 , S_2 and S_6) compared to southern estuary (S_4 and S_5) especially during neap tide despite a high salinity (Fig. 2.2). Silicate observed at S_2 , stagnant north estuary and S_6 was more or less similar, which was high during spring tide when river discharge was also high. During spring tide, silicate was 114.79-136.01µM (S) and 121.17-136.63µM (B) at S_2 , while during neap tide; it was 107.77-132.10µM (S) and 108.72-132.10µM (B). Similarly, at S_6 it was 117.55-133.49µM (S) and 113.76-137.81µM (B) during spring tide, whereas it was 118.33-127.12µM (S) and 111.43-154.71µM (B) during neap tide. At the Cochin inlet, the water column was stratified and the vertical gradient in silicate was also high. Silicate (spring 47.75; neap 56.56µM) was less at the bottom than surface (spring 106.67; neap 113.81µM) and, was higher during ebb than during flood period.

During post monsoon period, the silicate levels were lowered due to reduction in the river discharge. During this period, though river discharge was more during spring, silicate was high during neap phase at S₂, S₅ and S₆. The silicate at S₆ during spring was 109.2 μ M (S) and 105.63 μ M (B) while during neap, it was 117.73(S) and 118.92 μ M (B). At the Cochin inlet, silicate was low in the bottom (spring 41.3; neap 21.6 μ M) than the surface (spring 86.8; neap 48.2 μ M). Since the intrusion of salt water during neap period was extended up to S₄, there was considerable reduction in silicate levels at the S₄ bottom during neap. The average silicate observed at S₄ during spring tide was 104.5 μ M (S) and 106.52 μ M (B), while during neap tide it was 106.94 μ M (S) and 79.14 μ M (B).



Fig 2.13: Diurnal variation in Silicate (μ M) in Cochin estuary during three seasons. The stations S₁ to S₅ are represented by number 1 to 5.

During pre-monsoon, as the tidal activity increased, the salinity at S_1 and S_2 during both tides increased as compared to S_4 and S_5 , while the silicate levels decreased. Since the sea water incursion was not strong enough to mask the fresh water flow, silicate concentration at S_6 increased in the surface during spring tide when river discharge increased.

Dynamics of trace metals, rare earth elements and nutrients in the CE



Fig 2.14: Diurnal variation of silicate at S₆ (northern upstream) during different seasons (a) Post Monsoon, (b) Pre Monsoon, (c) Monsoon.

The average silicate levels at S_6 were 68.5μ M (S) and 45.5μ M (B) during spring while it was 45.6μ M (S) and 37.2μ M (B) during neap. At the southern upstream (S₅), since the water column was isohaline (~10), the silicate at both surface (41 - 45.9μ M) and bottom ($40.5 - 42.7\mu$ M) were slightly lower than its concentration in the northern upstream (except at bottom during neap phase).

2.4.7 Phosphate-P

Phosphorus is a major nutrient regulating the growth and production of phytoplankton. Natural sources of phosphates include phosphate-containing rocks and solid or liquid wastes. It enters the estuary through fertilizers, industrial effluents and cleaning compounds. Its concentration helps to predict the total biomass of phytoplankton. The form of phosphorus involved in the biogeochemical processes is the phosphates. Estuarine sediments are generally rich in phosphorus which may be liberated to overlying waters under favorable conditions.

Fig 2.15 depicts the diurnal variation of phosphate along the channel S_1 - S_5 in Cochin estuary during three different seasons. The diurnal variation of phosphate in S₆, northern upstream was represented in Fig 2.16. A marked spatial variation in phosphate was evident from the observation. During monsoon, the entire estuary showed moderate level of phosphate (spring 0.29-4.83: neap 0.12-4.85 μ M), the low concentrations towards the upstream (S₅ & S_6). The average phosphate concentration at northern upstream, S_6 was 0.69μ M (S) and 0.92μ M (B) during spring tide, while it was high 1.15μ M (S) and 1.20 μ M (B) during neap tide. In the south estuary S₅, phosphate was consistently low (0.19-0.53µM) irrespective of tidal variation. During monsoon (spring), PO₄ was slightly higher (2.15-3.12 μ M) at S₂ via external input, while it was high at minor inlet S_1 along with high nitrate and silicate during neap indicating higher river input. It's obvious that phosphate at the major inlet (with pH >7.7) was lower in the bottom (spring 1.50; neap 1.71μ M) than at the surface (spring 2.09; neap 2.08 μ M) due to its sea water origin. As the minor inlet is shallow (~ 2.5 m), the phosphate at bottom (spring 1.97; neap 1.86µM) is higher than at surface (spring 1.50; neap 1.80 µM) due to churning of sediments. During peak monsoon, the discharge through Muvattupuzha River and non-point sources from small tributaries/canals and sewage inputs added more phosphate to S_4 (pH 6.6-7.4) to increase the concentration to 1.56-2.53µM.



Fig 2.15: Diurnal variation of phosphate(μM) in Cochin estuary during different seasons. The stations S₁ to S₅ are represented by number 1 to 5.

During post monsoon period, the phosphate levels in the estuary varied from 0.18 to 3.2 μ M during spring and from 0.17 to 3.9 μ M during neap phase. There was phosphate enrichment at S₂ (3.9 μ M), S₁ (2.5 μ M) and S₃ (2.0 μ M) during neap tide. The variation in the phosphate concentration, especially at S₂ could be due to increased re-suspension or domestic discharge. Desorption of sediment phosphorus at higher salinity has been reported for the Cochin estuary (Rajagopal, 1974; Martin et al., 2008; Gireesh et al., 2013b). There are several aquaculture fields near S₂, where ammonia and phosphorus compounds are used to destroy the predators (Kartha and Karunakaran Nair, 1980). As a result, the N/P ratio in this region was low (<10) compared to its average estuarine value of 25.



Fig 2.16: Diurnal variation of phosphate at S₆ (northern upstream) during different seasons (a) Post Monsoon, (b) Pre Monsoon, (c) Monsoon.

Phosphate concentration in the estuary is strongly controlled by the pH during pre-monsoon season. It is apparent that phosphate concentration is low at pH < 6.9 (station S_5) due to adsorption to suspended particles. The phosphate concentration at S_5 during spring and neap tides were 0.03-1.48µM and 0.09-0.36µM, respectively. Phosphate was high in waters above pH 7 (S_1 , S_2 , S_4) following its release from particles into water column.

2.5 NUTRIENT FLUXES

The input and export fluxes of nutrients through Cochin estuary during 2 successive tides of peak summer monsoon are depicted in Fig 2.17. CHAPTER 2

During peak summer monsoon, the fluxes of all nutrients through southern upstream was high during spring phase corresponding to increased river discharge through southern rivers, however the fluxes of all the nutrients (except NH_4) through northern upstream was high during neap phase. The river discharge through Periyar was more or less similar during spring (196.7m3/s) and neap (196.4m3/s) tides, but there was an increase in the nutrient fluxes across the northern upstream (S₆) during neap phase, due to enrichment of these nutrients (nitrate, nitrite, silicate and phosphate) in the source water. However, the ammonia flux was high during spring tide due to its enriched levels.



Fig 2.17: Input and export fluxes of nutrients (tons/day) through Cochin estuary during summer monsoon (a) spring tide (b) neap tide.

The flux estimation reveals that the Cochin estuary receives 21.42 tNO₃, 0.39t NO₂, 0.30t PO₄, 96.67t SiO₄ and 7.41t NH₄ from northern upstream each day during spring phase and 29.25t NO₃, 0.48t NO₂, 1.63t PO₄, 140.01t SiO₄ and 5.87t NH₄ per day during neap phase of summer monsoon. The south of the estuary received high freshwater input, which increased the nutrient input during spring phase (379.4t NO₃; 5.06t NO₂; 85.94t NH₄; 17.68t

 PO_4 & 3662.94t SiO₄) and neap phase (118.95t NO₃; 2.38t NO₂; 11.70t NH₄; 11.46t PO₄ & 1813.93t SiO₄), which was very high compared to the fluxes from the northern upstream (Table 2.1).

Variables	Input through Periyar (t/d)	Input through southern rivers (t/d)	Net input (t/d)	Export through minor inlet (t/d)	Export through major inlet (t/d)	Net export (t/d)	Net flux (t/d)					
(a) spring tide												
NO ₃ -N	21.4	379.4	400.8	20.6	1085.0	1105.5	704.7					
NO ₂ -N	0.4	5.1	5.4	0.4	38.5	38.9	33.4					
NH ₄ -N	7.4	85.9	93.4	10.1	339.2	349.3	255.9					
DIN	29.2	470.4	499.6	31.0	1462.6	1493.6	994.0					
PO_4	0.3	17.7	18.0	4.3	194.4	198.7	180.7					
SIO_4	96.7	3662.9	3759.6	198.7	7842.6	8041.4	4281.8					
(b) Neap tide												
NO3-N	29.2	118.9	148.2	11.6	812.9	824.5	676.3					
NO2-N	0.5	2.4	2.9	0.3	22.5	22.8	20.0					
NH4-N	5.9	11.7	17.6	2.5	274.8	277.3	259.7					
DIN	35.6	133.0	168.6	14.5	1110.1	1124.6	956.0					
PO4	1.6	11.5	13.1	3.9	202.5	206.4	193.3					
SIO4	140.0	1813.9	1953.9	105.6	7896.5	8002.1	6048.2					

Table 2.1: Nutrient fluxes (td⁻¹) through Cochin estuary during peak summer monsoon (a) spring phase & (b) neap phase

Negative sign for input and export fluxes denotes landward fluxes and positive sign denotes seaward fluxes.

Negative sign for net flux represents net removal and positive sign represents net addition of nutrients other than riverine.

During spring tide, as the freshwater discharge into the estuary (through 6 rivers) was high, more freshwater reached the inlet and the estuary flushes out more nutrients during spring phase (except PO_4 & SiO₄). The export fluxes through the Cochin inlet during spring and neap tides was (1085 & 812.9)t NO₃, (38.5 & 22.5)t NO₂, (339.2 & 274.8)t NH₄, (194.4 & 202.5)t

PO₄ and (7842.6 & 7896.5)t SiO₄ per day. The corresponding fluxes through minor inlet (S₁) were relatively low (20.56 & 11.63)t NO₃, (0.41 & 0.31)t NO₂, (10.05 & 2.52)t NH₄, (4.29 & 3.88)t PO₄, (198.73 & 105.62)t SiO₄ per day. The net fluxes through Cochin estuary was calculated as the difference in the input and export fluxes (input-export). It should be noted that during monsoon period, the estuary exported 704.7, 33.43, 255.9, 180.73 and 4281.8 tons of NO₃, NO₂, NH₄, PO₄ and SiO₄ respectively per day during spring tide and 676.3, 19.95, 259.7, 193.31 and 6048.15 tons of NO₃, NO₂, NH₄, PO₄ and SiO₄ respectively for utrients is greater than the river input, indicating the internal processes (nonpoint sources) to be quite significant in the estuary during monsoon period.

Despite a higher input of nutrients during post monsoon, the export of nutrients from the Cochin estuary was high during monsoon, possibly due the increased monsoonal flushing. The nutrient inputs from southern rivers were higher than from the northern river (River Periyar). The input and export fluxes of nutrients through Cochin estuary during post monsoon are depicted in Fig 2.18. During post monsoon spring, the Cochin estuary received 444, 7.7, 64.6, 22.2 and 4149.9tons of NO₃, NO₂, NH₄, PO₄ and SiO₄ and exported 485, 16.6, 90.4, 85.5 and 3922 tons of NO₃, NO₂, NH₄, PO₄ and SiO₄ per day (Table 2.2). Similarly, during neap tide, the input and export fluxes varied from 383 to 468t DIN, 11 to 97t PO₄ and 3109 to 2476t SiO₄. To sum up, there was a net export of nutrients (except SiO₄) during post monsoon, while SiO₄ was retained in the estuary.

It is evident that between successive spring and neap tides, there was an increase in the net fluxes (Input-Export) of phosphate 63.3 to 86.6t, ammonia 25.8 to 43.1t and DIN 76.4 to 85.8t except for nitrate 41.7 to 35.8t, whereas silicate showed substantial removal during spring (-227t) and neap (-633t) phases.

	U N	/ 1 01	~ /	1 1								
Variables	Input through Periyar (t/d)	Input through southern rivers (t/d)	Net input (t/d)	Export through minor inlet (t/d)	Export through major inlet (t/d)	Net export (t/d)	Net flux (t/d)					
(a) Spring												
NO ₃ -N	74	370	444	26	459	485	41.7					
NO_2N	1.64	6.08	7.7	0.61	16	16.6	8.9					
NH ₄ -N	15.5	49.11	64.6	5	85.37	90.4	25.8					
PO_4	7	15.18	22.2	5	81	85.5	63					
SiO ₄	636.1	3513	4149	173	3749.8	3923	-227					
(b) Neap												
NO ₃ -N	55	22	280.6	8.28	308.12	316.4	35.8					
NO ₂ -N	1.06	3.1	4.14	0.26	10.8	11.1	6.9					
NH ₄ -N	24.6	73.1	97.7	6.5	134.3	140.8	43.1					
PO_4	3.9	7.2	11.1	1.04	96.7	97.7	86.6					
SiO_4	298.8	2810.8	3109	78.5	2398.1	2476	-633					

Table 2.2: Nutrient fluxes (t/d) through Cochin estuary during post monsoon during (a) spring phase (b) neap phase

Negative sign for input and export fluxes denotes landward fluxes and positive sign denotes seaward fluxes

Negative sign for net flux represents net removal and positive sign represents net addition of nutrients other than riverine.

During spring phase, the net nitrate flux was high due to well stratified condition following high river flow. However, higher fluxes of PO_4 and NH_4 during neap phase when the freshwater flow diminished could be due to the high concentration of these nutrients in the estuary. The water column during neap phase was vertically mixed at the Cochin inlet, which increased the export of phosphate and ammonia.
CHAPTER 2



Fig 2.18: Input and export fluxes of nutrients (tons/day) through Cochin estuary during post monsoon (c) spring phase (d) neap phase.

The input and export fluxes of nutrients during the pre-monsoon are depicted in Fig 2.19. The input of nutrients through rivers was relatively low due to reduced freshwater flow. The nutrient input during spring phase was slightly higher than during neap phase as freshwater input during spring period was relatively higher. The concentration of nutrients at S_6 was higher than at S_5 during this period. As the tides were towards upstream during neap, the input fluxes of inorganic nitrogen (DIN) and silicate from the north estuary was considerably low compared to its fluxes from the south estuary (Table 2.3). Input of nitrate, nitrite, ammonia and silicate was 7.40, 0.21, 7.85 and 29.77 tons from the north estuary and 14.35, 0.41, 13.50 and 134.44 tons from the south estuary during spring tide. The phosphate flux was higher from the north than south due to its higher concentration in the north. Approximately 1.13t of phosphate has entered the estuary from north estuary, while its input from the south estuary was negligible during pre-monsoon spring.



Figure 2.19: Input and export fluxes of nutrients (tons/day) through Cochin estuary during pre-monsoon (e) spring phase (f) neap phase.

Even though freshwater discharge was relatively high during premonsoon spring than neap, the nutrient delivery was less due to increased tidal activity and landward flow at the Cochin inlet to reduce the nutrient export. However, the export of nitrate (2.5t), nitrite (2.2t), and silicate (93.4t) through Cochin inlet was several times higher than their respective fluxes through the minor inlet (0.12t NO₃, 0.11t NO₂ & 25.38t SiO₄) during spring phase. The input of NO₃ and SiO₄ during pre-monsoon spring exceeded their export fluxes, indicating their estuarine removal (19.35t NO₃, and 45.42t SiO₄). Nitrate is normally consumed by phytoplankton, while the removal of silicate can be through biological uptake as well as adsorption on to suspended particles (Balachandran et al., 1999). However, the export fluxes of NO₂ and PO₄ during pre-monsoon spring was greater than their input fluxes resulting in a net export of 1.64t NO₂ and 10.67t PO₄. The net export of phosphate is due to desorption of phosphorus at high pH and salinity (Chambers et al., 1995). The high pH probably favors the sorption of Si to and release of P from sediment, as both reactions were competing for the same anion exchange sites at the inlet (Koski et al., 2001). Thus, SiO₄ is retained in the estuary due to its

burial in the sediment. The magnitude of P release also depends on the amount of Fe and Al oxides in the sediment. The enrichment of nitrite in the south estuary resulted in its net export.

Table 2.3:	Nutrient fluxes (t/d) through Cochin estuary during pre-monsoon:
	(a) Spring tide (b) neap tide.

Variables	Input through Periyar (t/d)	Input through southern rivers (t/d)	Net input (t/d)	Export through minor inlet (t/d)	Export through major inlet (t/d)	Net export (t/d)	Net flux (t/d)
			(a) Spr	ing tide			
NO3-N	7.4	14.35	21.75	-0.12	2.5	2.39	-19.35
NO2-N	0.21	0.41	0.62	0.11	2.2	2.26	1.64
NH4-N	7.85	13.5	21.35	5.66	-132.9	-127.24	*148.58
DIN	15.46	28.25	43.71	5.65	-128.3	-122.65	*166.36
PO4	1.13	-0.07	1.05	1.67	10.1	11.72	10.67
SIO4	29.77	134.44	164.21	25.38	93.4	118.79	-45.42
			(b) Ne	ap tide			
NO3-N	-21.56	-0.96	-22.51	-3.1	81.13	78.03	100.55
NO2-N	-0.62	0.55	-0.07	0.09	4.88	4.97	5.04
NH4-N	-13.65	-11.2	-24.85	3.31	245.68	248.99	273.84
DIN	-35.83	-11.6	-47.43	0.3	331	331.3	378.73
PO4	-1.37	0.06	-1.31	0.69	20.06	20.75	22.05
SIO4	-76.54	62.79	-13.75	-0.97	108.07	107.1	120.85

Negative sign for input and export fluxes denotes landward fluxes and positive sign denotes seaward fluxes

Negative sign for net flux represents net removal and positive sign represents net addition of nutrients other than riverine.* denotes additional input from the adjacent coast.

At the Cochin inlet, the landward (negative) fluxes of NH_4 indicate a daily net import of 148.6t NH_4 in the estuary. High concentrations of NH_4 during flood period were responsible for the negative fluxes during spring. The source of ammonia in the coastal waters may be the re-mineralization in the sediment (Dunn et al., 2008). Studies have shown that the nutrient

requirement for phytoplankton production over the western Indian inner shelf is met by benthic regeneration of ammonia (Naqvi et al., 2006; Pratihary et al., 2014). This ammonia-rich water has likely adverted in to the estuary due to strong flood currents (Vinita et al., 2015).

During pre-monsoon (neap), the river flow was considerably reduced and the fluxes of nutrients through northern rivers were negligible and directed landward. However there was riverine input of nitrite, silicate and phosphate in to the estuary through the south estuary during this period. Since the tidal activity was slightly lowered during neap there was a net export of all the nutrients at the major inlet. The net export during neap was 100.5t NO₃, 5.04t NO₂, 273.8t NH₄, 22.05t PO₄ and 120.85t SiO₄ per day. The maximum fluxes were for NH₄ and minimum for NO₂. The residual fluxes at Cochin inlet showed an ebb dominance, as the ebb currents could transport nutrients into the sea even during pre-monsoon (Vinita et al., 2015b).

2.5.1 Nutrient fluxes - a comparison with other estuaries

A comparison is made here on the net nutrient fluxes for Cochin estuary with other Indian estuaries such as Chaliyar, Tapi, Hoogly and Meghna (Table 2.4). Flux studies in Chaliyar estuary reported a net transport of nitrogen into the sea during wet and into the river during dry period (Xavier et al., 1993). The Meghna estuary exhibited a net addition of nutrients (Uddin et al., 2014), while Tapi estuary showed a net removal of P and N (Bapardekar et al., 2004) and Hoogly estuary showed a net removal of SiO₄ (Mukhopadhyay et al., 2006). Cochin estuary showed a net addition of nutrients during wet and a net removal (SiO₄ and NO₃) during dry periods. The N fluxes of our system were very high compared to the fluxes through other estuaries. However, the N and P fluxes during monsoon were comparable with that of the Hoogly estuary. The N fluxes during pre-monsoon were higher than for the Meghna estuary while it was comparable during monsoon and post monsoon periods. The P fluxes from Cochin estuary was lower than that reported for Meghna estuary.

	ΔP	Δ SiO ₄	Δ	Δ	Δ	Δ DIN	Location	Reference
Indian estuaries			NO_3	NO_2	NH_4			
indian estuaries								
Tapi	-17.3					-63.7	Madypradesh	Bapardekar et al., 2004
Hoogly	10.7	-84.9				67.7	West Bengal	Mukhopadhyay et al., 2006
Meghna (Pre)	337.0					28.5	Bangladesh	Uddin et al., 2014
Meghna (Mon)	690.3					79.7	,,	,,
Meghna (Post)	3370.9					1026.6	"	,,
Chaliyar			-0.1			-0.2	Kerala	Xavier et al., 1993
CE (Pre: spring)	10.7	-45.4	-19.4	1.6	-148.6	-166.4	Kerala	Present study
CE (Pre: Neap)	22.1	120.9	100.6	5.0	273.8	378.7	"	,,
CE (Mon: spring)	63.0	-227.0	41.7	8.9	25.8	76.0	"	,,
CE (Mon: Neap)	86.6	-633.0	35.8	6.9	43.1	85.0	"	,,
CE (Post: spring)	180.7	4281.8	704.7	33.4	255.9	994.0	"	,,
CE (Post: Neap)	193.3	6048.2	676.3	20.0	259.7	956.0	,,	,,

Table 2.4: Non-conservative/net fluxes $(\pm \Delta)$ td⁻¹for different estuaries in India.

CE – Cochin estuary

2.6 SUMMARY AND CONCLUSION

This is the first attempt to quantify the fluxes of nutrients from the Cochin estuary into the sea. The fresh water flow and tidal activity are found to regulate the distribution of salinity and nutrients in the Cochin estuary. There was stratification in the lower estuary during summer monsoon and post monsoon periods, whereas it was transformed into a well-mixed condition during pre-monsoon. The seawater incursion into the estuary was limited to the lower estuary during monsoon (spring tide), while it reached upto Arookutty (22 km south of the inlet) during post monsoon (lowered freshwater

56

flow). During pre-monsoon, the entire estuary changed to homogenous condition with complete vertical mixing. The influence of freshwater discharge was well reflected in the salinity structure during the spring-neap phases.

Depletion in dissolved oxygen in the north estuary (S_6) during premonsoon was due to heterotrophic re-mineralization, while upwelling was responsible for the low DO at the inlet during monsoon and post monsoon periods. The intrusion of oxygen deficient waters coupled with heterotrophic re-mineralization has reduced the DO at S_4 . An increase in dissolved oxygen was observed towards southern upstream (Thanneermukkam).

The most abundant inorganic form of nitrogen in the Cochin estuary, nitrate, was high at S_6 (Varapuzha). During pre-monsoon, ammonia was abundant at certain locations like Nedungad, Cochin inlet and Thanneermukkam. At Nedungad, ammonia was relatively high due to greater uptake of nitrate by phytoplankton. Nitrite was high in oxygen deficient waters, which decreased towards upstream. During low discharge period, bacteria mediated nitrification increased the nitrite levels in the south estuary.

The estuary showed high ammonia in the surface indicating its fresh water contribution. It was generally high during ebb tide and low during flood tide at the inlet, however during pre-monsoon neap; it was high during flood tide. The concentration of ammonia generally increased when dissolved oxygen decreased and vice versa. Thus, the lowest ammonia was recorded at southern upstream containing saturated dissolved oxygen. Higher ammonia along with the oxygen deficiency at Varapuzha (bottom) during dry season may be due to heterotrophic re-mineralization. At Nedungad (S_2), DO was saturated and ammonia was high indicating a maximum uptake of nitrate by

phytoplankton. The high chlorophylla coinciding with low nitrate are indications of productive waters at S₂.

The entire estuary showed moderate levels of phosphate, while south estuary showed minimum levels, which also showed high photosynthetic activity. Adsorption of P to suspended particles is likely in this region due to slightly acidic waters (pH < 6.9). The phosphate was maximum at Nedungad (S_2), possibly due to external inputs. In general, the phosphate distribution in the Cochin estuary is mostly controlled by the tidal activity, pH, river discharge and external inputs (sewage, agricultural activity and industrial effluents). The adsorption and desorption of P follows a buffering mechanism to regulate the phosphate concentrations in estuaries (Tallberg et al., 2000).

The major source of silicate in the estuary was the freshwater input. Silicate concentration was high during spring than neap period and showed an inverse trend with salinity. Silicate was slightly higher in the northern upstream than the southern upstream.

The transport of nutrients was generally directed seaward except for pre-monsoon (spring), when they were landward due to increased tidal forcing. The input of nutrients from the south estuary was several times higher than their respective fluxes from the north estuary, especially during post monsoon period. However, there was an import of NH₄ during pre-monsoon (spring) from the adjacent coastal region. The export was maximum during monsoon following increased freshwater flow. The export of nutrients during monsoon could also be due to the addition of nutrients other than rivers. The SiO₄ flux was maximum during monsoon, though it was substantially removed during estuarine mixing. There was net export of all the nutrients (except SiO₄) during post monsoon with maximum net flux for PO₄. While SiO₄

retained within the estuary. The adsorption of SiO_4 on to particles at higher pH at the inlet can be the reason for net removal during post monsoon. During pre-monsoon, the estuary showed a net export during neap because the net water flow was into the sea. However during spring phase, the estuary showed a net addition for NO₂ and PO₄ and a net removal for NO₃ and SiO₄. The maximum addition of PO₄ in the estuary occurs during post and pre-monsoon periods. The high pH might promote the sorption of Si to and release of P from sediment leading to a net burial of Si and release of phosphate in the estuary during post and pre-monsoon period. The removal of NO₃ during pre-monsoon spring was related to uptake by planktons (high chlorophyll *a*). It is apparent that the intra-tidal and spring-neap variations in the nutrients were well reflected in the net transport of nutrients through the Cochin inlet.

There was a reduction in the net fluxes of nutrients from spring to neap phases of monsoon due to variations in the river discharge. However, the net fluxes of phosphate and ammonia during post monsoon were high during neap compared to the spring period, because there was an increase in their ambient concentrations during the former period. The flux of all nutrients decreased considerably during pre-monsoon (spring) period, probably due to the relatively low nutrient concentration and the reduction in the flushing characteristics.

The present study shows that Cochin estuary is a source of all nutrients during monsoon and a sink for SiO₄ during post and pre monsoon. The removal (15-20%) of SiO₄ during post-monsoon was coincided by a net addition of N and PO₄. During pre-monsoon spring, 88.9% of NO₃ and 27.6% of SiO₄ was removed from the estuary due to biogeochemical processes. It has been estimated that approximately 148.6tons of NH₄ was retained in the

estuary as an additional input from coastal region. In view of this, Cochin estuary is a small source of nutrients to the adjacent coastal region.

The spring neap tidal oscillations play an important role in net transport of materials across the Cochin estuary during pre-monsoon, whereas freshwater discharge plays a major role during monsoon. During post monsoon, both freshwater discharge and tidal activity together controls the net transport. The internal processes such as nitrification, heterotrophic remineralization, and adsorption-desorption, stratification etc played important roles in the organic enrichment in the Cochin estuary (Gupta et al., 2009; Shoji et al., 2008; Gireesh et al., 2013a). The organic matter influences the metabolic functions and nutrient transports of estuaries. The upwelling and benthic regeneration are other important processes in the inlet region. Present study shows that despite a moderate utilization of nutrients, substantial amount of nutrients are exported from the Cochin estuary, especially during peak summer monsoon. The nutrients were generally high during this period due to intense rain fall and river flow. The six rivers are generally connected to numerous channels in the estuary, which gather more nutrients on their course. However, the high run off maintains a better flushing of the estuary to regulate the nutrient balance and safe guard the estuary from wide spread eutrophication.

The present estimates of nutrient fluxes from Cochin estuary was higher than some estuaries (Chaliyar and Tapi estuary), but comparable with that of Hoogly and Meghna estuaries. The P fluxes observed for Cochin estuary is extremely lower than that reported for Meghna estuary and comparable to that reported for Hoogly river estuary. The significance of the study is that the export fluxes from the Cochin estuary could be a major factor sustaining the spectacular monsoon fishery along the southwest coast of India. The Cochin estuary seems to be a source of nutrients to the adjacent coastal areas, as evidenced by the increased productivity and frequent occurrences of blooms in the adjacent coastal areas, even during dry season (Madhupratap et al., 1987; D'Silva et al., 2012). The eutrophication in coastal waters to a certain extent increase the fishery production however as a consequence of monsoon upwelling, when the hypoxia intensify over the western India shelf, the pelagic and benthic fishes tend to the surface layers to avoid oxygen deficient waters. Thus controlling nutrient input from estuary is of important for achieving a healthy coastal environment.

****\$\$****

Dynamics of trace metals, rare earth elements and nutrients in the CE

Hydrography and nutrient fluxes

Dynamics of trace metals, rare earth elements and nutrients in the CE

Chapter **3**

FRACTIONATION OF TRACE METALS IN SEDIMENT

- 3.1 Introduction
- 3.2 Sampling Strategy
- 3.3 Analytical Methodology
- 3.4 Results
- 3.5 Discussion
- 3.6 Summary and Conclusion

3.1 INTRODUCTION

Over the past five decades, there has been a rapid increase in the human settlement along the coastal regions globally. As a consequence, many coastal and estuarine systems have been severely affected by encroachment and environmental deterioration. Trace metal contamination was one of the important factors that have lead to widespread environmental contamination (Nriagu and Pacyna, 1988). The behavior of trace metals in aquatic medium depends on the environmental conditions, which in turn, activate their speciation and reactivity (Bruland and Lohan, 2003). It is already established that sediments are the ultimate sink of all trace metals entering an aquatic system and is a reliable indicator of natural process and anthropogenic

CHAPTER 3

activities (Forstner, 1981; Szefer, 2002). Therefore, quantification of trace metal input is an essential requirement for environmental monitoring.

Sediment is a complex heterogenous matrix comprising minerals and organic matter. The minerals present are usually weathering rock fragments and secondary minerals such as phyllosilicates or clay minerals, oxides of Fe, Al, and Mn and sometimes carbonates (usually CaCO₃). The organic matter may comprise of living organisms (mesofauna and microorganisms), dead plant/plankton materials (litter) and humic substances. The mobility of trace metals in sediments is controlled by the prevailing biogeochemical processes influenced by early digenesis such as sorption, precipitation, dissolution, advection, diffusion, bio-irrigation and bio-turbation etc. The organic matter facilitates early diagenesis (Rullkotter, 2002) and impact chemical speciation of metals by complexation (Luther III et al., 2001; Louis et al., 2009; Dang et al., 2014). At sediment-water interface, reductive dissolution of Fe-Mn oxyhydroxides facilitating remobilization of initially bound trace metals and scavenging of metals by newly formed Fe-Mn oxyhydroxides are the major processes (Gao et al., 2009; Scholz and Neumann, 2007).

Concentration of trace metal in sediment is an indicator of elemental contamination of a region; though it does not help to identify its bioavailability (Sundaray et al., 2011). This is because trace metals exist in different chemical forms within the sediments with different mobility and toxicity. Since the bioavailability of a metal ion depends on its chemical form in aqueous medium (Jain et al., 2004), understanding the mechanism of its geochemical distribution and fractionation is essential to minimize its pollution potential in estuarine and coastal area.

64

Cochin estuary (CE), located along the south west of India, is composed of a complex system of shallow network of canals spread over more than 256 km² (Qasim, 2003). The estuary is micro-tidal (amplitude \sim 1m) and the seasonal salinity distribution is largely influenced by the seawater intrusion and freshwater runoff (Revichandran et al., 2011). Depending upon the degree of mixing between the freshwater and saltwater, there is a progressive seasonal shift in the estuarine hydrography from a stratified (monsoon) to a partially mixed (post-monsoon) and well-mixed (pre-monsoon) conditions (Sankaranarayanan et al., 1986; Shiyaprasad et al., 2013). The bathymetry varies considerably from 10-13 m in the shipping channels to approximately 2-7 m in other parts of the estuary (Menon et al., 2000). The region annually receives about 320 cm rain, of which, nearly 75% occurs during June to September.

There are mainly six rivers debouching ~ $2 \times 10^9 \text{m}^3$ of freshwater into the estuary annually (Srinivas, 2000). Approximately, 33 % of the total fresh water discharge is through the River Periyar, which is the life line of the estuary. The estuary exhibit wide variation in its flushing time from ~ 1-2 days during monsoon to ~ 8-9 days during pre monsoon (Vinita et al., 2015a). The low lying swamps and tidal creeks with patches of mangroves support the breeding and growth of larvae and juveniles of many economically important species in the estuary (Jyothibabu et al., 2006). The estuary is known as a nursery ground of commercially important prawns and fishes (Gopalan et al., 1983). The fields around the estuary are suitable for aquaculture, as they support traditional, seasonal and perennial prawn fishery.

Dynamics of trace metals, rare earth elements and nutrients in the CE

A systematic change over the last five decades like shrinkage of the estuary due to reclamation (Gopalan et al., 1983) and increased discharge of pollutants have made an adverse impact on the environmental degradation of the estuary (Qasim and Madhupratap, 1979; Menon et al., 2000). Approximately 70% of Kerala's industries (fertilizer, pesticide, radio-active mineral processing, chemical and allied industries, petroleum refining and trace metal process and fish processing) are located along the banks of River Periyar.

It is estimated that nearly 0.104Mm³ of industrial and 260 tons of domestic wastes reach the Periyar River daily from the Kochi industrial belt (Balanchandran et al., 2006). The effluents from these industries contain organics, alcohols, ammonia, nitrates, phosphorous, trace metals such as Cd, Hg, Cr, Zn, Pb, Cu and rare earth element products, suspended solids, radiologicals, chlorides of metals, pesticides etc (Manju and Sujatha, 2012). A long term assessment of trace elements in the northern estuary (Balachandran et al., 2005) remarked an expanse of 18 times in Zn and 9 times in Cd level. There are so many reports showing the enrichment of Pb, Cd and Zn in the estuarine sediments (Balachandran et al., 2002: 2005: 2006; Kumar et al., 2010; Akhil et al, 2014; Manju and Sujatha, 2012; George et al., 2016; Renjith et al., 2016; Salas et al., 2017).

The present study was mainly focused on the Cochin Estuary and the adjoining coastal marine environment (Fig 3.1), which experiences two distinct regimes of alluvial loading: an export of sediment from the estuary during the dynamic season (monsoon) followed by silting environment during the remaining period (Nair et al., 1993). Geochemical studies have also shown

that there is a shift in the sediment characteristics from kaolinite dominant rivers and estuary to a fine-grained montmorrilonite on the coastal shelf (Nath et al., 2000). The montmorrilonite are thus, responsible for a greater accumulation of organic matter in the coastal environment due to their large surface area. The organic matter input to the Cochin estuary is primarily from terrigenous and lateral sources during monsoon; though there may be a contribution from the sea during the post and pre-monsoon periods (Gireeshkumar et al., 2013a).



Fig 3.1: Map of the study area showing locations of sampling sites

Trace metal accumulation in the sediments of Cochin estuary is influenced by the intense anthropogenic activities, complex morphology (Manju et al., 2013) and weak flow (Balachandran et al., 2005; Thomson et al., 2002). This has eventually reduced the benthic biodiversity and increased CHAPTER 3

the abundance of less tolerant species in ecologically sensitive areas in the estuary (Udayakumar et al., 2014; Martin et al., 2011). The diversity of the bio-resources in the estuary has a direct bearing on the adjacent coastal ecosystem as well (Priju et al., 2007). There has also been an enrichment of trace metals in coastal sediments during monsoon due to terrestrial loading (Balachandran et al., 2003; Manju et al., 2013; Sreekanth et al., 2015). The present study attempts to quantify the fractions of trace metals in Cochin estuary and its coastal sediments to identify the mechanism of trace metal accumulation and their mobility.

3.2 SAMPLING STRATEGY

The geographic positions of sampling stations are given in Table 3.1. In order to understand trace metal loading to the estuarine regions and its subsequent effect on coastal region, surface sediments were collected from 8 locations in the north and central estuary, in addition to the adjacent coastal stations at a depth of 5.3, 6.8, 27 and 30 m. The coastal stations at ~5 and ~27 m were located north of the Cochin inlet and stations at ~6.5 and ~30 m were located south of the Cochin inlet. The sampling was carried out during October 2011. Water sample from each locations were collected using a 5L Niskin sampler (Hydrobios Kiel). The sample for dissolved oxygen (DO) was collected without air bubbles in a glass bottle, fixed onboard and later analyzed according to Winkler's method (Grasshoff et al., 1983). Nutrients were analysed following standard procedures (Grasshoff et al., 1983). Salinity was measured using a salinometer (DIGI Auto, Model 5, Tsurumi-Seiki). Sediments were collected using a Van-veen stainless steel grab (0.042 m²), sealed in plastic bottles, kept in ice, transported to the laboratory and stored at -20° C until analysis.

SI No	Sampling Locations	St. code	Lat (°N)	Long(°E)
1	Moolampilly	\mathbf{S}_1	10.0312	76.2666
2	Vallarpadam north	\mathbf{S}_2	10.0174	76.2442
3	Bolghatty	S ₃	9.9913	76.2680
4	Vallarpadam-Bolghatty bridge	\mathbf{S}_4	9.9891	76.2607
5	Vallarpaadam-Vypeen bridge	S_5	9.9823	76.2445
6	Cochin inlet (10-15m)	S_6	9.9702	76.2415
7	Ernakulam Channel	\mathbf{S}_7	9.9454	76.2880
8	Mattanchery channel	S_8	9.9341	76.2688
9	Near shore (5.3m)	S ₉	9.9734	76.1870
10	Near shore (6.8m)	S_{10}	9.9293	76.2264
11	Off shore (27m)	S ₁₁	10.004	76.0833
12	Offshore (30m)	S ₁₂	9.9167	76.0833

Table 3.1: Station locations in Cochin estuary and adjacent coastal region

3.3 ANALYTICAL METHODOLOGY

3.3.1 General sediment characteristics

The sediment samples were portioned into three for the analysis of total metals, organic carbon and grain size. For grain size analysis, the inorganic carbonates were removed using 10% HCl and organic matter using H_2O_2 (Folk, 1980). For texture analysis, the sediments were washed with distilled water, dried (at 30°C) and treated with sodium hexametaphosphate to disperse the sediments and the particles were separated following the pipette analysis (Krumbein and Pettijohn, 1938). Sediment organic carbon (OC) was analyzed by wet chromic acid oxidation method suggested by El Wakeel and Riley, 1957. The amount of total organic matter (TOM) was computed by multiplying the OC values with 1.80 (Muller et al., 1986).

3.3.2 Trace metal analysis

For the analysis of trace metals, the sediment samples and rock standards (MAG 1) was dried (at 60°C), powdered (mortar and pestle) and acid digested with a mixture of suprapure HF+HNO₃+HClO₄ (7:3:1) in Teflon vessels on a hot plate to dryness (Balaram and Rao, 2002).

		Standard Re					
Sl No	Element	Certified value (mg/kg)	Observed value (mg/kg)	RSD (%)	PAAS		
1	Zn	130 ± 6	130 ± 0.265	0.20	85		
2	Mn	710	759 ± 3.563	0.47	850		
3	Cr	97 ± 8	97 ± 0.562	0.58	110		
4	Ni	53 ± 8	53 ± 0.543	1.02	55		
5	Cu	30 ± 3	30 ± 0.071	0.24	50		
6	Со	20 ± 1.6	20.4 ± 0.164	0.82	23		
7	V	140 ± 6	140 ± 1.389	0.99	150		
8	Cd	0.2 ± 0.03	0.17 ± 0.005	2.50	-		
9	Sn	3.6	3.8 ± 0.053	1.47	-		
10	Pb	24 ± 3	24 ± 0.343	1.43	20		
11	Zr	130 ± 13	126 ± 0.512	0.39	210		
12	Nb	12	12 ±0.037	0.31	19		
13	Hf	3.7 ± 0.5	3.7 ± 0.036	0.97	5		

Table 3.2: Trace metal concentrations (mg/kg) in Standard reference material(MAG 1) and RSD (%) in analysis and are given below.

*Standard reference material (MAG 1) was obtained from Geological survey of India (GSI). *PAAS is the Global Shale Value used for normalisation of trace metals.

The residues were re-dissolved in 2% ultrapure HNO₃ (containing 1 ppm Rhodium as internal standard) and analyzed using an Inductively Coupled Plasma-Mass Spectrometer (ICP-MS). The Certified Reference Material (MAG 1) was obtained from Geological Survey of India to validate the method (Table 3.2). Precision and accuracy of the analysis was checked by

the replicate analysis of the certified reference material (MAG-1) obtained from Geological survey of India. % error of the analysis ranged from 0.2 to 7.40% for MAG-1 reference material. The relative standard deviation, RSD (standard deviation/ mean of triplicate samples) ranged from 0.204 to 2.50% for MAG-1 reference material.

3.3.3 Metal fractionation

Fractionation of sediment metals was achieved through sequential extraction procedure (SEP) suggested by Tessier et al., 1979 (Fig 3.2). In brief, sediment trace metals can be progressively extracted in to different fractions based on their relative mobility in different solvents. The different extracts contain the metals as exchangeable, hydrous oxide bound, organic bound and lattice bound fractions. During the extractions, the metals of anthropogenic origin normally tend to concentrate in the first three fractions, while the metals bound to residual (fourth) fractions are of natural origin or lithogenous. The exchangeable metal fractions (Fr 1) can be removed by changing the ionic strength, which allow leaching of metals sorbed to the surface of sediment. The carbonate bound fractions are susceptible to pH variation and extraction of sediment using1M ammonium acetate (pH 7) can yield the above two fractions. The metals bound to Fe-Mn oxide (Fr 2) can be extracted using 0.5 M NH₂OH.HCl. The metals bound to organic phase (Fr 3) can be removed through oxidation for which, H₂O₂ and 3.2M ammonium acetate (acidified to pH 2) can be used. The residual fraction (Fr 4) consists of metals strongly incorporated into the primary and secondary minerals. This fraction is the hardest to remove and requires the use of strong acids (HF+HNO₃+HClO₄) to break down the silicate structures. For the present study, all the extraction procedure was performed in a clean environment.



Fig 3.2: Schematic representation of metal fractionation in sediments (Tessier et al., 1979).

3.3.4 Clay mineralogy

Mineralogical composition of the clay fractions (< 2μ) in sediments were analyzed using a XRD. The sample preparations for XRD measurement are depicted in Fig 3.3. The diffractograms were obtained using Cu/40kv/20 ma as X-ray source. The XRD patterns were recorded with intensity (cps) vs 2θ distinguished from 3.0000-35.0000 in continuous scan mode with speed of 1.5000 deg/min. In order to facilitate XRD analysis, sediments (< 2μ) were washed with Milli-Q water to remove salt and confirmed with silver nitrate test for chloride. The samples were washed by dilute HCl to remove inorganic carbon and then treated with 50% H₂O₂ to remove organic carbon. They were finally washed with Milli-Q water to remove the remaining acid, dried (50°C), cooled in desiccators, homogenized to fine powder and mounted on XRD for measurement. The minerals were identified and quantified from X-ray diffractograms (Appendix 2b).



Fig 3.3: The sample preparations for XRD

3.3.5. Data processing

3.3.5.1. Normalisation factor

The significance of normalisation is to check the similarity of pattern for many elements supporting homogeneity (Taylor and McLennan, 1985), which is caused by the mixing of materials derived from different sources, with different age and rock types, during weathering, transport and deposition (Dupre et al., 1996; Nath et al., 2000). The elemental concentrations are normalized with those of Post Archean Australian Shale reference material (PAAS) for plotting shale normalized pattern.

3.3.5.2. Risk Assessment code

Based on the Tessier fractionation, the risk assessment code (RAC) was used to evaluate the potential risk due to each metal. The RAC assessment was expressed as the percent of Fr 1 in the total metal. The criteria governing NF and the RAC are summarized in Table 3.3.

Table	3.3:	Criteria	for	the	pollution:	Normalisation	Factor	(NF)	and	Risk
Asses	sment	t Code (F	RAC).						

No	ormalisati	on Factor (NF)	Risk Asse	ssment C	code (RAC)
NF	Class	Pollution status	Fr.1 (%)	Class	Risk
< 1	0	Low	< 1	1	No risk
1 - 3	1	Moderate	1-10	2	Low risk
3 - 6	2	Severe	11-30	3	Moderate risk
> 6	3	Extreme	31-50	4	High risk
			> 50	5	Very high risk

3.3.5.3 Canadian sediment quality guidelines for the protection of aquatic life

As the sediment quality guidelines are not available for Indian ecosystems, Canadian sediment quality guidelines were used as a tool to compare with the functioning of healthy aquatic ecosystems (CCME, 2001). These criteria are based on several field research studies demonstrating the interaction between chemicals and aquatic organisms to establish a cause and effect relationship for that particular organism. Comparison of measured concentrations of various contaminants following the above guidelines will provide a probable degree of contamination and its impact on the ecology (Table 3.4). The guidelines are based on the two stages, threshold effect levels (TEL) and probable effect levels (PEL). The TEL and PEL are used to identify the following three ranges of chemical concentrations with regard to biological effects. It is likely that the TEL will be adopted as the interim marine sediment quality guidelines ISQGs (CCME, 2001):

- < TEL Minimal effect range within which adverse effects rarely occur.
- TEL to PEL Possible effect range within which adverse effects occasionally occur.
- > PEL Probable effect range within which adverse effects frequently occur.

3.4 RESULTS

3.4.1 General characteristics of sediment in the study domain

3.4.1.1 Sediment texture

Textural parameters showed a significant spatial variation throughout the study domain (Fig 3.4). This was evidenced by the variation in the sand (1.56-70.38%), silt (5.04-45.98%) and clay (22.6-61.06%) fractions in the sediments. Among the 8 stations in the estuary, the sand content was highest at S_5 . Silt was greater at S_4 while clay showed high values at S_8 .



Fig 3.4: Sand, silt and clay fractions (%) in the sediments of Cochin estuary and adjacent coastal region during October 2011.

The estuary was dominated by clayey-sand and silty-clay texture with a progressive increase in the fine fractions from upper to the central estuary, which is in support with earlier studies (Reddy et al., 1992, Nath et al., 2000; Renjith et al., 2016). The sandy nature at S_5 (central estuary) was noteworthy.

Similar distribution during earlier studies during monsoon suggests that there is a possible down estuarine bed load movement (Menon et al., 2000). From our study, the Mattanchery channel (silty-clay) seems to be more clayey compared to Ernakulam channel (sandy-silt). The bar mouth and surrounding regions were sandy and low in clay and silt, possibly due to stronger flushing. The coastal sediments were dominated by silty-clay fraction (Fig 3.4). The near shore areas (~5m, ~7 m) showed higher clay content than offshore regions (~30 m).

3.4.1.2 Clay minerals

Clay minerals are a group of important hydrous aluminium phyllosilicates with a layer (sheet like) structure and very small particle size. They may contain significant amount of iron, alkali metals, or alkaline earths. Clay mineral constitutes a major fraction of sediments and their reactivity regulates the overall physic-chemical environment (Whitehouse et al., 1960). The clay mineral composition depends on the regional climate, provenance, rock type etc. The composition of clay minerals is an indicator of the sediment dispersal pattern in various environments (Grim, 1968). Information on clay minerals is essential to identify the origin, early diagenesis, chemical weathering and depositional behavior of the region.

The XRD diffratograms of the sediments from the estuarine and coastal stations clearly show prominent peaks corresponding to the minerals kaolinite and smectite whereas the peaks corresponding to illite, chlorite, gibbsite and quartz were less prominent (Fig 3.5). Thus, kaolinite was found to be the dominant clay mineral in the estuary and coastal region except at S_2 , S_9 and S_{11} , where smectite was the dominant fraction (Appendix 2b). These minerals generally reflect the terrestrial origin and subsequent weathering of

laterites. The laterite and red soils of the drainage basins are the major source of kaolinite, while black and peat soils over the hinterland are the source of smectite and illite (Rao et al., 1995). Previous studies have reported a gradual decrease in the kaolinite content concomitant to an increase in smectite towards the north of Cochin estuary. As smectite was mostly associated with fine grained sediments, the dominance of this fraction in the estuarine stations (S_2) may be attributed to the severe flow restrictions in the region (Balachandran et al., 2006). Greater surface area and smaller size of smectite allows strong bonding with organic matter, which undergo flocculation over a wide range of salinity (Degens and Ittekkot, 1984). Increased flocculation in the flow restricted northern region has been reported in CE (Balachandran et al., 2005). Since the salinity in this station varied narrowly, the flocculation was for a major reason for the dominance of smectite in this region. The high concentrations of smectite and strontium (Sr⁸⁷/Sr⁸⁶) ratio in the upstream of Mandovi and Zuari estuary are attributed to the extreme chemical alteration (Rao et al., 2015).



Fig 3.5: Clay minerals (%) in the sediments of Cochin estuary and adjacent coastal region (S_1-S_{12}) during October 2011

In the inner shelf (30 m), the sediment showed higher kaolinite content than the near shore region. One of the earlier studies has suggested that deposition of kaolinite along the inner shelf sediments might have taken place due to the lowering of sea level during the Halocene period (Reddy et al., 1992). However, in the absence of any marked variations in the source material (Reddy et al., 1992; Nath et al., 2000) the spatial variation in the abundance of kaolinite and smectite in the estuary and adjacent coastal stations can be assumed to be due to their transport differentiation and flocculation in the estuary.

3.4.1.3 Sediment organic carbon

Organic carbon content in the sediment varied from 6.53 to 36.52mg/g (dry weight, Fig 3.6). It was relatively high along the coastal stations (24.22 to 36.62mg/g) compared to the estuarine stations (12.7 to 25.81mg/g). This is because the sediment texture can significantly influence the accumulation of organic carbon.



Fig 3.6: Organic carbon (mg/g) in the sediments of Cochin estuary and adjacent coastal region during October 2011.

Dynamics of trace metals, rare earth elements and nutrients in the CE

It was observed that silty-clay sediments along stations S_2 , S_4 , S_8 , S_9 , S_{10} , S_{11} and S_{12} have high organic carbon while stations with sandy sediments (S_5 and S_6) showed low organic carbon. The organic carbon in the near shore sediments was much higher compared to that in offshore sediments, which may be due to higher productivity along the inner shelf. There are several reports showing higher productivity in the near shore area (Balachandran et al., 2008b; Madhu et al., 2011).

3.4.2 Distribution of trace metals and transition elements

The concentration of trace metals in estuarine sediments varied as Zn (112.5 - 435.7)mg/kg; Cr (84.2 - 183)mg/kg; Mn (298.5 - 1285)mg/kg; Pb (16.7 - 41.9)mg/kg; Cd (0.10 - 2.42)mg/kg; Ni (15.54 - 102.9)mg/kg; Co (8.13 - 21.66)mg/kg; Sn (4.35 - 7.54)mg/kg; Cu (27.23 - 65.7)mg/kg and V (45.6 - 152.1)mg/kg. The results show that Mn is the most abundant metal and Cd as the least abundant trace metal in estuarine sediment, with their maximum concentrations towards the north estuary stations (S₁, S₂, S₃) and minimum near the Cochin inlet (S₅, S₆). The transition elements also varied as: Zr (47.8 - 106.3)mg/kg; Hf (1.27 - 3.03)mg/kg and Nb (4.9 - 15.8)mg/kg.

The concentrations of trace metals in the coastal sediments were as follows: Zn (245.3 - 316.4) mg/kg; Cr (124.9 - 184.5) mg/kg; Mn (261.8-393) mg/kg; Pb (23.9 - 25.6) mg/kg; Cd (0.16 - 0.19) mg/kg; Ni (67.7 - 95.4) mg/kg; Co (15.2 -17.2) mg/kg; Sn (4.4 - 9.9)mg/kg; Cu (48.9 - 64.22) mg/kg and V (107 - 121.6)mg/kg. The trace metals showed near shore enrichment and a decrease towards deeper station (30 m), supporting the earlier reports (Laluraj, 2009). The transition elements (Zr, Hf and Nb) in coastal sediments also varied from (92.4 - 107.7)mg/kg; (2.7 - 3.03)mg/kg and (11.2 - 13.8)mg/kg respectively.

Dynamics of trace metals, rare earth elements and nutrients in the CE

3.4.3. Sediment Quality Guidelines and evaluation of pollution criteria

Geochemical normalization of an element is estimated as the ratio of concentration of the element in the sample to the concentration of the same element in the global shale or the primitive mantle of the earth (Nath et al., 2000). The pollution status of an environment can be evaluated based on the normalisation factors. Thus, the normalisation factor (NF) less than 1 refers to low contamination, NF < 3 refers to moderate contamination, NF > 3 indicates severe contamination and NF > 6 indicates extreme contamination. The shale normalized value of elements in both estuarine and coastal sediments are depicted in Fig 3.7.



Fig 3.7: Shale normalized pattern of trace metals showing the level of contamination in Cochin estuary and adjacent coastal region during October 2011

Based on the Fig 3.7, it can be seen that Zn is moderately contaminated near the Cochin inlet (S_6 , S_5 , S_7), whereas it is severely contaminated at all

other estuarine and coastal stations. Mn and V showed moderate contamination at S_2 while all other stations showed low contamination. Pb (except S_6) and Cr (except S_5 , S_6) showed moderate contamination for both estuarine and coastal stations. Ni and Cu showed moderate contamination for all the stations except S_5 , S_6 and S_7 , whereas Co, Zr, Nb and Hf showed low contamination for both estuarine and coastal stations.

The results of the trace metal contents in sediments from the present study were checked for their quality using standard sediment quality guidelines (Table 3.4). The concentration of Zn in S₅ and S₆ with sandy texture was < TEL shows that an adverse effect can rarely occur in these regions. The Zn at S₇, S₉ and S₁₂ were in the range TEL - PEL indicating that adverse effect can occur in these regions occasionally, whereas the stations S₁, S₂, S₃, S₄, S₈, S₁₀ and S₁₁ were showing values > PEL, which is an indication that these regions are prone to frequent adverse effect due to trace metal pollution.

Table 3.4 :	Sedime	ent q	uality	guidelines	for the p	protec	ction of a	quatic	life
	based	on	the	Canadian	Council	of	Minister	s of	the
	Enviro	nme	nt (CO	CME)					

Elements	Observed value	TEL	PEL	ERL	ERM
Cu	27.2 - 65.7	18.7	110	34	270
Zn	112.5 - 435.7	124	270	150	410
Cr	84.2 - 184.5	52.3	160	81	370
Cd	0.10 -2.42	0.68	4.2	1.2	9.6
Pb	16.7-41.9	30.2	110	46.7	218
Ni	15.54-102.9	15.9	43	20.9	51.6
Mn	261.8 - 1285	460	na	na	na

*na = no available data; ERL= effect low range; ERM = effect medium range; TEL = threshold effect level; PEL = probable effect level.

The concentrations of Pb in S_1 , S_2 , S_3 and S_4 covering northern and central estuary are between TEL and PEL indicating the adverse effect due to this metal to be occasional. Other stations with low Pb content indicate less contamination. Cr concentration at all stations were > TEL, while it was > PEL at stations S_2 , S_4 , S_8 and S_{10} , S_{11} and S_{12} showing that adverse effects due to Cr can occur frequently in these regions. The Cd was > TEL at S_1 , S_3 and S_4 , indicating that the adverse effect of Cd contamination can occur intermittently in these regions. The Ni in sediment S_5 was below TEL while all other stations showed values higher than TEL and above PEL in most of the stations (except S_5 , S_6 and S_7). The Cu concentration in the studied region was between TEL and PEL highlighting the possibility of occasional adverse effects. The Mn showed concentration higher than TEL in north estuary S_1 , S_2 and S_3 , while all other stations contained Mn < TEL indicating that the chances of any adverse effect are very rare in these stations.

3.4.4 Fractionation of metals

3.4.4.1 Zinc

Distribution of Zinc in different fractions of estuarine and coastal sediments is presented in Fig 3.8. The organic phase contained the lowest fraction (Fr 3) of Zn (0.08 - 4.07% of total concentration) in both estuarine and coastal sediments except at S₇ and S₈, where Fr 3 forms the major non residual binding phase. However, Fr 3 (0.8 - 17.6mg/kg) showed an increase with the increasing Zn loading in estuarine sediments (r^2 =0.81, Table 3.5). Instead, the association of Zn with Fe/Mn oxy hydroxide (Fr 2) was significant (~0.8 to 29.8mg/kg, Fig 3.8). It was also seen that Fr 2, the concentration of Zn in Fe/Mn oxy hydroxide fraction (0.3 to 8.62 % of total Zn) gradually increased with the increasing Zn loading in the estuarine sediments (r^2 = 0.88, Table

CHAPTER 3

3.5), while such a linear trend was not observed for the coastal sediments ($r^2 = 0.24$, Table 3.6). This indicates that Fe/Mn oxy hydroxide phase could be a major binding phase for Zn deposition in the estuarine sediments.



Fig 3.8: Total Zn and its four different fractions in the sediments of Cochin estuary and adjacent coastal region on October 2011.

The highest fractions of non-residual Zn (0.6 to 25.1% of the total Zn) were found to associate with exchangeable and carbonate/bicarbonate phases (Fr 1) in the estuarine sediments (except $S_7 \& S_8$). This fraction also increased towards the north estuary to peak at S_1 . However, most of the non-residual Zn was associated with Fe/Mn oxy hydroxide phase (Fr 2) towards the coastal

environment. Fr 1 (2.1-109.6mg/kg) also did not increase with the increasing Zn loading, indicating that adequate Fe-Mn oxide binding sites were already available for its association in the sediment. The residual fraction (Fr 4) was the major fraction of the total Zn (\sim 103.4 to 330.5 mg/kg, Figure 3.8) in both the environments. The Zn concentration in residual fraction was even higher than its concentrations in the adjacent source rocks of Granulitic terrain (Nath et al., 2000).

3.4.4.2 Chromium

Distributions of Cr in different fractions in the estuarine and coastal sediments (Fig 3.9) showed exchangeable and carbonate/bicarbonate phases (Fr 1) as the poorest fractions (0.1-1.5% of total Cr). Table 3.7 shows that the absolute concentration of Cr (0.08 to 2.53mg/kg) in Fr 1 did not gradually increase with the increasing Cr loading in the estuarine sediments (similar to Zn) ($r^2 = 0.14$) whereas Fr 1 showed a gradual increase with increasing Cr loading towards the coast ($r^2 = 0.75$, Table 3.8). Concentration of Cr in Fr 2 (Fe/Mn oxy hydroxide) was higher than Fr 1 and ranged from 1.1 to 3.02 % of total Cr in sediments. Table 3.7 shows that Fr 2 (1.68 - 3.64mg/kg) did not gradually increase with the increasing Cr loading in estuarine sediments ($r^2 = 0.25$) however Fr 2 gradually decreased with the increasing Cr loading in the coastal sediments ($r^2 = -0.61$; Table 3.8). This is probably because Cr forms thermodynamically weak complexes with Fr 1 and Fr 2.

The highest factions of non-residual Cr were found to associate with organic binding phases (Fr 3) and found to be the preeminent non residual binding phase for Cr (1.31 to 17.4%) in the entire domain. Fr 3 (2.17-21.92 mg/kg) showed a gradual increase with increase in total Cr in the sediments

 $(r_{\text{estuary}}^2 = 0.75; r_{\text{coast}}^2 = 0.57)$. The Cr in Fr 3 showed strong linear correlation with organic matter ($r^2 = 0.72$) in the estuary and a weak correlation in the coast ($r^2 = -0.28$).



Fig 3.9: Total Cr and its four different fractions in the sediments of Cochin estuary and adjacent coastal region on October 2011.

The geochemical fractionation study generally reveals that a major part of Cr in the sediments is present as lithogenous (Fr 4) fraction (50.0-173.9mg/kg) derived from the crustal sources of felsic composition. It is observed that the concentrations of Cr in the residual fraction gradually increased with the increasing total Cr loading (Table 3.7 & Table 3.8) in sediments.

3.4.4.3 Cadmium

Cadmium showed wide variation in its affinity towards different fractions (Fig 3.10). The dominant fraction (6.8-49% of total Cd) was associated with exchangeable and carbonate/bicarbonate phases (Fr 1) at S_1 (north estuary) and S_5 (central estuary). This fraction is highly toxic, since a change in the environmental conditions (ionic concentration, pH) will make it available to organisms. When with the salinity increases, this fraction will be released to the water. Similarly, when the pH decreases also, this fraction will be released to water column and become available to biota. Cd bound to Fr 1 in S_2 and S_4 were higher than residual fraction.

Fe/Mn oxy hydroxide (Fr 2) was the dominant non-residual fraction of Cd (8.2 to 40.4% of total Cd) associated with the sediments at S₂, S₃, S₄ and S₇ (north and central estuary). The concentration of Cd in Fr 1 (0.01 to 0.97mg/kg) and Fr 2 (0.01 to 0.87mg/kg) gradually increase with the increasing Cd loading in sediments. The lowest fraction of the total concentration of Cd (vary from 2.9 to 25.46 % of the total Cd) was present as organic binding phases (Fr 3) except at coastal region south of inlet (S₁₂) where Fr 3 forms the major binding phase (42.9%). The absolute concentration of Cd in Fr 3 showed a gradual increase with increase in total Cd in sediments ($r_{estuary}^2$ =0.84; r_{coast}^2 =0.73) (Table 3.7 & Table 3.8).

The major part of the total Cd in the sediments of S_3 , S_6 , S_7 , S_8 , S_9 , S_{10} and S_{11} was present within the structure of the sediments (Fr 4) (0.04 to
1.16mg/kg) and suggests that Cd entered through anthopogenic input get deposited and incorporated into the crystal lattice of minerals in these sediments.



Fig 3.10: Total Cd and its four different fractions in the sediments of Cochin estuary and adjacent coastal stations on October 2011.

3.4.4.4 Lead

Distribution of Pb in different fraction of sediments is presented in (Fig 3.11). It is observed that the lowest concentration of Pb (~0.5 to 6.0% of total Pb) was present as soluble, exchangeable and carbonate/bicarbonate bound fraction (Fr 1). The results from the present study show that Pb in Fr 1 (0.1 to 2.4mg/kg) did not increase with increasing Pb concentration in sediments

(Table 3.5 & 3.6). The organic bound fraction (Fr 3) was higher than Fr 1 and varied from ~ 3.8 to 17.8% of total Pb.



Fig 3.11: Total Pb and its four different fractions in the sediments of Cochin estuary and adjacent coastal stations on October 2011

Organic binding phase was the major non residual fraction of Pb in the estuarine stations S_2 , S_3 , S_4 , S_6 and S_8 with high organic matter. Pb in Fr.3 showed a gradual increase with increasing organic matter in estuarine sediments (r^2 =0.78, Table 3.5). However Fr 3 (0.9 to 6.1 mg/kg) did not show any increase with increasing total Pb loading in the sediments. Concentration of Fr 2 was higher than the Fr 1 and varied from ~ 7.6 to 18.6% of total Pb.

Metals bound to Fe-Mn oxyhydrodide was the (1.8 to 7.8mg/kg) was most dominant in the estuarine stations S_1 , S_5 and S_7 and coastal stations S_9 , S_{10} , S_{11} , S_{12} . Even though organic matter was higher in coastal sediments than estuarine sediments, Pb was preferentially associated with Fe/Mn oxy hydroxide phase than organic bound phase in the coastal sediments. The residual fraction of Pb (Fr. 4, 10.8 to 31.3 mg/kg) linearly increased with total Pb in sediments ($r^2_{estuary} = 0.95$; $r^2_{coast} = 0.96$, Table 3.5 & 3.6).

3.4.4.5 Nickel

Distribution of Ni in different binding phases of the studied sediments is presented in Fig 3.12. The relative affinity of Ni for different non-residual binding phases in the sediments was in the order of organic carbon binding phase > Fe/Mn oxyhydroxide phase > exchangeable carbonate/bicarbonate phases. Ni in Fr 1, Fr 2 and Fr 3 varied as 0.73 to 2.19mg/kg; 0.44 to 3.28mg/kg and 0.91 to 10.5mg/kg respectively. It is observed that Ni in Fr 3 gradually increased with increase in Ni loading (r^2 =0.84) in estuarine sediments (Table 3.7). However, Ni in Fr 2 gradually increased with increase in Ni loading (r^2 =0.98) in coastal sediments (Table 3.8) as it forms stable association with Fe/Mnoxyhydroxide phase in coastal sediments.

Most of the non residual fraction of Ni was bound to organic phase indicates that Ni forms thermodynamically stable complexes with organic matter in estuarine and coastal sediments. However the weak correlation of Ni in Fr 3 with organic matter in coastal sediments can be due to the difference in the nature of organic matter present. The association of Ni with sulphides can also make the difference as metal bound to sulphides were also extracted with Fr 3. A major part of the total Ni in the sediments was present within the structure of the sediments (Fr.4) (10.9 to 92.1mg/kg). The concentration of Ni (in the residual fraction) gradually increased with the total Ni concentrations in the sediments ($r^2_{estuary}$ = 0.99; r^2_{coast} = 0.99) (Table 3.7 & Table 3.8).



Fig 3.12: Total Ni and its four different fractions in the sediments of Cochin estuary and adjacent coastal stations on October 2011

3.4.4.6 Copper

Distribution of Cu in different binding phases of the studied sediments is presented in Fig 3.13. Concentration of Cu as Fe/Mn oxyhydroxide phase (Fr 2) and water-soluble, exchangeable, and carbonate/bicarbonate complexes

(Fr 1) were low and found to vary from (0.01 to 0.25%) and (0.0 to 0.6%) of the total Cu in the sediments. The Fr 1 was entirely absent in coastal sediments. The highest fractions of non-residual Cu (\sim 0.03 to 4.1% of the total Cu) were found to associate with organic binding phases in the sediments (Fr 3) (Fig 3.13).



Fig 3.13: Total Cu and its four different fractions in the sediments of Cochin estuary and adjacent coastal stations on October 2011

The concentration of Cu in Fr 3 (~0.02 to2.41 mg/kg) did not showed an increase with the increasing Cu loading in the sediments. The association of Cu with organic matter, Fr 3 increased with organic matter in coastal sediments, however such a linear inter relation was not observed in estuarine sediments (Table 3.9 & Table 3.10). The relative affinity of Cu for different non-residual binding phases in the sediments was in the order of organic carbon binding phase > exchangeable carbonate/bicarbonate phases > Fe/Mn oxyhydroxide phase. Major fractions of the total Cu (\sim 24.3 to 50.7mg/kg) were present within the structure of the sediments (Fr.4). Figure 3.13 shows the variation of residual fraction with changing total Cu content in the sediments. The range of Cu concentrations in residual fraction of the sediment was in close correspondence with the geology of the adjacent source rock (granulitic terrain) (Nath et al., 2000).

3.4.4.7 Manganese

Distribution of Mn in different binding phases of the sediments is shown in Fig 3.14. The relative affinity of Mn for different non-residual binding phases in the sediments was in the order of exchangeable carbonate/bicarbonate phases > Fe/Mn oxyhydroxide phase > organic carbon binding phase except in S_2 , where Fe-Mn oxyhydroxide phase found to be the preeminent binding phase in the sediments. The concentrations of organic phase of Mn (Fr 3) were low and found to vary from 2.01 to 7.6% of the total Mn content in the studied sediments. The fraction of total Mn associated with Fe/Mn oxyhydroxide binding phase (Fr.2) and exchangeable carbonate/bicarbonate phases (Fr. 1) was found to vary from 2.04 to 18.8% and 5.2 to 19.4% of total Mn in sediments.

Major portion of Mn in sediments was found to associate with the residual fraction (Fr 4) of sediments (56.03 to 84.6%). The absolute concentrations of Mn in Fr 4 vary from 189.8 to 720mg/kg. Mn concentration in Fr 1 (15.5 to 203.5 mg/kg), Fr.2 (6.1 to 233.6mg/kg) and Fr 3 (6 to

52.7mg/kg) were found to increase gradually with the increasing total Mn content in the estuarine sediments (r^2 0.96; r^2 0.97; r^2 0.88) (Table 3.5 & Table 3.6). However in coastal sediments Mn in Fr 1 and Fr 4 gradually increased with increased Mn loading in sediments.



Fig 3.14: Total Mn and its four different fractions in the sediments of Cochin estuary and adjacent coastal stations on October 2011.

3.4.4.8 Vanadium

The relative affinity of V for different non-residual binding phases in the sediments was in the order of Fe/Mn oxyhydroxide phase > organic carbon

binding phase > exchangeable carbonate/bicarbonate phases except in sediments of S_2 , where exchangeable, carbonate/bicarbonate phases (Fr 1) was the dominant non residual binding phase of vanadium (Fig 3.15). Vanadium in Fr 1, Fr 2 and Fr 3 was found to vary from 0.2 to 12.1mg/kg, 6.4 to 15.9mg/kg and 0.8 to 6.1mg/kg respectively. A major portion of Vanadium in sediments (38 to 123.9 mg/kg) was found to associate with sediment residual fraction which account for 73.9 to 91.8 % of total V in the sediments.



Fig 3.15: Total V and its four different fractions in the sediments of Cochin estuary and adjacent coastal stations on October 2011.

Dynamics of trace metals, rare earth elements and nutrients in the CE

3.4.4.9 Tin

Distribution of Sn in different binding phases of the studied sediments is presented in Fig 3.16. It shows that the lowest fractions of the total concentration of Sn (vary from ~0.0 to 18.7% of the total Sn) were present as water soluble, exchangeable, and carbonate/bicarbonate-Sn complexes (Fr 1) in the sediments. However at S_{12} , Fr 1 was the dominant non-residual fraction of Sn.



Fig 3.16: Total Sn and its four different fractions in the sediments of Cochin estuary and adjacent coastal stations on October 2011.

The Sn in Fe-Mn oxyhydroxide binding phase (Fr 2) was higher than in Fr 1 and varies from 2.01 to 29.05% of total Sn in the sediment. The absolute concentration of Sn in Fr 1 (non-detectable to 1.18mg/kg) and Fr.2 (0.11 to 2.75mg/kg) did not showed a gradual increase with increasing Sn loadings in sediments. This indicates Sn forms thermodynamically weaker association in Fr 1 and Fr 2 than other binding sites in sediments. The highest fractions of non-residual Sn were found to associate with organic binding phases (3.9 to 62.5% of total Sn) in the sediments (Fr 3) and probably the finest binding phase for Sn in the studied sediments. However the absolute concentration of Sn did not showed significant correlation with organic matter in sediments. This showed that nature of organic matter is a great factor influencing the binding of Sn to organic phases. Major part of the Sn in the sediments (23.3 to 80.6 % of total Sn) was present in the residual fraction except in S₂ and S₁₀, where most of the Sn was in organic binding phase.

3.4.4.10 Cobalt

Distribution of Co in different binding phases of the studied sediments is presented in Fig 3.17. The relative affinity of Co for different non-residual binding phases in the sediments was in the order of organic binding phase (Fr 3) > Fe/Mn oxyhydroxide phase (Fr.2) > exchangeable carbonate/bicarbonate phases (Fr 1) except in S₉, where Fr.1 was the major non residual binding phase.

The concentration of Co in Fr 1, Fr 2 and Fr 3 varied as (0.53 to 1.84)mg/kg, (0.34 to 1.95)mg/kg and (0.65 to 2.95)mg/kg respectively and this together accounts for only 13 to 27.2% of total Co. Major part of the total Co in the sediments was associated with residual fraction (Fr.4) in all the sediments and varied from 6.51 to 16.36mg/kg.



Fig 3.17: Total Co and its four different fractions in the sediments of Cochin estuary and adjacent coastal stations on October 2011

3.4.4.11 Zirconium, Niobium and Hafnium

The relative affinity of Zr, Nb, and Hf for different non-residual binding phases in the sediments was in the order of organic matter binding phase > Fe/Mn oxyhydroxide phase > exchangeable carbonate/bicarbonate phases. However the order slightly changed for Hf in coastal sediments with more affinity towards Fe/Mn oxyhydroxide phases (Fig 3.18 - Fig 3.20). For Nb, affinity towards different binding phases varied spatially. Major part of the Zr (70.7 to 99.7 % of total Zr), Nb (80.2 to 98.2 % of total Nb) and Hf

(66.8 to 99.2 % of total Zr) in the sediments were present in the residual fraction which was in close correspondence to the geology of the adjacent source rock and are stable in the sedimentary environment.



Fig 3.18: Total Zr and its four different fractions in the sediments of Cochin estuary and adjacent coastal stations on October 2011



Fig 3.19: Total Nb and its four different fractions in the sediments of Cochin estuary and adjacent coastal stations on October 2011.





Fig 3.20: Total Hf and its four different fractions in the sediments of Cochin estuary and adjacent coastal stations on October 2011



texture	and or	ganic	carbo	n in e	stuari	ne sec	limen	ts											
Variable	Mn1	Mn2	Mn3	Mn4	TMn	Pb1	Pb2	Pb3	Pb4	TPb	Zn1	Zn2	Zn3	Zn4	TZn	Sand	Silt	Clay	00
Mn1	1.00																		
Mn2	0.99	1.00																	
Mn3	0.91	0.87	1.00																
Mn4	-0.43	-0.47	-0.13	1.00															
TMn	96.0	0.98	0.88	-0.30	1.00														
Pb1	96.0	0.98	0.82	-0.41	0.99	1.00													
Pb2	-0.05	-0.18	0.23	0.73	-0.07	-0.17	1.00												
Pb3	0.10	-0.04	0.34	0.50	0.00	-0.11	0.88	1.00											
Pb4	0.68	0.62	0.79	0.16	0.70	0.64	0.48	0.43	1.00										
qdT	0.57	0.48	0.75	0.33	0.57	0.48	0.71	0.67	0.95	1.00									
Zn1	-0.01	-0.12	0.21	0.67	0.01	-0.06	0.89	0.66	09.0	0.74	1.00								
Zn2	0.30	0.17	0.49	0.49	0.25	0.15	0.86	0.86	0.77	0.91	0.77	1.00							
Zn3	0.27	0.15	0.50	0.45	0.24	0.16	0.86	0.70	0.78	0.89	0.87	0.88	1.00						
Zn4	0.46	0.36	0.73	0.40	0.41	0.29	0.74	0.85	0.75	0.88	0.60	0.88	0.76	1.00					
TZn	0.43	0.32	0.71	0.36	0.37	0.26	0.74	0.83	0.75	0.88	0.69	0.84	0.81	96.0	1.00				
Sand	-0.43	-0.33	-0.59	0.22	-0.22	-0.16	-0.28	-0.57	-0.29	-0.39	-0.02	-0.41	-0.38	-0.66	-0.64	1.00			
Silt	0.22	0.12	0.43	-0.03	0.05	-0.01	0.38	0.52	0.11	0.26	0.05	0.32	0.40	0.53	0.49	-0.89	1.00		
Clay	0.54	0.47	0.60	-0.37	0.35	0.31	0.10	0.48	0.42	0.43	-0.02	0.40	0.25	0.62	0.63	-0.86	0.52	1.00	
0C	0.41	0.31	0.56	-0.08	0.23	0.15	0.38	0.74	0.33	0.46	0.16	0.52	0.35	0.76	0.76	-0.89	0.67	0.90	1.00
		Val	ues in	plod i	are dif	ferent .	from () with	a signi	ficanc	e leve	l alphc	1>0.05	2 XI	STAT	2014			

Dynamics of trace metals, rare earth elements and nutrients in the CE

102

Table and or	3.6 Co ganic (orrelatio	on ma in co	trix (F astal s	edime	n) for nts	trace 1	metal	(Mn, J	Pb and	d Zn) i	n diffe	erent f	ractio	ns, sec	liment	t textu	e	
Variable	Mn1	Mn2	Mn3	Mn4	TMn	Pb1	Pb2	Pb3	Pb4	TPb	Zn1	Zn2	Zn3	Zn4	TZn	Sand	Silt	Clay (ЭС
Mn1	1.00																		
Mn2	0.08	1.00																	
Mn3	0.08	0.98	1.00																
Mn4	0.94	-0.05	0.01	1.00															
TMn	96.0	0.04	0.07	0.99	1.00														
Pb1	-0.41	0.87	0.83	-0.54	-0.47	1.00													
Pb2	0.31	0.12	-0.05	-0.02	0.10	0.08	1.00												
Pb3	-0.40	0.56	0.67	-0.23	-0.26	0.61	-0.73	1.00											
Pb4	0.16	0.51	0.65	0.36	0.32	0.27	-0.71	0.83	1.00										
TPb	0.02	0.69	0.81	0.16	0.14	0.53	-0.61	0.91	96.0	1.00									
Zn1	-0.96	-0.36	-0.35	-0.86	-0.92	0.14	-0.35	0.23	-0.27	-0.20	1.00								
Zn2	-0.91	-0.49	-0.49	-0.82	-0.88	0.01	-0.25	0.07	-0.41	-0.35	0.99	1.00							
Zn3	-0.96	-0.34	-0.34	-0.88	-0.93	0.16	-0.30	0.21	-0.31	-0.22	1.00	66.0	1.00						
Zn4	0.51	06.0	0.88	0.36	0.46	0.57	0.26	0.30	0.50	09.0	-0.73	-0.82	-0.72	1.00					
TZn	-0.46	0.51	0.38	-0.73	-0.62	0.77	0.57	0.02	-0.40	-0.13	0.27	0.24	0.32	0.25	1.00				
Sand	-0.06	0.94	0.98	-0.08	-0.03	0.84	-0.24	0.81	0.73	0.89	-0.20	-0.36	-0.20	0.77	0.32	1.00			
Silt	-0.44	-0.90	-0.93	-0.37	-0.44	-0.58	-0.02	-0.49	-0.69	-0.77	0.66	0.78	0.66	-0.97	-0.12	-0.86	1.00		
Clay	0.86	0.59	0.58	0.75	0.82	0.11	0.29	-0.01	0.42	0.40	-0.96	-0.99	-0.96	0.88	-0.13	0.44	-0.83	1.00	
0C	0.84	0.60	0.60	0.73	0.81	0.14	0.29	0.00	0.43	0.41	-0.96	-099	-0.96	0.89	-0.11	0.46	-0.85	1.00	1.00

Dynamics of trace metals, rare earth elements and nutrients in the CE

103

CHAPT	ER 3									1	⁷ ra	ctio	nat	ion	of t	rac	e m	ieta	ls ii	n see	diment
and	· 0C																		0	0 1.00	
kture a	Clay																		1.0	0.9	
ent tex	l Silt																	0 1.00	0.52	0.67	
edime	Sand																1.00	-0.89	-0.86	-0.89	
ns, se	INI															1.00	-0.91	0.65	0.95	0.95	
ractic	Ni4														1.00	1.00	-0.91	0.65	96.0	0.95	
rent f	Ni3													1.00	0.81	0.84	-0.67	0.45	0.73	0.69	
ı diffe	Ni2												1.00	0.78	0.67	0.70	-0.66	0.60	0.55	0.73	
Ni) in	Ni1											1.00	0.73	0.67	0.31	0.37	-0.32	0.43	0.12	0.28	
r and	TCr										1.00	0.36	0.68	0.83	66.0	66.0	-0.95	0.72	0.95	0.92	
Cd, C	Cr4									1.00	96.0	0.30	0.64	0.78	1.00	66.0	0.91	0.66	0.95	0.95	
etal ((Cr3								1.00	0.66	0.65	0.71	0.86	0.84	0.68	0.72	0.50 -	0.32	0.57	0.71	
ace m	r2 (00.	.73	.29 (.27 (. TT.	.70	.40	.30 (.33	.30 -(.43 (.08	.44	
for tra	1 C						00	20 1	.58 0	36 0	38 0	.66 0	30 0	.71 0	36 0	40 0	.14 -0	02 0	29 0	.17 0	
son) : ents	I Cr					0	4 1.	6 0.	4.0.	9 0.	1 0.	6 0.	4 0.	2 0.	2 0.	7 0.	-0.	9-0-	7 0.	7 0.	
(Pear edime	TC				0	6 1.6	9 0.1	6 0.7	2 0.7	8 0.1	8 0.2	9 0.5	9 0.7	0 0.4	1 0.2	3 0.2	4 -0.2	6 0.1	9 0.1	5 0.3	
natrix rine s	Cd4				1.0	. 0.5	-0.0	0.3	0.5	0.15	0.15	0.2	0.6	0.3	0.2	0.2	-0.1	0.0	0.1	0.3	
tion n estua	Cd3			1.00	0.09	0.84	0.22	0.65	0.59	0.27	0.31	0.49	0.54	0.37	0.29	0.33	-0.35	0.34	0.26	0.39	
orrelat on in	Cd2		1.00	96.0	0.27	0.95	0.19	0.73	0.67	0.20	0.24	0.54	0.62	0.39	0.23	0.28	-0.26	0.25	0.19	0.35	
.7 Co carbo	Cd1	1.00	0.91	0.81	0.23	0.89	0.24	0.75	0.63	0.00	0.02	0.55	0.50	0.34	0.03	0.08	0.02	0.03	-0.06	0.12	
Table 3 organic	Variable	Cd1	Cd2	Cd3	Cd4	TCd	Cr1	Cr2	Cr3	Cr4	TCr	Nil	Ni2	Ni3	Ni4	INi	Sand	Silt	Clay	0C	

104

Dynamics of trace metals, rare earth elements and nutrients in the CE

Variable	Cd1	Cd2	Cd3	Cd4	TCd	Cr1	Cr2	Cr3	Cr4	TCr	IIN	Ni2	SIN	Ni4	INI	Sand	Silt (Clay (OC
Cd1	1.00						-		-		-								
Cd2	06.0	1.00																	
Cd3	0.17	0.59	1.00																
Cd4	-0.37	-0.72	-0.95	1.00															
TCd	0.57	0.80	0.73	-0.66	1.00														
Cr1	0.81	0.63	-0.10	0.05	0.58	1.00													
Cr2	-0.43	-0.76	-0.91	1.00	-0.64	0.01	1.00												
Cr3	0.77	0.49	-0.34	0.25	0.34	0.97	0.20	1.00											
Cr4	0.77	0.82	0.40	-0.38	0.91	0.87	-0.39	0.71	1.00										
TCr	0.80	0.92	0.58	-0.60	0.95	0.75	-0.61	0.57	0.97	1.00									
Nil	-0.29	-0.67	-0.98	0.91	-0.85	-0.08	0.88	0.17	-0.56	-0.72	1.00								
Ni2	0.66	0.72	0.38	-0.31	0.91	0.84	-0.31	0.68	66.0	0.93	-0.55	1.00							
Ni3	0.34	0.28	-0.03	0.19	0.62	0.79	0.20	0.70	0.78	09.0	-0.16	0.86	1.00						
Ni4	0.74	0.78	0.37	-0.33	06.0	0.87	-0.34	0.71	1.00	0.95	-0.54	66 .0	0.82	1.00					
TNi	0.76	0.84	0.48	-0.46	0.94	0.82	-0.47	0.64	1.00	96.0	-0.64	96.0	0.74	66.0	1.00				
Sand	-0.23	-0.14	0.07	0.23	0.44	0.30	0.29	0.18	0.42	0.25	-0.18	0.56	0.82	0.47	0.40	1.00			
Silt	0.61	0.62	0.29	-0.58	0.07	0.04	-0.65	0.06	0.07	0.27	-0.23	-0.09	-0.53	0.01	0.10	-0.86	1.00		
Clay	-0.84	-0.94	-0.59	0.79	-0.60	-0.41	0.84	-0.30	-0.58	-0.75	0.62	-0.46	0.05	-0.53	-0.62	0.44	-0.83	1.00	
0C	-0.83	-0.94	-0.59	0.79	-0.59	-0.39	0.84	-0.28	-0.56	-0.73	0.61	-0.44	0.07	-0.51	-0.60	0.46	-0.85	1.00	1.00
]

VeriebleS13S14S14S14V1V3V3V4V1C10C10S14S14C10S14S1	UIBAIIIC	cal pull		maili			0													
8u1008u20461008u30280108u40100108u40100108u40100108u40100108u40100108u40100108u40100108u50100108u50100108u50100108u50100108u50100108u50100108u50100108u50100108u50100108u50100108u50100108u50100108u50100108u50100108u50100108u6011 <th>Variable</th> <th>Sn1</th> <th>Sn2</th> <th>Sn3</th> <th>Sn4</th> <th>TSn</th> <th>Ν</th> <th>V2</th> <th>V3</th> <th>V4</th> <th>ΛL</th> <th>Cu1</th> <th>Cu2</th> <th>Cu3</th> <th>Cu4</th> <th>TCu</th> <th>Sand</th> <th>Silt</th> <th>Clay</th> <th>0C</th>	Variable	Sn1	Sn2	Sn3	Sn4	TSn	Ν	V2	V3	V4	ΛL	Cu1	Cu2	Cu3	Cu4	TCu	Sand	Silt	Clay	0C
strong 046 100 strong 028 049 100 strong 028 039 100 strong 028 039 100 strong 028 038 038 030 100 strong 030 038 038 049 100 strong 031 032 038 038 039 030 100 strong 037 036 038 038 039 030 030 031 030 031 030 031 030 031	Sn1	1.00																		
8n3 028 004 100 8n4 0.57 0.10 0.40 100 8n4 0.57 0.10 0.38 0.40 100 1730 0.16 0.38 0.38 0.40 100 1730 0.88 0.53 0.48 100 $\cdot \cdot $	Sn2	0.46	1.00																	
Sub 0.57 0.10 0.40 1.00 TS 0.16 0.06 0.38 0.08 1.00 TS 0.16 0.06 0.38 0.98 1.00 V1 0.88 0.53 0.38 0.48 0.14 1.00 V2 0.16 0.16 0.16 0.12 0.26 0.21 1.00 V3 0.43 0.16 0.16 0.26 0.21 0.06 0.16 0.12 0.26 0.21 0.06 V3 0.05 0.16 0.16 0.21 0.26 0.21 0.06 0.21 0.07 0.21 1.01 V3 0.06 0.16 0.01 0.26 0.26 0.26 0.24 1.00 V4 0.25 0.11 0.02 0.21 0.26 0.24 0.24 0.24 1.00 V4 0.26 0.11 0.02 0.21 0.26 0.24 0.24 1.00 V4 0.25 0.11 0.02 0.21 0.24 0.24 1.00 V4 0.22 0.11 0.22 0.24 0.24 0.24 1.00 V4 0.24 0.24 0.24 0.24 0.24 0.24 1.00 V4 0.24 0.24 0.24 0.24 0.24 0.24 1.00 V4 0.25 0.24 0.24 0.24 0.24 0.24 0.24 1.00 V4 0.24 0.24 0.24 0.24 0.24 <	Sn3	0.28	0.04	1.00																
TS00.160.060.380.081.00V10.880.530.380.480.141.00V20.370.060.150.260.221.00V30.330.050.120.260.221.00V30.330.050.120.260.210.320.441.00V30.380.130.160.120.260.210.320.441.00V40.580.330.010.050.310.450.491.00V40.580.110.100.510.510.520.491.00V40.250.110.100.250.210.250.240.250.240.25V40.260.110.100.250.210.250.240.250.240.260.250.25V40.250.110.100.250.250.250.240.260.241.002.22V40.260.210.200.210.250.250.250.250.260.250.260.26V40.250.260.260.260.260.260.260.260.260.260.260.260.260.26V40.260.270.260.260.260.260.260.260.260.260.260.260.260.260.26V40.260.260.260.26	Sn4	-0.57	-0.19	-0.40	1.00															
V10.880.530.380.480.14100V20.870.060.160.120.260.231.00V30.430.150.120.260.210.050.441.00V40.580.150.150.260.210.530.491.00V40.580.330.050.510.530.491.00V40.580.310.010.590.580.510.590.541.00V40.580.170.010.590.580.510.591.00V40.580.130.100.590.580.510.500.511.00V60.220.110.050.230.240.250.241.002.22.2V60.290.110.050.230.240.250.241.002.22.2V70.210.110.050.230.240.250.241.002.22.2V70.230.240.260.240.250.240.250.242.22.2V60.240.250.250.260.260.260.240.260.242.22.2V70.230.240.260.250.250.250.240.260.242.22.22.2V70.230.240.260.260.260.260.260.260.260.26<	TSn	-0.16	-0.06	-0.38	0.08	1.00														
V20.370.060.160.120.260.221.002.10.320.441.00V30.430.150.150.120.260.210.320.441.00V40.580.330.050.140.020.510.500.510.500.510.500.510.530.541.00V40.580.330.040.010.500.510.530.541.002.242.240.530.541.00V100.050.110.050.230.240.250.240.250.241.002.242.242.260.250.241.00V110.040.220.110.050.310.320.341.002.241.002.242.242.252.242.252.242.262.242.262.242.262.242.262.242.262.242.262.242.262.242.262.242.262.242.262.242.262.242.262.262.242.26 </th <th>Ŋ</th> <td>0.88</td> <td>0.53</td> <td>0.38</td> <td>-0.48</td> <td>0.14</td> <td>1.00</td> <td></td>	Ŋ	0.88	0.53	0.38	-0.48	0.14	1.00													
V3 0.43 0.15 0.12 0.26 0.21 0.32 0.44 1.00 V4 0.58 0.33 0.05 0.14 0.02 0.51 0.65 0.49 1.00 V1 0.66 0.42 0.01 0.23 0.24 0.26 0.29 1.00 V1 0.06 0.42 0.01 0.59 0.68 0.21 0.99 1.00 V1 0.09 0.23 0.11 0.01 0.58 0.21 0.83 0.24 0.26 0.20 0.41 1.00 Cu2 0.10 0.02 0.11 0.02 0.23 0.24 0.26 0.21 0.29 0.24 1.00 Cu3 0.01 0.02 0.11 0.02 0.21 0.28 0.24 0.26 0.21 0.21 0.21 0.21 0.22 Cu3 0.21 0.02 0.21 0.28 0.24 0.26 0.21 0.21 0.21 0.22 0.21 0.22 Cu3 0.21 0.02 0.21 0.23 0.24 0.26 0.21 0.21 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 0.22 0.21 <th>V2</th> <td>0.37</td> <td>-0.06</td> <td>0.16</td> <td>-0.12</td> <td>-0.26</td> <td>0.22</td> <td>1.00</td> <td></td>	V2	0.37	-0.06	0.16	-0.12	-0.26	0.22	1.00												
V4 0.58 0.32 0.05 0.14 0.02 0.51 0.56 0.49 1.00 1.00 TV 0.66 0.42 0.01 0.01 0.59 0.68 0.51 0.99 1.00 1.00 Cull 0.02 0.11 0.01 0.58 0.21 0.83 0.14 0.23 0.34 1.00 Cull 0.02 0.11 0.01 0.05 0.21 0.23 0.24 0.24 0.26 0.23 0.34 1.00 Cull 0.22 0.11 0.05 0.21 0.31 0.23 0.24 0.24 0.25 0.24 1.00 Cull 0.23 0.24 0.24 0.26 0.24 0.26 0.24 1.00 1.00 Cull 0.25 0.29 0.21 0.01 0.02 0.21 0.21 0.24 0.24 1.00 Cull 0.25 0.29 0.21 0.21 0.21 0.21 0.24 0.24 0.26 0.24 0.26 0.24 Cull 0.25 0.29 0.21 0.21 0.21 0.21 0.21 0.21 0.21 0.22 0.22 0.24 0.24 0.24 0.24 0.26 0.24 0.26 0.24 0.26 0.24 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 <th< th=""><th>V3</th><td>0.43</td><td>-0.15</td><td>-0.12</td><td>0.26</td><td>-0.21</td><td>0.32</td><td>0.44</td><td>1.00</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>	V3	0.43	-0.15	-0.12	0.26	-0.21	0.32	0.44	1.00											
TV 0.66 0.42 0.01 0.01 0.01 0.28 0.21 0.02 0.02 0.02 </th <th>V4</th> <td>0.58</td> <td>0.33</td> <td>-0.05</td> <td>0.14</td> <td>0.02</td> <td>0.51</td> <td>0.65</td> <td>0.49</td> <td>1.00</td> <td></td>	V4	0.58	0.33	-0.05	0.14	0.02	0.51	0.65	0.49	1.00										
U1 0.09 0.23 0.11 0.01 0.28 0.21 0.23 0.21 0.23 0.21 0.23 0.21 0.23 0.21 0.23 0.21 0.02 0.11 0.05 0.21 0.24 0.25 0.24 0.25 0.24 0.25 0.24 1.00 Cu3 0.21 0.04 0.08 0.24 0.26 0.24 0.24 0.24 0.24 1.00 Cu3 0.21 0.04 0.08 0.24 0.24 0.24 0.24 1.00 1.02 Cu3 0.29 0.07 0.21 0.26 0.24 0.26 0.24 0.24 1.00 Cu3 0.29 0.08 0.29 0.26 0.24 0.26 0.24 1.00 Cu3 0.11 0.23 0.07 0.21 0.26 0.26 0.26 0.26 0.26 0.26 Sand 0.11 0.23 0.07 0.21 0.26 0.26 0.26 0.26 0.26 0.26 0.26 Sand 0.11 0.23 0.06 0.07 0.21 0.26 0.26 0.26 0.26 0.26 0.26 0.26 Sand 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 Sand 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 0.26 Sand 0.26 0.26 $0.$	ΤV	0.66	0.42	-0.03	0.04	-0.01	0.59	0.68	0.51	0.99	1.00									
Cu2 -0.40 0.22 0.11 -0.05 -0.31 -0.38 0.24 0.26 0.20 0.41 100 Cu3 0.21 0.04 0.08 0.14 -0.01 0.94 0.26 0.47 0.25 0.48 100 Cu4 0.25 0.29 0.24 0.26 0.47 0.26 0.47 0.26 0.48 100 Cu4 0.35 0.29 0.05 0.67 0.26 0.47 0.26 0.47 0.26 0.49 100 Cu4 0.11 0.02 0.07 0.27 0.24 100 0.26 0.47 0.26 0.41 0.05 0.74 100 Sand -0.13 0.29 0.06 -0.07 0.17 0.14 0.26	Cu1	0.09	-0.23	0.17	-0.10	-0.58	-0.21	0.83	0.18	0.35	0.34	1.00								
Cu3 0.21 0.04 0.08 -0.14 -0.01 0.91 0.26 0.47 0.51 0.93 0.48 1.00 Cu4 0.35 0.29 0.05 0.37 -0.05 0.37 -0.05 0.32 0.68 0.94 0.92 0.41 -0.05 0.54 1.00 Tu0 0.18 0.11 -0.23 0.60 -0.07 0.17 0.48 0.92 0.41 -0.05 0.54 1.00 Sand -0.13 0.39 0.69 -0.07 0.17 0.17 0.48 0.21 0.05 0.19 0.26 0.79 0.72 Sand -0.13 0.39 0.69 -0.07 0.17 0.71 0.28 0.70 0.29 1.00 7 7 Sand -0.13 0.39 0.69 -0.07 0.17 0.72 0.71 0.94 1.00 7 7 Sand -0.13 0.29 0.01 0.17 0.17 0.12 0.72 0.19 0.76 0.79 1.06 Sand -0.14 -0.15 0.01 0.12 0.14 0.72 0.14 0.17 0.05 0.79 0.79 0.79 0.70 Sand -0.14 -0.15 0.14 0.12 0.14 0.14 0.14 0.14 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16 0.16	Cu2	-0.40	0.22	0.11	-0.05	-0.31	-0.38	0.24	-0.45	-0.26	-0.20	0.41	1.00							
Cu4 0.35 0.29 -0.05 0.37 -0.05 0.32 0.68 0.94 0.92 0.41 -0.05 0.54 1.00 TCu 0.18 0.11 -0.23 0.60 -0.07 0.10 0.52 0.56 0.86 0.80 0.35 -0.19 0.40 0.94 1.00 Sand -0.13 0.39 0.38 0.69 -0.07 0.017 -0.17 -0.17 -0.48 0.71 -0.68 0.20 0.20 0.94 1.00 -0.78 -1.06 Sand -0.13 0.39 0.50 -0.07 -0.17 -0.17 -0.48 0.71 -0.68 0.20 0.94 1.00 -0.78 -0.79 Sand -0.13 0.39 0.50 -0.17 -0.17 -0.14 -0.15 -0.17 -0.17 -0.18 -0.12 -0.79 -0.78 -0.79 -0.78 -0.79 -0.78 -0.79 -0.78 -0.79	Cu3	0.21	-0.04	0.08	-0.14	-0.41	-0.01	0.94	0.26	0.47	0.51	0.93	0.48	1.00						
TCu 0.18 0.11 -0.23 0.60 -0.07 0.10 0.52 0.56 0.86 0.80 0.35 -0.19 0.40 1.00 1.00 Sand -0.13 0.39 0.38 0.69 -0.017 -0.17 -0.48 -0.71 -0.69 -0.71 -0.17 -0.48 -0.71 -0.69 -0.71 -0.88 1.00 -0.79 -0.88 1.00 Silt 0.03 0.51 0.59 0.50 0.51 0.61 0.70 0.51 -0.79 -0.88 1.00 Silt 0.03 0.51 0.59 0.50 0.50 0.51 0.60 0.51 0.61	Cu4	0.35	0.29	-0.05	0.37	-0.05	0.32	0.68	0.48	0.94	0.92	0.41	-0.05	0.54	1.00					
Sand -0.13 0.38 -0.69 -0.07 -0.17 -0.14 -0.51 -0.68 -0.07 -0.15 -0.17 -0.14 -0.58 -0.07 -0.15 -0.18 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.08 1.00 -0.09 0.70 -0.89 1.00 -0.09 0.70 -0.84 1.00 -0.01 0.01 0.11 0.12 0.21 0.14 0.17 0.05 0.17 0.13 0.70 0.84 1.00 -0.05 0.70 0.84 0.80 1.00 Class 0.11 0.12 0.13 0.14 0.47 0.47 0.55 0.05 0.14 0.10 0.15 0.10 0.12 0.10 0.10 0.10	TCu	0.18	0.11	-0.23	0.60	-0.07	0.10	0.52	0.56	0.86	0.80	0.35	-0.19	0.40	0.94	1.00				
Site 0.03 0.52 -0.50 0.01 -0.13 -0.01 0.15 0.38 0.50 0.52 0.01 0.17 0.63 0.70 -0.89 1.00 Clay 0.21 0.14 -0.15 0.28 0.32 0.15 0.54 -0.09 0.76 -0.84 1.00 OC 0.09 0.15 -0.32 0.14 0.47 0.52 -0.54 -0.09 0.76 0.84 -0.86 1.00 OC 0.09 0.15 -0.32 0.14 0.43 0.81 0.75 0.07 -0.25 0.70 -0.89 1.00	Sand	-0.13	-0.39	0.38	-0.69	-0.07	-0.17	-0.17	-0.48	-0.71	-0.68	0.07	0.21	-0.05	-0.79	-0.88	1.00			
Clay 0.21 0.14 -0.15 0.28 0.32 0.14 0.47 0.75 0.68 -0.15 -0.09 0.76 0.84 -0.86 0.52 1.00 OC 0.09 0.15 -0.38 0.57 0.43 0.81 0.75 0.07 -0.25 0.20 0.84 -0.67 0.67 0.09 1.00	Silt	0.03	0.52	-0.50	0.61	-0.13	-0.01	0.15	0.38	0.50	0.52	0.02	0.14	0.17	0.63	0.70	-0.89	1.00		
OC 0.09 0.15 -0.38 0.62 0.32 0.15 0.37 0.43 0.81 0.75 0.07 -0.25 0.20 0.87 0.92 -0.89 0.67 0.90 1.00	Clay	0.21	0.14	-0.15	0.59	0.28	0.32	0.14	0.47	0.75	0.68	-0.15	-0.54	-0.09	0.76	0.84	-0.86	0.52	1.00	
	0C	0.09	0.15	-0.38	0.62	0.32	0.15	0.37	0.43	0.81	0.75	0.07	-0.25	0.20	0.87	0.92	-0.89	0.67	06.0	1.00

Dynamics of trace metals, rare earth elements and nutrients in the CE

106

Table 3 and org	3.10 C ganic c	orrela carbon	tion m	aatrix astal s	(Pears sedime	son) fc ents	or trace	e meta	l (Sn,	V and	(Cu)	in diffe	rent fra	ictions,	sedim	ient tex	ture	
Variable	Sn1	Sn2	Sn3	Sn4	TSn	V1	V2	V3	V4	AL	Cul	Cu2 Cr	13 Cu4	TCu	Sand	Silt	Clay (ЭС
Sn1	1.00																	
Sn2	-0.50	1.00																
Sn3	-0.68	0.66	1.00															
Sn4	0.02	0.31	-0.46	1.00														
TSn	-0.63	0.95	0.55	0.50	1.00													
V1	-0.38	-0.58	0.09	-0.58	-0.48	1.00												
V2	-0.84	0.30	0.85	-0.55	0.29	0.58	1.00											
V3	0.44	-0.63	-0.96	0.54	-0.43	0.06	-0.70	1.00										
V4	-0.40	-0.57	0.10	-0.58	-0.46	1.00	0.59	0.05	1.00									
ΤV	-0.63	-0.34	0.27	-0.48	-0.21	0.96	0.74	-0.07	96.0	1.00								
Cu1	-0.30	0.05	-0.43	0.83	0.34	-0.03	-0.22	0.64	-0.02	0.09	1.00							
Cu2	-0.54	0.60	0.05	0.83	0.81	-0.34	0.01	0.15	-0.32	-0.10	0.83	1.00						
Cu3	-0.93	0.77	0.69	0.22	0.87	0.00	0.67	-0.48	0.03	0.29	0.36	0.73 1.	00					
Cu4	-0.36	-0.15	0.64	-0.94	-0.27	0.70	0.80	-0.63	0.70	0.68 -	0.65	-0.59 0.	10 1.0					
TCu	0.21	-0.80	-0.10	-0.81	-0.87	0.78	0.22	0.06	0.77	0.58 -	0.50	-0.84 -0.	55 0.7	0 1.00				
Sand	-0.57	0.13	0.83	-0.82	0.02	0.59	0.92	-0.78	0.59	0.65 -	0.56	-0.36 0.	37 0.9	6 0.47	1.00			
Silt	0.80	-0.59	-0.98	0.45	-0.53	-0.25	-0.93	0.88	-0.27	-0.45	0.31	-0.11 -0.	75 -0.6	8 0.04	-0.86	1.00		
Clay	-0.80	0.91	0.84	0.10	0.91	-0.19	0.65	-0.72	-0.17	0.08	0.07	0.58 0.	94 0.1	6 -0.58	0.44	-0.83	1.00	
0C	-0.81	0.90	0.85	0.08	06.0	-0.17	0.67	-0.72	-0.14	0.11	0.06	0.57 0.	94 0.1	8 -0.56	0.46	-0.85	1.00 1	00.

Dynamics of trace metals, rare earth elements and nutrients in the CE

107

3.5 DISCUSSION

3.5.1 Distribution of trace metals

There are numerous data sets available regarding the trace metal enrichment and their ecological implications in surface sediments of the Cochin estuary (Balachandran et al., 2005, 2006; Maheswari et al., 2006; Mohan et al., 2012; Deepulal et al., 2012; Selvam et al., 2012; Ramasamy et al., 2012; Martin et al., 2012; George et al., 2016) and its impact on the coastal environment (Nath et al., 2000; Balachandran et al., 2003, 2006; Laluraj, 2009; Sudhananth et al., 2011; Udayakumar et al., 2014; Sreekanth et al., 2015). The concentrations of various fractions of trace metals in sediments in the Cochin estuary and adjacent coastal region measured during the present study are comparable with the aforesaid literatures.

The trace metals were low in the sandy sediments at station 5 and 6. Stations with clayey texture showed high metal accumulation which was maximum at S_1 and S_2 . The existence of a tidally energetic condition at station 5 and 6 probably did not allow flocculation of fine particles over the sandy sediments, which led to a lower trace metal deposition. On the other hand, despite a relatively low clay and organic carbon at S_1 , the high concentration of trace metals in sediments could be due to the flow-restrictions in the region and high loading through the River Periyar. The weak flushing resulted in entrapment of fine particles and trace metals in sediments in the northern estuary (Balachandran et al., 2005). Moreover, % of smectite was higher in this region (S_1), which can hold more trace metals due to its high cation exchange capacity (80-120) and large surface area. The enrichment of metals in the coastal region (S_9 , S_{10} , S_{11} , and S_{12}) generally indicates that export fluxes from the Cochin Estuary are quite significant at times. Most of the trace

metals (Zn, Mn, Cr, Ni, Cd, Pb, Cu and Sn) in sediments (except station 5 and 6) showed values exceeding the standard sediment quality guidelines (SQG). The normalisation of metals with crustal abundance showed moderate to severe contamination for Zn and moderate contamination for other metals (Mn, Cr, Ni, Cu, Sn, Pb) indicating their anthropogenic input into the estuary. High enrichment factor for Zn, Cd, Cu, Pb and Fe in the north estuary was reported earlier (Balachandran et al., 2005). Zn and Cd observed during the present study were lower than their reported values, while the concentrations of Cu and Pb were slightly high. Estuarine and coastal sediments in general, showed a long-term accumulation of trace metals in sediments. In many cases, it was observed that anthropogenic inputs exceeded the natural weathering processes.

3.5.2 Fractionation of trace metals

The results obtained from the sediment fractionation showed that among the trace metals studied, all the metals were largely associated with the residual phase, although they are available in other fractions. The exchangeable and carbonate/bicarbonate fractions are the important nonresidual binding phases for some trace metals (Zn, Cd and Mn). An increase in the labile fraction of Cd and Mn with the increase in total Cd and Mn loading would exacerbate the potential toxicity of these metals in the estuarine and coastal environment. The risk assessment studies (RAC) also support this view indicating a medium risk for Zn and Mn against a high risk for Cd in the estuary. The concentration of Cd in the exchangeable fraction is in good agreement with the results of previous studies in CE (Mohan et al., 2012, Manju, 2014; Renjith et al., 2016). The organic phase was found to be a major binding phase of Pb, Cr, Ni and Sn in the estuarine sediments. The estuarine sediments enriched in organic matter could make stable bonds with these metals. However, there was a difference in the metal-organic complexes formed in the estuary and coastal sediments. High organic matter in coastal sediments is not an effective scavenger of most of the heavy metals studied however high affinity of vanadium towards the organic matter favored their stable binding in coastal sediments. It has been observed that marine derived organic matter in this coastal region has relatively less complexing capacity towards Ni, Cu, and Pb (Chakraborty et al., 2016). The results from the present study substantiate this, as the organic fractions of these metals have showed a decreasing trend towards the coastal sediments.

There are many studies showing the thermodynamic stability of Cuorganic complexes in sediments (Nair et al., 1992; Prasad et al., 2006; Mohan et al., 2012; Manju, 2014; Renjith et al., 2016). Cu in estuarine sediments can forms thermodynamically stronger complexes with the organic phase due to its high ionic potential and exchange rate (Chakraborty et al., 2016). The present study showed a different trend, as the major portion of Cu was residual and only a small portion was associated with organic phase. Cu in organic binding phase showed an increase with increased Cu loading in sediments however, there was poor correlation with organic carbon has been attributed to the variation in the binding capacity of organic compounds (Zhang et al., 2015). The natural organic compounds present in the environment are (1) aminocompounds (amino acid, protein, peptide, ammonia, etc), (2) carbohydrates (cellulose, amino- sugar, etc.), (3) lipids, isoprenoids and steroids (fat, fatty acid, alcohol etc.), (4) heterocyclic compounds, (5) phenol, quinone and humic compounds and (6) hydrocarbons (Jianfeg et al., 1984., Renjith et al., 2013; Manju and Sujatha, 2012; Gireesh Kumar et al., 2013, Sarkar et al., 2016). The composition, structure, reducing and adsorbing capacities and, the metamorphic forms of organic matter may affect the association of the metal on to the sediment. An earlier study has point out that mineralization under oxidizing conditions may release trace metals attached to the organic matter to the overlying water column (Jain et al., 2004). The labile organic matter undergoes mineralisation faster than the refractive organic matter (Gireeshkumar et al., 2013a). However, humic substances and other high molecular weight compounds take longer time for mineralization (Klucakova et al., 2017). Thus, the humic-bound metals are relatively stable in sediments. Thus, the nature of organic matter has an important role in the fractionation of trace metals in sediments. Moreover, the oxidized fraction comprises trace metals bound to the sulfides. One important pathway of metal immobilization is through sulfate reduction to form sulfide-rich sediments. Many of the trace metals are found to form insoluble metal sulfides that result in their low bio-availability (Sundelin and Erikkson, 2001).

It was observed that Fe-Mn oxy hydroxide is an efficient scavenger of metals like Zn, Ni, Pb, Cd, Mn and V in both the estuarine and coastal environments. The metals bound to Fe-Mn oxy hydroxide phases are found to be increasing with increasing metal loadings in the sediments. This can be due to the increased concentration of Fe-Mn oxides in this region. Several studies have highlighted the role of Fe-Mn oxides as excellent scavengers of trace metals in sediments (Nair et al., 1992; Balachandran et al., 2006, Shynu et al., 2012; Mohan et al., 2012).



Microbially mediated redox reactions can also partially reduce the insoluble Fe^{+3} and Mn^{+2} oxides and release them into pore water and overlying water (Petersen et al., 1995). Moreover, changes in ionic composition and pH also increase the mobility of metals adsorbed to mineral surfaces. However, high concentrations of metals are not the only requirement for their bioaccumulation, rather, it is purely species specific and dependent on the ecological responses of an organism (Maheswari et al., 1997, 2006; Renjitha et al., 2016).

The residual fractions were found to be the dominant in concentrating most of the trace metals in the estuarine and coastal sediments. Mn and Zn concentrations in the residual fraction were higher than the value reported for adjacent source rock (Nath et al, 2000) and the Standard Sediment Quality Guidelines (SQG). The enrichment of metals in the residual fraction may be due to the transformation of non-residual fractions under favorable condition to mask their adverse impact to biota. Trace metals are naturally persistent and non biodegradable. Therefore, they can only be converted to stable forms of reduced availability (Balachandran et al., 2003; 2006). They suggested that there is an annual transformation in the behavior of elements from an inorganic deposition to organic association followed by sedimentary modifications and normalization along the coastal environment of Cochin. From the present study, it is clear that the nature of binding phase has great impact on the mobility of trace metals in the sediments.

A comparison has been made with fractionation of sediment trace metals by other researchers in Cochin estuary. Table 3.11 shows that there is an increase in the exchangeable Cd (Fr 1) in the estuarine sediments over the years. The concentrations of Mn, Cr, Pb, Ni and Cu in exchangeable/carbonate fractions have decreased over the years, but the % of Zn in Fr.1 was similar to previous reports. The high concentration of organic-bound Cu initially has now decreased substantially through its masking into a stable residual phase. Except for these two aspects, the present results are in good agreement with the earlier studies.

The estuarine sediments are exposed to dynamic environment, where short-term fluctuations in the tides, turbidity, salinity, pH, redox conditions, dissolved oxygen etc. bring out significant alterations in their texture and responses to the environment. A small change in these conditions may severely alter the sediment properties and their bioavailability, which should be considered to minimize the pollution levels and to conserve the estuarine and coastal regions.

3.6 SUMMARY AND CONCLUSION

The texture of the estuarine sediments varied from clayey-sand to siltyclay; while the coastal sediments were mainly silty-clay and the bar mouth and surrounding regions were sandy due to strong flushing. There was a progressive increase in the fine fractions from upper to the central estuary. The near shore areas (~5-6.5 m) showed high clay content and organic enrichment compared to the offshore stations (~30 m). It was observed that the silty-clay sediments preferentially accumulated organic carbon while the sandy sediments were impoverished in organic carbon. Organic carbon was relatively high in the coastal sediments than in estuarine sediments. Higher OC in the near shore sediments than in offshore sediments can due to higher clay content and high productivity along the inner shelf. Kaolinite and smectite were the dominant clay minerals observed in the study region with relatively lower percentage of chlorite, illite, quartz and gibbsite. The trace metals were generally low in the sandy sediments and high in clayey sediments. High metal accumulation was observed in north estuary, where smectite content was also high. The concentrations of most of the trace metals (Zn, Mn, Cr, Ni, Cd, Pb, Cu and Sn) in silty-clay dominant sediments were exceeding the standard sediment quality guidelines. The normalisation of metals with crustal levels showed moderate to severe contamination for Zn and moderate contamination for other metals (Mn, Cr, Ni, Cu, Sn, Pb) indicating their anthropogenic input into the estuary. The present study identified increased anthropogenic input of trace metals into the estuary and its impact on the coastal region. The trace metal composition in sediments of the Cochin estuary and adjacent coastal stations was comparable with the previous studies.

Exchangeable, carbonate/bicarbonates phases were the important nonresidual fractions of Zn, Cd and Mn. The enrichment of labile fractions of Cd and Mn with increase in total metal content (Cd_T and Mn_T) indicates that this would exacerbate the potential toxicity of these metal in the estuarine and coastal environment. The health risk assessment studies (RAC) also support this showing a medium risk for Zn and Mn and high risk for Cd in sediments, as it was the most mobile element in the sediments. These studies underline the potential health risk about the habitat in the north estuary due to trace metal pollution.

The organic phase was the major binding phase for Pb, Cr and Ni in the organic-rich estuarine sediments showing its stability in binding metals. The organic enrichment in the estuary was found to play an important role in the immobilization of Pb, Cr and Ni. High organic matter in coastal sediments is not an effective scavenger of most of the trace metals studied however its high affinity towards Vanadium makes it more stable in coastal sediments. Vanadium in estuarine sediments was mostly concentrated in the Fe-Mn oxy hydoxide phase rather than organic phase. Major portion of Cu in the estuarine sediments was as residual form and there was only a weak association with organic phase. The non-residual fractions of most of the trace metals in coastal sediments were preferably associated with Fe-Mn oxy hydroxide fraction. Thus, Fe-Mn oxyhydroxide was found to be a good scavenger of Zn, Ni, Pb, Cd and Mn in both estuarine and coastal environments. It was also observed that the second host for Sn and Pb, varied with respect to the sediment texture and organic content of the sediments.

The residual phase was found to be a prominent host for all metals in estuarine and coastal sediments. The residual fraction represented an essentially non-labile component and thus acts as an important sink for trace metals in sediments. Mn and Zn in residual fraction were higher than the value reported for adjacent source rock and the standard sediment quality guidelines (SQG). The enrichment of metals in the residual fraction may be due to the transformation of non-residual forms to stable residual forms to mask its toxicity to biota. The deposition of trace metals in sediments was found to be due to the combined effect of the transport, dispersal and deposition processes taking place from their sources in the rivers to sinks in the estuary and coastal region. It was observed that the high sorption capacity of clay and organic matter strengthen the metals bound to their lattice. The nature of metal binding phase played an important role in the fate and transport of trace metals in the study region. It is well established that bioavailability of a metal is not only a factor for its bioaccumulation; it actually depends on the species specific physiological and ecological conditions. However, measurement of such metal

fractions will be useful to understand the metal release from sediments under different conditions and implement for the natural water management systems.

	F. 4 1001						
Metal	Fr.1 (%)	Fr.2 (%)	Fr.3 (%)	Fr.4 (%)	TM(mg/kg)	Instrumentation	Reference
	2.1 - 11.4	22.2 - 38.8	23 - 54	13 - 52	29.3 - 290.5	FAAS	Rajamani et al, 1994
Zn	0.6 - 33	2.4 - 19.3	6.7 - 43.4	28 - 88.8	135.3 - 258.9	AAS	Nair et al, 1992
	0.6-25.1	1.2 - 8.6	0.7 - 4.1	63.7 - 92.5	112.5 - 435.7	ICP-MS	Present study
	23.6 - 30.2	13.0 - 31.0	14.8 - 38.3	4.8 - 66.5	123.6 - 280.5		Rajamani et al, 1994
μ	2.7 - 49.5	1.4 - 25.7	10.1 - 40	27.8 - 82.2	7.0 -170		Nair et al, 1992
	5.2 -19.4	2.04 -18.8	2.0 - 7.6	28.5 - 84.6	298.5 - 720	ICP-MS	Present study
	6 - 19.6	bdl - 4.2	8.6 - 44.9	35 - 80.8	0.97 - 9.4		Rajamani et al, 1994
Cd	2.1 - 41.5	1.7 - 26.1	15.2 - 59.5	23.3 - 69.4	0.34 - 8.4		Nair et al, 1992
	9.5 - 48.8	13.2 - 40.4	2.99 -11.6	5.9 - 62.4	0.1 - 2.4	ICP-MS	Present study
	1.1 - 2.5	13.5 - 34.1	34 - 45	22.8 - 40.6	5.5 - 51.3		Rajamani et al, 1994
č	0.69 -16.2	0.0 - 4.4	7.6 - 49.2	46.6 - 89.8	1.3 - 50.2		Nair et al, 1992
3	0.3 - 3.0	3.5 - 26	16 - 32		22.2 - 82.2	GFAAS	Chakraborty et al, 2016
	0.1 - 0.6	0.03 - 0.25	1.1 - 4.1	90.5 - 96.9	27.2 - 65.7	ICP-MS	Present study
	1.81 - 13.4	bdl - 29.1	16.2 - 32.6	32.6 - 75.4	33.2 - 117.1		Rajamani et al, 1994
Ż	1.54 - 23.03	1.1 - 20.9	1.4 - 28.6	52.5 - 92.7	3.5 - 49.9		Nair et al, 1992
2	0.3 - 2.0	20- 48	7.0 - 30	-	24.4 - 34.2	GFAAS	Chakraborty et al, 2016
	0.5 - 6.0	7.6 -18.6	3.8 - 17.8	64.6 - 87.1	16.7 - 41.9	ICP-MS	Present study
	0.53 - 8.7	0.42 - 4.99	3.1 - 21.9	71.4 - 92.0	1.86 - 70.6		Nair et al, 1992
ïz	0.8 - 3	4.0 -15.0	22.0 -27.0		21.1 - 36.1	GFAAS	Chakraborty et al, 2016
	0.9 - 4.7	1.8 - 4.4	3.9 - 21.5	70.3 - 89.5	15.5 - 102.9	ICP-MS	Present study
č	0.16 - 8.5	0.03 - 0.5	2.2 - 27.6	71 - 95.6	4.1 - 99.7		Nair et al, 1992
5	0.1 - 1.5	1.4 - 3.02	8.0 - 17.4	59.1 - 89.6	84.2 - 183	ICP-MS	Present study
*bdl-b	elow detection	limit; TM- Tota	il metal (mg/kg				

****\$\$****

Dynamics of trace metals, rare earth elements and nutrients in the CE

Table 3.11 Metal fractions in Cochin estuarine sediments – a comparison with previous

116

<u>Chapter</u> **4** RARE EARTH ELEMENTS

- 4.1 Introduction
- 4.2 Sampling Strategy
- 4.3 Analytical Methodology
- 4.4 Results
- 4.5 Discussion
- 4.6 Summary and Conclusion

4.1 INTRODUCTION

The International Union of Pure and Applied Chemistry (IUPAC) define the rare earth elements (REE) as a collection of seventeen chemical elements in the periodic table that include 15 lanthanides (Lanthanum, Cerium, Praseodymium, Neodymium, Samarium, Europium, Gadolinium, Terbium, Dysprosium, Holmium, Erbium, Thulium, Ytterbium and Lutetium) and two other elements Scandium and Yttrium. Scandium and Yttrium are considered as rare earth elements since they tend to occur in the same ore deposits as the lanthanides and exhibit similar chemical properties. The rare earth elements are often described as 'light-group rare earth element' (LREE) or 'heavy group rare earth element' (HREE). The definition of a LREE and HREE is based on the electronic configuration of each rare earth element. The LREE are defined as Lanthanum (atomic number 57) through Gadolinium (atomic number 64). The LREE have in common increasing unpaired electrons from 0 to 7. The HREE are defined as Terbium (atomic number 65)



through Lutetium (atomic number 71) and also Yttrium (atomic number 39). The HREE differ from the first eight lanthanides in that they have paired electrons (a clockwise and counter clockwise spinning electron). The LREEs have no paired electrons. Yttrium is included in HREE group based on its similar ionic radius and chemical properties. Scandium is also trivalent; however, its other properties are not similar enough to classify it as either LREE or HREE.



Fig 4.1a: Abundance of Rare Earth Element, Thorium and Uranium per million atoms of silicon (https://en.wikipedia.org/wiki/File:Elemental_abundances.svg).

Despite their name, rare earth elements (with the exception of radioactive promethium) are relatively plentiful in the Earth crust (Fig. 4.1a) The more abundant REE are similar in crustal concentration to common place industrial metals such as chromium, nickel, copper and lead. Even the least

abundant REE, thulium is nearly 200 times more common than gold. In contrast to ordinary base and precious metals, REE have very little tendency to become concentrated in exploitable ore deposits. It was due to the scarcity of the mineral that led to the term 'rare earth'. Thorium and uranium are the two commonly occurring actinides, which have major roles in aquatic biogeochemical processes.

REE are widely used in modern technological devices such as super conductors, magnets, electronic polishers, refining catalysts and hybrid car components (primary batteries). Rare earths are also used as active ions in luminescent materials used in opto electronics (Nd:YAG laser). Erbium-doped fiber amplifiers are significant devices in optical communication systems. Phosphors with rare earth dopants are also widely used in cathode ray tube technology such as television sets; europium as a phosphor dopant made good red phosphors possible. Yttrium iron garnet spheres have been useful as tunable microwave resonators. Rare earth oxides are mixed with tungsten to improve its thermal resistance during welding.

Rare earth elements have become important geochemical tracers to understand the chemical evolution of the earth's crust (Goldstein and Jacobsen, 1988; McLennan, 1989; Gaillardet, 1995; Dupre et al., 1996). Moreover, REE have been used as analogues for actinide elements for the disposal of radioactive wastes, demonstrating their immobility in weathering environment (Wood, 1990). There are mainly two processes controlling the REE variation in sediments (i) weathering at source and (ii) sedimentological characters in the environment (Nath et al; 1997, 2000, Pattan et al., 2005). The major sources of REE to the ocean are rivers (Goldstein et al., 1988), ocean crust and sediment diagenesis, though it is small relative to the other two (Elderfield and Sholkovitz, 1987). REE in marine sediments are partitioned into terrestrially derived lithogenous fraction, biogeneous fraction, carbonate fluorapatite, montorillonite, the zeolite, phillipsite, glauconite and Fe/Mn oxide hydride fraction (Piper et al., 2013). Light rare earths (LREE) are either scavenged onto the surface particles such as Mn/Fe oxyhydroxides and clay minerals or precipitated as REE phosphates (Byrne and Kim, 1990). The formations of both promote the enrichment of HREEs in seawater. Certain minerals such as alumino silicates or appatite tend to have high concentration of REE while other minerals have very low concentrations (Graf, 1977).

Several studies have indicated that REE are significantly mobilized during weathering and they behave non-conservatively (Braun et al., 1990; Sholkovitz et al., 1994). River sediments generally have residual and labile fractions. The labile fractions are sensitive to environmental changes such as redox conditions, salinity, chelates, pH, adsorption desorption, complexation, precipitation etc. The provenance and distribution of REEs in Cochin estuary and adjacent coastal region have been studied early (Nath et al., 2000; Siby, 2004; Laluraj 2009; Deepulal et al., 2012). However, very little is known about their fractionation in sediments. Accordingly, the objective of this study is:

To study the distribution and fractionation of rare earths and radioactive uranium and thorium in the sediments of Cochin estuary and adjacent coastal regions to infer their provenance, behaviour and depositional conditions.



4.2 SAMPLING STRATEGY

In order to understand the distribution and fractionation of rare earths, thorium and uranium, the surface sediments were collected from 8 locations from the north and central region of Cochin estuary, in addition to the adjacent coastal stations at a depth of 5, 6.5, 27 and 30m (Fig 4.1b). The coastal stations at ~5 and ~27 m were located north of the Cochin inlet and stations at ~6.5 and ~30 m were located south of the Cochin inlet. The geographic positions of sampling stations are given in Table 4.1. The sampling was carried out during October 2011. Sediments were collected using a Van veen stainless steel grab ($0.042m^2$), sealed in plastic bottles, kept in ice, transported to the laboratory and stored at -20°C until analysis. Salinity of bottom water measured using CTD are given in Table 4.3.



Fig 4.1b: Map of the study area showing locations of sampling sites



Dynamics of trace metals, rare earth elements and nutrients in the CE

SI	Sampling Locations	St. code	Lat (°N)	Long(°E)
No				
1	Moolampilly	\mathbf{S}_1	10.0312	76.2666
2	Vallarpadam north	S_2	10.0174	76.2442
3	Bolghatty	S ₃	9.9913	76.2680
4	Vallarpadam-Bolghatty bridge	\mathbf{S}_4	9.9891	76.2607
5	Vallarpaadam-Vypeen bridge	S_5	9.9823	76.2445
6	Cochin inlet (10-15m)	S_6	9.9702	76.2415
7	Ernakulam Channel	S_7	9.9454	76.2880
8	Mattanchery channel	S_8	9.9341	76.2688
9	Near shore (5.3m)	S_9	9.9734	76.1870
10	Near shore (6.8m)	\mathbf{S}_{10}	9.9293	76.2264
11	Off shore (27m)	S_{11}	10.0040	76.0833
12	Offshore(30m)	S ₁₂	9.9167	76.0833

Table 4.1: Station locations in Cochin estuary and adjacent coastal region

4.3 ANALYTICAL METHODOLOGY

The sediment samples were portioned into three for the analysis of total metals, organic carbon and grain size. Before the grain size analysis, the inorganic carbonate was removed using 10% HCl and organic matter was removed using H_2O_2 (Folk, 1980). For texture analysis, the sediments were washed with distilled water, dried and treated with sodium hexametaphosphate to disperse the sediments and the particles were separated following the pipette analysis (Krumbein and Petti John, 1938). Shepard nomenclature was used to describe the sediment texture (Shepard, 1954). Sediment organic

carbon (OC) was analyzed by wet chromic acid oxidation method suggested by El Wakeel and Riley, 1957. The amount of total organic matter (TOM) was computed by multiplying the OC values with 1.80 (Muller et al., 1986).

SI		Standard F	Reference Material	(MAG 1)	
No	Element	Certified value (mg/kg)	Observed value (mg/kg)	RSD (%)	PAAS (mg/kg)
1	Sc	17 ± 1	17.2 ± 0.161	0.934	16
2	Y	-	28 ± 0.372	1.328	26
3	La	43	43 ± 0.562	1.306	38.20
4	Ce	88	88 ± 0.984	1.119	79.60
5	Pr	-	9.3 ± 0.012	0.125	8.83
6	Nd	38	38 ± 0.349	0.918	33.90
7	Sm	7.5	7.5 ± 0.111	1.476	5.55
8	Eu	1.6	1.6 ± 0.017	1.074	1.08
9	Gd	5.8	5.8 ± 0.036	0.612	4.66
10	Tb	0.96	1.0 ± 0.012	1.208	0.77
11	Dy	5.2	5.2 ± 0.128	2.463	4.68
12	Но	1	1.0 ± 0.014	1.351	0.99
13	Er	3	3.0 ± 0.046	1.521	2.85
14	Tm	0.43	0.43 ± 0.000	0.048	0.41
15	Yb	2.6	2.6 ± 0.016	0.622	2.82
16	Lu	0.4	0.4 ± 0.010	2.535	0.433
17	Th	-	11.9 ± 0.319	2.682	14.6
18	U	-	2.7 ± 0.131	4.852	3.1

Table 4.2: REE, Th and U concentrations (mg/kg) in Standard Reference Material (MAG 1), RSD (%) in analysis and are given below.

*Standard reference material (MAG 1) was obtained from Geological survey of India (GSI). *PAAS is the Global Shale Value used for normalisation of REE.

For the analysis of trace metals, the sediment samples and rock standards (MAG 1) were dried (at 60°C), powdered (mortar and pestle) and acid digested with a mixture of supra pure HF+HNO₃+HClO₄ (7:3:1) in teflon vessels on a hot plate (Balaram and Rao, 2002). The residues were redissolved in 2% ultrapure HNO₃ (containing 1ppm Rhodium as internal
standard) and analyzed using an Inductively Coupled Plasma-Mass Spectrometer (ICP-MS). The Certified Reference Material (MAG 1), obtained from Geological Survey of India was used to calibrate the method. Fractionation of sediment metal was achieved through sequential extraction procedure (SEP) suggested by Tessier et al., 1979 (Methodology in detail given in chapter 3). Precision and accuracy of the analysis was checked by the replicate analysis of the certified reference material (MAG-1) obtained from Geological survey of India. % error of the analysis ranged from 0.05 to 5.38% for MAG-1 reference material. The relative standard deviation (standard deviation/ mean of triplicate samples) ranged from 0.048 to 4.8% for MAG-1 reference material (Table 4.2).

4.4 **RESULTS**

4.4.1 General Characteristics of sediments

The organic carbon and textural pattern of sediments from Cochin estuary and adjacent coastal stations are given in Table 4.3. The organic carbon in the estuarine and coastal stations varied from 6.53 to 25.81mg/g and 24.22 to 36.62mg/g respectively. In the estuary, the organic carbon was low at S_5 and S_6 , while all other estuarine stations showed high organic carbon. The coastal sediments showed higher organic carbon than estuarine sediments. The texture of sediments in the estuary varied from sandy-silt to silty-clay whereas in coastal sediments, the texture was silty-clay for all stations (in detail given in chapter 3). The dominant clay minerals identified in the study region was kaolinite and smectite (in detail given in chapter 3).



Stations	Salinity	Organic carbon (mg/g)	Sediment Texture
\mathbf{S}_1	4.68	17.22	Clayey sand
\mathbf{S}_2	5.44	19.97	Silty clay
S_3	9.13	22.45	Sandy clay
\mathbf{S}_4	9.84	24.1	Silty clay
S_5	10.17	6.53	Clayey sand
S_6	25.02	12.7	Clayey sand
\mathbf{S}_7	10.09	13.43	Sandy silt
S_8	11.66	25.81	Silty clay
S_9	28.86	36.62	Silty clay
\mathbf{S}_{10}	23.42	32.15	Silty clay
S_{11}	26.28	27.28	Silty clay
S ₁₂	30.58	24.22	Silty clay

 Table 4.3:
 Organic Carbon and Textural pattern of estuarine and coastal sediments and salinity in the study region

4.4.2 Distribution of REE (Lanthanides, Sc and Y)

The distributions of REE in the estuarine and coastal sediments are presented in Fig 4.3-Fig 4.18. The distribution of REEs in the sediments of Cochin estuary varied as Sc: (7.72 to 20.62)mg/kg; La: (24.46 to 102)mg/kg; Ce: (44.26 to 171.28)mg/kg; Pr: (4.33 to 17.37)mg/kg; Nd:(18.13 to 70.49)mg/kg; Sm: (3.41 to 11.11)mg/kg; Eu: (1.16 to 2.18) mg/kg; Gd: (2.39 to 8.05)mg/kg; Tb: (0.38 to 1.23) mg/kg; Dy: (2.12 to 6.63)mg/kg; Ho: (0.43 to 1.31)mg/kg; Er: (1.3 to 3.85) mg/kg; Tm: (0.18 to 0.60) mg/kg; Yb: (1.10 to 3.16) mg/kg: Lu: (0.17 to 0.47) mg/kg; Y: (12.62 - 38.90)mg/kg respectively.

REE in the sediments of coastal stations were lower than estuarine stations and varied as Sc: (14.92 to 18.8)mg/kg ; La: (44.02 to 55.73)mg/kg ; Ce: (91 to 102.05)mg/kg; Pr: (8.85 to 10.25)mg/kg; Nd: (37.68 to

43.08)mg/kg; Sm: (7.01 to 7.56)mg/kg; Eu: (1.78 to 1.91)mg/kg; Gd: (5.62 to 6.19)mg/kg; Tb: (0.90 to 1.05)mg/kg; Dy: (4.94 to 5.89)mg/kg; Ho: (0.82 to 1.20)mg/kg; Er: (2.82 to 3.49)mg/kg; Tm: (0.39 to 0.48)mg/kg; Yb: (2.3 to 2.87)mg/kg: Lu: (0.33 to 0.40)mg/kg; Y (26.75 - 33.98) mg/kg respectively.

Ce was markedly high in sediments of estuarine and coastal stations as it was the most abundant REE on earth crust. Lutetium was the least abundant rare earth present in the estuarine and coastal sediments. Rare earth elements (REE) were generally concentrated in the northern stations (S_1, S_2) and showed a gradual decrease towards the central estuary (S_3, S_4, S_7, S_8) and remained at minimum concentrations near the Cochin inlet with sandy sediments ($S_5 \& S_6$). The light REE (LREE) and heavy REE (HREE) showed minimum variations in its spatial distribution. LREEs were slightly lower at S_2 (north estuary). In the coastal region, LREE showed a gradual decrease towards deeper depths, whereas HREE showed higher concentration at intermediate depths [S_{10} (6.8m) & S_{11} (27m)]. Scandium showed pattern similar to that of LREE and Yttrium showed pattern similar to that of HREE in their distribution (Fig 4.3 and Fig 4.4). Correlation tests were carried out to understand the inter-relation between clay minerals and REE in the sediments using statistical software XLSTAT 2014. No significant correlation ($r^2 < 0.5$) was observed between the dominant clay minerals (smectite, kaolinite) and total REE. However REE in Fr 1 showed a linear decrease with increasing kaolinite (Appendix 3; $r^2 > 0.5$)

4.4.3 Distribution of Actinides (Thorium and Uranium)

Thorium is an abundant element in earth crust, as it has three times higher concentration than uranium. Thorium in the estuarine sediments varied from 6.62 to 35.61mg/kg and in coastal sediments varied from 10.95 to 12.66mg/kg (Fig 4.19). Thorium naturally occurs in thorite, uranothorite, thorianite and is a major component of monazite. It is present in significant amount in the minerals zircon, titanite, gadolinite and betafite. Thorium concentrations of about 15.2 to 18.9mg/kg have been reported in fluvial sediments (<4m) of this estuary (Nath et al., 2000), which has only riverine sources. The concentration of Th observed in the estuarine sediments and coastal sediments was comparable with the values reported by Laluraj, 2009, however higher than the values reported by other studies (Shajan et al., 2001; Deepulal et al., 2012). Markedly high concentration in the northern station (S₁) could be due to industrial waste, which has been reported in earlier studies also (Laluraj, 2009; Joseph et al., 1984).

Uranium

Uranium, a radioactive material, is very reactive and naturally present in rocks and soils. U in the estuarine sediments varied from 0.95 to 6.99mg/kg and the coastal sediments showed values in the range 2.2 to 3.31mg/kg (Fig 4.20). The concentration of U observed in the estuarine sediments were lower than the values reported by Laluraj, 2009, but are high compared to other studies in estuary (Shajan et al., 2001; Deepulal et al., 2012). However U concentration in coastal sediments observed during the present study was higher than the earlier reports from coastal region (Shajan et al., 2001; Laluraj, 2009; Deepulal et al., 2012).

4.4.4 Shale Normalisation to identify the elemental enrichment

Shale normalisation values for Rare earth elements, thorium and uranium with respect to global shale along the study region are given in Table 4.4. The normalisation factor (NF) <1 refers to low contamination, 1 < NF < 3

indicates moderate contamination, 3 < NF < 6 indicates severe contamination and NF > 6 indicates extreme contamination. Rare earth elements and uranium showed moderate contamination for estuarine sediments namely S₁, S₂, S₃ & S₄ and for coastal sediments namely S₉, S₁₀, S₁₁ & S₁₂ respectively. However the central estuarine stations S₅, S₆, S₇ & S₈ showed low of level contamination for all REEs, uranium and thorium. Thorium showed moderate contamination in estuarine stations S₁, S₂, S₃ & S₄, while all other sediments are of low contamination with respect to thorium.

 Table 4.4:
 Shale normalized values of Rare earths, U and Th in the estuarine and coastal sediments

Station	La _n	Ce _n	Prn	$\mathbf{Nd}_{\mathbf{n}}$	Sm _n	Eun	$\mathbf{Gd}_{\mathbf{n}}$	Tb _n	$\mathbf{D}\mathbf{y}_{\mathbf{n}}$	Hon	Ern	Tm n	Yb _n	Lun	$\mathbf{U}_{\mathbf{n}}$	Th _n
S_1	2.67	1.78	1.95	2.08	2.00	1.98	1.73	1.52	1.32	1.23	1.27	1.32	1.02	1.00	2.26	2.44
S_2	1.38	1.46	1.15	1.32	1.39	1.66	1.34	1.31	1.22	1.17	1.26	1.47	1.07	1.03	1.97	1.13
S_3	1.57	2.15	1.97	1.43	1.55	2.02	1.59	1.59	1.42	1.32	1.35	1.31	1.12	1.09	1.64	1.29
S_4	1.45	1.45	1.16	1.34	1.36	1.67	1.26	1.24	1.15	1.06	1.11	1.06	0.90	0.85	1.48	1.20
S_5	0.72	0.58	0.51	0.55	0.61	1.25	0.51	0.49	0.45	0.43	0.45	0.45	0.39	0.39	0.31	0.45
S_6	0.64	0.56	0.49	0.53	0.62	1.14	0.52	0.53	0.51	0.48	0.49	0.48	0.42	0.40	0.35	0.66
S_7	0.87	0.75	0.68	0.73	0.86	1.48	0.74	0.71	0.66	0.63	0.64	0.63	0.52	0.53	0.53	0.69
S_8	0.79	0.95	0.66	0.81	0.83	1.07	0.87	0.87	0.84	0.83	0.88	0.89	0.80	0.76	1.12	0.86
S_9	1.46	1.33	1.16	1.27	1.32	1.74	1.21	1.16	1.06	0.97	0.99	0.96	0.81	0.77	0.71	0.86
\mathbf{S}_{10}	1.27	1.34	1.11	1.27	1.36	1.69	1.30	1.31	1.24	1.15	1.18	1.15	0.96	0.92	1.07	0.87
S_{11}	1.25	1.28	1.09	1.22	1.36	1.77	1.33	1.35	1.26	1.21	1.23	1.19	1.02	0.93	1.04	0.79
S_{12}	1.15	1.14	1.00	1.11	1.26	1.65	1.23	1.23	1.14	1.11	1.13	1.06	0.92	0.88	1.00	0.75

4.4.5. Sources of Rare earth elements

The shale normalized rare earth patterns for estuarine and coastal sediments were depicted in Fig 4.2. Ratios (La_n/Yb_n) , (La_n/Gd_n) , (La_n/Sm_n) , (Sm_n/Yb_n) , and (Gd_n/Yb_n) are used to identify relative enrichment or depletion of LREE, MREE, and HREE (Table 4.5). These ratios indicated LREE enrichment over HREE (La_n/Yb_n) , LREE enrichment over MREE (La_n/Gd_n)

Rare earth elements	CHAPTER 4
	-

and La_n/Sm_n), and MREE enrichment over HREE $(Sm_n/Yb_n, and Gd_n/Yb_n)$ respectively.

missing of LDEE MDEE and LIDEE to identify their

1 able 4.5.	Geoche	inical ratios		LUCI	E, MIKEE al	ій г	IKEE 10	Identify	men
	relative	enrichment	in	the	sediments	of	Cochin	estuary	and
	adjacent	coastal stati	ons	•					

Table 15. Casel

Station	(La/Yb) _n	$(La/Gd)_n$	(La/Sm) _n	$(Gd/Yb)_n$	(Sm/Yb) _n
S_1	2.61	1.55	1.33	1.69	1.96
\mathbf{S}_2	1.29	1.03	1.00	1.25	1.29
S_3	1.40	1.0	1.01	1.42	1.39
\mathbf{S}_4	1.60	1.15	1.06	1.39	1.51
S_5	1.84	1.40	1.17	1.31	1.57
\mathbf{S}_{6}	1.52	1.24	1.04	1.23	1.47
\mathbf{S}_7	1.67	1.17	1.01	1.43	1.65
S_8	1.00	0.92	0.96	1.09	1.04
S_9	1.79	1.21	1.10	1.48	1.62
\mathbf{S}_{10}	1.32	0.97	0.93	1.36	1.42
\mathbf{S}_{11}	1.23	0.94	0.92	1.30	1.33
S ₁₂	1.26	0.94	0.91	1.34	1.38

Shale normalized REE pattern of S_8 which are almost consistent, except for Ce and Eu (Fig 4.2). La_n/Yb_n at S_8 is 1; (La_n/Gd_n and La_n/Sm_n) ratios ~ 1; (Sm_n/Yb_n, and Gd_n/Yb_n) ratios ~ 1 and are similar to the values of terrigenous input (Nath et al., 1997). The remaining sediments exhibited an enrichment of light REE relative to HREE (Fig 4.2), with La_n/Yb_n ratios of (1.40-2.51) and (1.23-1.71) of PAAS in estuarine and coastal sediments. All the sediments exhibited enrichment in MREE over HREE, with (La_n/Gd_n) and (La_n/Sm_n) ratios of (1.25-1.69) and (1.29-1.96). Shale normalized REE pattern showed Steep HREE curve and flat LREE curve in the central estuary (S₃, S₄, S₅, S₆, S₇) and in the coastal stations (S₉, S₁₀, S₁₁, S₁₂). Flat shale normalized pattern and LREE enrichment are the characteristics of intermediate and felsic source rock (McLennan, 1989; Nath et al., 1997). These are in agreement with other studies from Cochin estuary and adjacent continental shelf (Laluraj et al., 2009; Deepulal et al., 2012). Shale normalized pattern observed in the sediments of north estuary (S_1 and S_2) showed a deviation from the normal pattern, possibly due to the additional supply of REEs into the northern estuary.



Fig 4.2: Shale normalized pattern of rare earths (REE) from estuarine and coastal sediments during October 2011.

4.4.6. Geochemical proxies to identify the depositional characteristics4.4.6.1 Ce/Ce* ratio

Due to their multiple oxidation states, Ce and Eu tend to behave differently from their strictly trivalent neighbors under different oxygenation conditions (Piper, 1974, Nath et al., 1997). Based on this, a cerium anomaly relates to the increase or decrease in solubility of Cerium. Under reducing conditions, Ce^{3+} is relatively soluble while under oxidizing conditions CeO_2 precipitates. The Ce anomaly, its enrichment or depletion compared to its neighboring elements, is estimated as Ce/Ce*, where Ce is Ce sample/Ce PAAS and Ce* is the predicted values obtained by the linear interpolation between shale-normalized La and Pr values.

130

CHAPTER 4

$$Ce^{*} = \frac{(La_{Sample}/La_{PAAS}) + (Pr_{Sample}/Pr_{PAAS})}{2}$$

Ce/Ce* values >1, <1 and 1 indicates positive, negative and no anomaly, respectively.

Ce/ Ce* ratios calculated in the sediments of Cochin estuary and adjacent coastal station are given in Table 4. 6. Here the Ce/Ce* values were generally positive (0.95 - 1.30), except at S₁. A Ce anomaly ~1 at S₅, S₆, S₇ & S₉ means that the REE are generally derived from terrigenous sources. Still higher values at S₂, S₃, S₄, S₈, S₁₀, S₁₁ and S₁₂ are indicative of its accumulation in sediments in these regions. Thomas et al., 2003 state that zircons show a positive Ce anomaly due to the incorporation of Ce⁴⁺ into zircon, since Ce⁴⁺ has the same charge (0.97 Å) and ionic radius of Zr⁴⁺ (0.84 Å).

Stations	Eu/Eu*	Ce/Ce*	U/Th
S ₁	1.06	0.77	0.20
S_2	1.22	1.15	0.37
S_3	1.28	1.22	0.27
S_4	1.27	1.11	0.26
S_5	2.23	0.95	0.14
S_6	2.01	0.98	0.11
S_7	1.85	0.97	0.16
S_8	1.27	1.30	0.27
S 9	1.37	1.02	0.17
S_{10}	1.27	1.13	0.26
S_{11}	1.31	1.09	0.28
S ₁₂	1.32	1.06	0.28

Table 4.6:Geochemical proxies to identify the deposition conditions of REE
in sediments of Cochin estuary and adjacent coastal stations.

A negative Ce anomaly (0.77) was noticed at S_1 (northern estuary) indicating its depletion, which has been already reported from this region (Deepulal et al., 2012; Manoj et al., 2016).

4.4.6.2 Eu/Eu* ratio

Europium anomaly is calculated as Eu/Eu*, where Eu is Eu Sample/Eu PAAS and Eu* is a predicted value obtained by the linear interpolation between shale-normalized Sm and Gd values.

$$Eu^* = \frac{La_{Sample}/La_{PAAS}) + (Pr_{Sample}/Pr_{PAAS})}{2}$$

Eu/Eu* values >1, <1 and 1 indicates positive, negative and no anomaly, respectively.

Eu/ Eu* ratios calculated in the sediments of Cochin estuary and adjacent coastal station was given in Table 4.6. The estuarine and coastal sediments showed Eu/Eu* > 1, the values varies as 1.06 to 2.23 showing positive Eu anomaly indicating a dominant terrestrial source. High values were observed at S₅, S₆ and S₇. This was consistent with the results of earlier studies (Nath et al., 2000; Laluraj, 2009; Deepulal et al., 2012). The positive Eu anomaly occurs in areas prominent in plagioclase-rich charnockites and gneissic terrain of southern India (Allen et al., 1985) and also for the rocks of feldspar origin (Taylor and McLennan, 1985). Eu anomaly is controlled by supply from the rivers, biological productivity, and redox condition at the sampling stations. The sediments derived from moderately weathered sediments show a positive Eu anomaly (Manoj et al., 2016). The scavenging of REE by organic coatings and Fe/Mnoxyhydroxides can increase the REE in sediments (Shynu et al., 2013). During the present study high positive Eu anomalies observed at stations (S₅, S₆ and S₇) were due to the depletion of other REE (including Sm and Gd) relative to PAAS. The sandy texture with low clay and organic matter could be responsible for low REE in these regions indicating that sediment texture is a major factor controlling geochemistry of the study area.

4.4.6.3 Uranium as paleo redox proxy:

Due to the enrichment in the suboxic and anoxic waters, U is generally used as an indicator to identify redox environments (Nath et al., 1997 and references there in). Since Th is immobile at low temperature, U/Th ratio is sensitive to changes in U concentration, which varies with redox state of the water. Under oxidized condition, uranium will be in +6 oxidation state (U^{6+}) which is soluble and lead to its release from sediments to lower the U/Th ratios. Under reduced conditions, U^{4+} get deposited in sediments elevating U/Th ratios. The U/Th ratios (Table 4.6) in the sediments of Cochin estuary and adjacent coastal stations were < 1 (0.11 to 0.37) indicating a fairly oxidized environment.

Authegenic Uranium, a parameter based on U and Th concentrations has been suggested as an index of bottom water anoxia and ancient sedimentary sequences (Nath et al., 1997). It is calculated using the formula,

Authigenic U= (Total U- Thorium/3).

Nath et al., 1997 and Wignal and Myers, 1988 place the boundary between oxic and dysoxic environment at 2ppm of authigenic uranium. The sediments of Cochin estuary and adjacent coastal stations showed authigenic uranium less than 1ppm.

Table	4.7:	Rare	earth	elements	(Lanthanides,	Sc	&	Y)	and	Actinides	5
		(Thor	ium ar	nd Uraniu	m) in the sedir	nents	s of	Co	chin (estuary - a	ł
		comp	arison	with prev	ious studies (va	lues	are	giv	en in	mg/kg).	

Elements	Nath et al, 2000	Shajan, 2001	Laluraj, 2009	Deepulal et al, 2012	Present study
La	47 - 78	-	32.6 - 96.3	28.5 - 52.4	24.26 - 102
Ce	103 - 174	-	75.9 -171.5	50.8 -102.9	44.26 - 171.28
Pr	-	-	7.6 - 16.6	4.9 - 10.0	4.33 - 17.37
Nd	34 - 86	-	21.2 - 70.3	18.6 - 38.8	18.13 - 70.49
Sm	8.05 - 11.8	-	3.5 -11.9	3.6 - 6.9	3.41 - 11.11
Eu	1.84 - 2.49	-	1.0-2.2	1.25 - 1.59	1.16 - 2.18
Gd	-	-	3.8-9.4	2.75 - 5.65	2.39 - 8.05
Tb	0.85-1.44	-	0.3-1.4	0.44 - 0.89	0.38 - 1.23
Dy	-	-	1.8- 6.7	2.32 - 4.92	2.12-6.63
Ho	-	-	0.3-1.3	0.46 - 1.03	0.43 - 1.31
Er	-	-	0.92-3.9	1.3 - 2.8	1.3 - 3.85
Tm	-	-	0.11- 0.53	0.18 - 0.38	0.18 - 0.60
Yb	2.12 - 3.21	-	0.61-3.0	1.08 - 2.34	1.10 - 3.16
Lu	0.31 -0.45	-	0.1- 0.47	0.15 - 0.36	0.17 - 0.47
Sc	19.9 - 27.1	3.0 - 26.0	6.1 - 21	-	7.72 - 20.62
Y	-	2.0 - 27.0	16 - 44.6	-	12.62 - 38.90
U	4.11 - 19.3	0.5 - 3.1	1.7-9.3	1,09 - 2.80	0.95 - 6.99
Th	12.2 - 18.3	3.7 - 12.2	12.5-31.6	6.34 - 11.31	6.62 - 35.61

*The studies of Nath et al., 2000 was carried out in finer sediments ($<4\mu$ m), while all other studies used bulk sediments: The data obtained from Laluraj, 2009 was the average value for central and north estuary.

The geochemical proxies such as Ce/Ce^{*}, Th/U ratios and authigenic uranium was used to identify the redox conditions prevailing in the study region. The results showed that an oxygenized condition prevailed in the estuarine and coastal region, except at S_1 , where high uranium with a negative Ce anomaly was indicative of a reducing environment. Eh of sediment in this region was (-150 to 50) also showing moderately reducing condition. However, the bottom saturation (DO >4.53mg/l) with mild redox conditions (U/Th 0.20 and authigenic uranium < 1ppm) are supportive of an oxidized environment. Therefore, the Ce anomaly at S_1 needs further explanation.

Elements	Nath et al, 2000	Shajan, 2001	Laluraj, 2009	Deepulal et al, 2012	Present study
La	51 - 61	-	34.9	13.95 - 21.9	44.02 - 55.73
Ce	105 - 151	-	60.9	27.7 -50.9	91-102.05
Pr	-	-	6	2.8 - 4.8	8.85 -10.25
Nd	42 -78	-	22.4	11.1 - 19.7	37.68 - 43.08
Sm	8.86 - 10.8	-	3.6	1.9 - 3.6	7.01 - 7.56
Eu	1.92 - 2.22	-	0.58	0.54 - 0.79	1.78 - 1.91
Gd	-	-	2.8	1.37 - 3.05	5.62 - 6.19
Tb	0.91 - 1.5	-	0.41	0.19 - 0.49	0.90 - 1.05
Dy	-	-	1.9	1.01 - 2.65	4.94 - 5.89
Но	-	-	0.36	0.21 - 0.52	0.82 - 1.20
Er	-	-	1.1	0.53 - 1.41	2.82 - 3.49
Tm	-	-	0.13	0.08 - 0.18	0.39 - 0.48
Yb	2.23 - 2.87	-	0.79	0.5 - 1.1	2.3 - 2.87
Lu	0.34- 0.55	-	0.12	0.08 - 0.18	0.33 - 0.40
Sc	21.3 - 24.8	12.0- 59.0	9.61	-	14.92 - 18.8
Y	-	7.0 - 24.0	13.6	-	26.75 - 33.98
U	2.92 - 4.68	0.27 - 2.56	2.1	1.01 - 1.47	2.2 - 3.31
Th	12.1 - 14.9	1.29 - 10.72	16.5	5.53 - 5.74	10.95 - 12.66

Table 4.8: Rare earth elements (Lanthanides, Sc & Y) and Actinides (Thorium and Uranium) in the coastal sediments - a comparison with previous studies (values are given in mg/kg)

*The data obtained from Laluraj, 2009 was the average value for near shore region, south of Kerala

4.4.7 Fractionation of REE (Lanthanides, Sc, Y)

The fractions of REE in the estuarine and coastal sediments (Fig 4.3-4.18) showed that the maximum LREE (La, Ce, Pr, Nd and Sm) concentrations associated with residual fraction. The contribution of LREE in the residual fraction (Fr.4) of the estuarine sediments (%) was La (56 to 84.8), Ce (47.8 to 76), Pr (34.6 to 76.3), Nd (43.7 to 74) and Sm (17.3 to 84.9) respectively. The percentage of LREE in residual fraction (Fr₄) in the coastal sediments varied as La: (61.4 to 75.1)% ; Ce : (55.1 to 55.5)%; Pr : (44.7 to 61.6)%; Nd: (60.4 to 67)%; Sm : (56.4 to 72.2)%.

Organic/sulphidic binding phase (Fr.3) was the second dominant fraction of LREE in the estuarine and coastal sediments. The LREE in organic/sulphidic binding phase (Fr 3) varied from 4.56 to 62.3% in estuarine sediments and 12.96 to 24.2% in coastal sediments. The Fe-Mnoxyhydroxide phase (Fr.2) was lower than organic binding phase for LREE. The LREE in Fe-Mnoxyhydroxide binding phase (Fr 2) varied from 8.1 to 20% in estuarine sediments and 9.2 to 13.3% in coastal sediments. The exchangeable/carbonate bound fractions (Fr 1) of LREE was the lowest fraction in both estuarine and coastal sediments and which account for only 0.69 to 7.67% (estuary) and 0.76 to 1.7% (coast) of LREE. Scandium showed a pattern similar to that of LREE in its fractionation.

The middle rare earth (MREE) and heavy rare earth elements (HREE) in the residual fraction (Fr 4) of both estuarine and coastal sediments were relatively lower than LREE. The MREE and HREE contents in residual fraction (Fr 4) in the estuary varied as Eu: (15.2 to 84.9)%; Gd : (10.4 to 66.3)%; Tb : (8.7 to 65.4)%; Dy: (3.9 to 63.4)%; Ho : (4.4 to 62.6)%: Er (8.7 to 63.6)%; Tm : (14.2 to 65.3)%; Yb : (19.4 to 66.9)% and Lu: (18.7 to 68.3)% respectively. The MREE and HREE in residual fraction (Fr.4) in the coastal region was lower than estuarine sediment and varied as Eu: (36.1 to 61)%; Gd : (33.1 to 50.8)%; Tb : (31.6 to 49.3)%; Dy: (30.7 to 45.4)%; Ho :

(28.3 to 43.3)%: Er (30 to 44.7)%; Tm : (32.7 to 44.8)%; Yb : (3.9 to 39.1)% and Lu: (35.4 to 47.4)% respectively.

The maximum fraction of heavy rare earths, HREE (Tb, Dy, Ho, Er, Tm, Yb and Lu) was found in organic/sulphidic binding phase (Fr.3) in the estuarine sediments of stations S1, S2, S3, S4, S8 and coastal sediment of stations S₉, S₁₀, S₁₁, S₁₂. Sediments in these stations have high organic matter with clayey texture that could favour the strong association of HREE on to it. However in other stations with sandy sediments (S_5, S_6, S_7) the HREE was intensely found in residual binding phase. The percentage of HREE in organic/sulphidic binding phase (Fr.3) varied from 6.8 to 68% in estuarine sediments and 24.5 to 75.7% in coastal sediments. The Fe-Mnoxyhydroxide binding phase (Fr.2) for HREE was lower than organic binding phase. The percentage of HREE in Fe-Mnoxyhydroxide phase (Fr.2) varied from 10.0 to 27.7% in estuarine sediments and 12.1 to 25.8% in coastal sediments. The exchangeable/carbonate bound fractions (Fr.1) of HREE were the lowest fraction found in both estuarine and coastal sediments and which account for only 1.0 to 10.7% of HREE in estuarine sediments and 1.0 to 1.9% of HREE in coastal sediments. Yttrium showed the pattern similar to that of heavy rare earths in its fractionation. Even though REE were mostly concentrated in clayey sediments no significant correlation exists between dominant clay minerals (kaolinite & smectite) and total REE ($R^2 < 0.5$). Perhaps, this could be due to the scavenging of REE by organic matter/organic coatings and Fe-Mn oxyhydroxides in the study region. However REE in Fr 1 showed a linear decrease with increasing kaolinite showing the more available binding sites on Smectite for REE





Fig 4.3: Total scandium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.4: Total yttrium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.5: Total lanthanum and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.6: Total cerium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.7: Total praseodymium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.8: Total neodymium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.9: Total samarium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.10: Total europium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.11: Total gadolinium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.12: Total terbium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.13: Total dysprosium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.14 : Total holmium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.15: Total erbium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.16: Total thulium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.





Fig 4.17: Total ytterbium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.



Fig 4.18 : Total lutetium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011.

4.4.8 Fractionation of Actinides (Th and U)

The fractions of actinides in the estuarine and coastal sediments (Fig 4.19-4.20) showed (86.5 - 99.5)% of Thorium in the residual phase followed by organic phase (0.1-7.8)%. The thorium in Fe/Mn oxyhydroxide and exchangeable/carbonate phases were the lowest (0.0 to 0.2)% in both estuarine and coastal stations.





Fig 4.19: Total thorium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011





Fig 4.20: Total uranium and its four different fractions in the sediments of Cochin estuary and adjacent coastal region during October 2011

Organic/sulphidic phase (Fr.3) was the preferred fraction for Uranium (55.1 to 76.4 %) in the estuary at S_1 , S_2 , S_3 , S_4 , S_7 and S_8 due to high organic matter. Stations with low organic matter and sandy sediments ($S_5 \& S_6$) exhibited U (41.7 to 45.5)% predominantly in residual phase (Fr 4). U in exchangeable/carbonate (Fr 1) and Fe/Mn oxyhyrdoxide (Fr 2) phases were poorer than Fr 3 and Fr 4 and varied as (1.4 - 8.0)% and (2.7 - 11.0)% in

estuarine sediments. The estuarine sediments showed similar fractionation pattern for uranium and HREE.

Although the coastal sediments were enriched in organic matter, U was preferentially found in the residual phase (56.5 - 63.5)% as compared to the organic phase (20.1 - 26.8%). The uranium in Fe/Mn oxyhydroxide was much lower (1.3 - 2.8)%. The exchangeable/carbonate phase (Fr.1) was the lowest fraction (1.2 - 2.4%) in coastal sediments. The present study shows that organic matter is not a factor controlling the distribution of uranium in coastal sediments. In the estuarine sediments however, the enrichment of uranium was strongly depend upon the organic matter and clay in the sediments.

4.5 **DISCUSSION**

The sedimentary REE distribution can provide very useful information on the geological and oceanographic processes of a region (Piper, 1974; Murray and Leinen, 1993). There are many studies to reveal the source and depositional pathways of REE in Cochin estuary and its adjacent coastal region (Nath et al., 2000; Siby, 2004; Laluraj, 2009; Deepulal et al., 2012). The present study is a first time attempt to carry out the fractionation of rare earths, thorium and uranium in the sediments of Cochin estuary and adjacent coastal region to understand the fate, transport and depositional characteristics of these elements. Ce was markedly high in sediments of estuarine and coastal stations as it was the most abundant rare earth in earth crust. Lutetium was the least abundant rare earth present in the estuarine and coastal sediments. Enrichment of rare earth elements (REE) in the north estuary (S₁, S₂) and a gradual decrease towards the central estuary (S₃, S₄, S₇, S₈) and to a lowest level near the Cochin inlet (S₅ & S₆) are some of the notable observations. Further towards the coast, the REE showed a gradual increase and the values similar to the central estuary. This generally indicates the mobility of elements from riverine sources downstream. The deviations in normalized pattern in the north estuary from the global shale could be due to external input. The dynamic nature of the inlet may support swiping of finer particles leading to impoverishment of elements in the sediments. This is consistent with the earlier results showing swift currents at Cochin inlet restricts materials to settle down (Srinivas et al., 2004; Laluraj, 2009). Similar kind of observations with near shore enrichment in the coastal region was reported in previous study (Laluraj, 2009).

The concentration of rare earths, uranium and thorium in estuarine sediments was comparable with the earlier studies (Laluraj, 2009; Deepulal et al., 2012) (Table 4.7 & 4.8). The higher levels of rare earths, uranium and thorium in the coastal sediments during the present study may be due to a corresponding increase in the clay content. Earlier studies (Laluraj, 2009; Deepulal et al., 2012) carried out in coastal region have mostly noticed a sandy texture in this region. Therefore, sediment texture plays an important role in the distribution of rare earths, uranium and thorium in the coastal environment. The fine-grained sediments have greater surface area and can provide more efficient environment for the adsorption of these elements (Wasserman et al., 1998).

Geochemistry of the REE in the Cochin estuary and adjacent coastal region show that there is an enrichment of light rare earth elements (LREE) compared to middle rare earth elements (MREE) and heavy rare earth elements (HREE), which is consistent with previous studies from this region. The LREE enrichment in the sediments of Cochin estuary and adjacent coastal region was preferentially linked to greater felsic granitic compositions (Nath et al., 2000; Laluraj et al., 2009; Deepulal et al., 2012).

Ce anomaly in the study region showed the terrigeneous nature of sediments under oxidized conditions. The slightly negative Ce anomaly in the north estuary may be associated with the biological productivity in the region (Manoj et al., 2016). The feldspars and their secondary products, which are enriched in Eu, might be the cause of a slight positive Eu anomaly in the study region, where the terrain is characterized by plagioclase-rich charnockites (Allen et al., 1985) and feldspar (Taylor and McLennan, 1985). The positive Eu anomalies can occurs in moderately weathered sediments and the Cochin estuarine sediments can be considered as moderately weathered (Manoj et al., 2016).

The fractionation studies were carried out to understand the mechanism behind the deposition of REE, and actinides (Thorium and Uranium) in the Cochin estuary and adjacent coastal region. It was observed that residual phase was the major binding phase for LREE. However for HREE, the preeminent binding phase varied with organic matter in the sediments. It was observed that the preeminent binding phase for HREEs was organic/sulphidic phase in sediments with high organic matter. However, in sediments with low organic matter, HREE was preferentially found in residual phase. Scandium showed fractionation similar to LREEs, while yttrim showed fractionation similar to HREE due to their identical chemical properties. These kinds of fractionation of REE are crucial in their enrichment in the sediments. The high thermodynamic stability of LREE in residual phase could be a reason for its enrichment in sediments. The stability of HREE depends upon the nature of organic compound to which it is associated. The source of

organic matter is primarily detritus in nature. The thick tropical vegetation in the catchment areas of rivers and estuaries, mangrove vegetation, high primary productivity, seasonal hypoxia and fine particle size of sediments are responsible for higher concentration of organic matter in the estuary and shelf (Manju et al., 2012; Gireeshkumar et al., 2013). The metal binding to organic matter remains complex with ion exchange and surface adsorption being the major pathways (Brown et al., 2000). It is clear from the study that organic matter content and grain-size are significantly affecting the rare earth concentration in the study area. There are reports that Fe/Mn oxyhydroxides are major carriers of REE in marine sediments and become unstable during fluctuating Eh/pH in the OMZ (Palmer and Elderfield, 1986). The present study shows that the association of REE with Fe/Mn oxyhydroxide was relatively low compared to that in residual and organic phases. The strong neagative correlation between kaolinite and REE in Fr 1 shows that REE in Fr 1 was mostly associated with Smectite.

Thorium in the present study was mainly concentrated in the residual phases. Thorium is generally resistant to weathering and alteration processes (Taylor and McLennan, 1985; Nath et al., 1997). It is significantly present in the minerals, zircon and monazite sands of Kerala. High concentration of thorium in the northern estuary can be due to industrial input.

From the results, it is clear that the fractionation of uranium was similar to that of HREE in estuarine sediments. More specifically, the enrichment of uranium in estuarine sediments was strongly dependent on the organic matter. The estuarine sediments with lower organic matter and low clay content contained low uranium content, while sediments with high organic matter showed high uranium content. Conversely, the coastal
sediments containing high organic matter preferentially accumulated uranium in a relatively resistant residual phase. Therefore, organic matter does not support the deposition of U. The presence of high uranium in coastal stations can also be due to bio-mineralization. Uranium was found to precipitate as stable uranyl phosphate mineral along the south west coast of India (Sreedhar et al., 2014). Phosphate solubilizing bacteria present in the monazites of south west coast of India have the ability to release phosphate for uranium mineralization (Desouza et al., 2000; Sreedhar et al., 2014). The present study shows that factors such as organic matter, Fe/Mn oxyhydroxides, sediment texture and mineral surface area are all contributing to the enrichment of REE in the sediments, which in turn, depends on the redox conditions, ion exchange capacity of clay minerals, rate of sedimentation/ flocculation and rate of weathering.

4. 6 SUMMARY AND CONCLUSION

The present study is a first time attempt to carry out the fractionation of rare earths, thorium and uranium in the sediments of Cochin estuary and adjacent coastal region to understand the fate, transport and depositional characteristics of these elements. Rare earth element (REE) geochemistry of sediments from Cochin estuary and adjacent coastal stations showed that the light rare earth elements (LREEs) are the more abundant compared to middle rare earth elements (MREE) and heavy rare earth elements (HREE). Cerium was markedly high in sediments of estuarine and coastal stations as it was the most abundant rare earth in earth crust. Lutetium was the least abundant rare earth present in the estuarine and coastal sediments. Rare earth elements (REE) were concentrated in the northern stations and showed a gradual decrease towards the central estuary and lowest concentration was observed near the Cochin inlet with sandy sediments. The dynamic nature of the inlet may support swiping of finer particles leading to impoverishment of elements in the sediments. The coastal sediments showed high concentration of rare earths and the values were similar to the central estuary. Shale normalized values of rare earths showed felsic source. A deviation in shale normalized pattern observed at northern estuary can be due to additional supply of REE into the estuary in addition to detritus. High thorium content in theses station too supports the external source.

Positive Ce anomaly observed in the study region supports the terrigeneous origin. A slight negative Ce anomaly at northern station could be due to biological productivity. No anomaly or slightly positive Eu anomaly in both estuarine and coastal sediments supports its feldspar origin. Redox constraints like U/Th ratio and authigenic U supports the oxygenated conditions in both estuarine and coastal sediments. Sequential extractions show that LREE was mostly associated with residual phase, while HREE was preeminently found in organic phase in organic rich clayey sediments and in residual phase in organic poor sandy sediments. The high thermodynamic stability of LREE in residual phase could be a reason for its enrichment in sediments. The stability of HREE in sediments depends upon the nature of organic compound to which it is associated. Thorium in the sediment was stabilized in the residual phases.

The organic matter present in estuarine environment makes uranium immobile in sediments, while in the coastal environment, the biomineralization seems to be responsible for its accumulation in residual phase. The association of REE with Fe/Mn oxyhydroxides and •

exchangeable/carbonate phases was relatively low compared to that in residual and organic phases. The study thus highlights the major role of particle size and organic matter in controlling the deposition of REE in the estuarine and coastal environments.

****\$\$****



Chapter **5**

INPUT AND EXPORT FLUXES OF TRACE METALS IN THE COCHIN ESTUARY

5.1 Introduction

5.2 Study region and Methodology

5.3 Results and Discussion

5.4 Summary and Conclusion

5.1. INTRODUCTION

Rivers are by far, the most important media for the transport of solids to the sea. Estuaries act as a transition zone, where a major part of the continental weathering materials are trapped and the remainder is transported to the sea (Aston, 1978; Salomons and Forstner, 1984). Besides, estuaries also act as geochemical filters of toxic materials and pollutants (Kennish, 1991; Spencer et al., 2006) that enhance the deterioration in estuaries. One class of such pollutants comprises the trace metals. The trace metals are naturally found in their background levels on earth's crust such as Shale's, Sandstones, and Metamorphic and Igneous rocks (Benjamin and Honeyman, 2000). The release of metals from the crustal rock is influenced by natural events such as wind and rain. chemical weathering, notwithstanding large-scale anthropogenic disturbances such as mining, construction and coal burning.

Thus, the major sources of trace metals to aquatic systems are rivers, atmospheric fallout and anthropogenic activities. Since these elements are persistent in the sediment they are also delayed sources of contamination to the aquatic systems (Kennish et al., 1991; Wood et al., 2012; Machado et al., 2014).

The major types of rock in the study area are crystalline rocks of Archean age, sediments of tertiary age and laterite capping on crystalline rocks and sediments of sub-recent to recent age (Rao et al., 1983; Mallik et al., 1987). Charnockite, Khondalite, Granite gneiss, and Granites are the dominant rocks and at places, they are traversed by basic rocks. Charnockite is widespread in the hill ranges of the Western Ghats from where the rivers of Kerala originate. The heavy mineral suite of the coastal and river sediments of Kerala consists of opaque, hornblende, hypersthene, tremolite/actinolite, pyroxene, epidote, zircon, sillimite, kyanite, staurolite. andalusite. monazite, rutile, sphene, apatite and tourmaline (Reddy and Rao, 1992). The light minerals consist of quartz, feldspar and some mica, while the opaque trace minerals include ilmenite, a little magnetite, rutile, spinel and leucoxene. In the lowlands, the soils vary from sandy to clayey and the erosion status is slight to moderate in the Achankovil, Muvattupuzha and Periyar river basins.

When trace metals enter an estuary, they are exposed to an environment where the chemical and physical properties (pH, redox, surface area and ionic strength) undergo rapid changes (Wen et al., 1999). The important estuarine processes are the precipitation, adsorption and desorption. The chemically reactive parts of suspended particles are normally organic matter (humic/fulvic acid), oxides of Fe, Mg, Al, Si and charged clay. The binding sites on them are capable of scavenging metals from solution. There are several physical (columbic interactions in the surface) and chemical (interior covalent bonds) factors controlling the strength of metal attachment to the particles. The partitioning of trace metals between dissolved and particulate phases is dependent on the in situ processes such as coagulation, flocculation, re-suspension, bioturbation and sedimentation (Santschi et al., 1997). Besides, major meteorological events like storms, high tidal amplitudes and an increased river flow can also alter the partitioning of metals in estuaries.

Desorption of trace metals from particulate material is also an important process in estuaries. Luoma (1983 &1989) had shown that metals present in the particulate form in the aqueous phase are more bio-available than the dissolved metals. Thus, for a real assessment of trace metal pollution, the concentration of metals in suspended matter is an obligatory parameter in addition to their concentrations in dissolved fraction (Church, 1986). More than 90 % of suspended particles brought by rivers are deposited in estuaries and coastal zones, thus only a small percentage reaches the open sea. Therefore, the ultimate sink of most trace metals discharged into the aquatic environment is the estuarine sediment. Estuaries are normally transport zones of both dissolved and particulate matter to the oceans and hence, sources and distribution of trace metals in estuaries and rivers are very essential to evaluate the geochemical budget (Riley and Chester, 1971). Since estuarine systems are dynamic due to strong chemical and physical gradients, studies on trace metal cycling is more difficult in estuaries compared to other aquatic systems (Morel et al., 1991; Millward and Turner, 1995).

The Cochin estuary is one of the largest tropical estuarine systems along the west coast of India. The hydrography of the Cochin estuary is

greatly influenced by tides, freshwater water flow and estuarine circulation (Revichandran et al., 2011; Shivaprasad et al., 2013). The flow characteristics and the mixing pattern of Cochin estuary are very peculiar and have great influence in the bio-resources, transportation and waste disposal (Balachandran et al., 2005). The fluvial transport was found to be a perennial and major process behind the evolution and morphology of the Vembanad Lake, also known as the Cochin Estuary (Qasim, 2003). Increased discharge of industrial effluents through the River Perivar has changed the northern estuary into a heavily polluted region, while the sluggish south estuary receives domestic and agricultural effluents from upstream (Kuttanad agricultural fields). Human interventions have led to wide spread reclamation of the estuary, while the construction of hydraulic barriers in the northern and southern parts have created severe flow restrictions and increased sedimentation in the estuary (Balachandran et al., 2005; 2008c).

Discharge of effluents has substantially increased the concentrations of many of the dissolved metals in the estuary (Anju et al., 2011). The concentrations of Zn, Cd and Ni in the estuarine sediments have enriched to such an extent that Cochin Estuary is placed among the most impacted estuaries in the world (Balachandran et al., 2005; Martin et al., 2012; Renjith et al., 2016; George et al., 2016; Salas et al., 2017). There are also reports of minor pollution with respect to Fe, Mg, Cr and Mn in the CE and adjacent coastal waters (Bindu et al., 2015). Hg levels in Cochin estuary (up to 3.0mg/kg in fish) was higher than similar kind of fish caught from other weltands in India (Ramaswamy et al., 2017). Realizing the pressures on Cochin estuary, previous study suggested for adequate conservation plans to minimize the inevitable pollution, consequent to developmental activities

(Ramaswamy et al., 2017). The increase in the Zn concentration in CE has also altered the community of cyanobacteria in the Cochin Estuary (Anas et al., 2015). Therefore, an efficient estuarine management is required based on the activities and pressures operating on this ecosystem (Elliotta and McLusky, 2002; Millennium Ecosystem Assessment, 2005; European Commission, 2012). In view of this, the present study proposes to identify the source, fate and transport of trace metals in the Cochin estuary.

5.2. STUDY REGION AND METHODOLOGY

5.2.1. Regional Setting

Metal fluxes through the Cochin estuary can be quantified by measuring their respective fluxes across the river end stations and at Cochin inlet, where exchange of materials between the estuary and sea occurs. A total of 14 locations were selected for the present study to identify the source, fate and transport of trace metals in the Cochin estuary (Fig 5.1). The geographic locations of sampling sites (Table 5.1) are labeled from S_1 to S_{14} . They are S_1 (Neeleswaram), an upstream site of River Periyar near to Central Water Commission (CWC) Gauge station. S₂ is at Purappallikkavu where a temporary sand barrier constructed prevents salt water intrusion during premonsoon period (During the present sampling period, the bund construction at S_2 was in progress). S_3 is at Moolampilly where River Periyar enters the Cochin Estuary (CE) after receiving effluents from the industrial area (Eloor). S_5 is at Vettikkattumuku in the upper reaches of the River Muvattupuzha. Since this river bifurcates before its entry to the CE, samples were collected from these two branches of Muvattupuzha i.e S₆ (Murinjupuzha) and S₇ (Ithipuzha). S_8 (Kariyar) is another branch of Muvattupuzha river entering the north Kuttanad region. The major activities in this region are clam shell

processing, paddy cultivation and fish farming. There were lot of weeds in the Ithipuzha and Kariyar during the sampling period. Thanneermukkam barrier remains closed from December to April every year, during which, the southern part of the estuary remains almost stagnant. S_9 is Kidangoor, an upstream end of River Meenachil (CWC Gauge station). S_{10} is at Kumarakom, the southernmost part of the Vembanad Lake where River Meenachil enters the CE. S_{11} is at Pallathuruthy in south Alappuzha through which, River Pamba enters the Kuttanad agricultural fields. S_{12} is at Keezhvaypur (CWC Gauge station) in the River Manimala (tributary of River Pamba). S_{13} is at Aranmula (upstream end of River Pamba) and S_{14} is at Mudiyyoorkonnam (in the River Achankoil). This region was turbid during the sampling period.

Sl No	Station name	Station code	GPS coo	rdinates
1	Periyar	S_1	10° 10' 22.3"N	76° 27' 38.1"E
2	Purappillikkavu	S_2	10° 8' 39.2"N	76° 16' 59.9"E
3	Moolampilly	S_3	10° 2' 17.6"N	76° 16' 8.2"E
4	Cochin inlet	S_4	9° 58' 15.3"N	76°14 ' 42.1"E
5	Muvattupuzha	S_5	9° 48'21.1"N	76° 27' 14"E
6	Murinjupuzha	S_6	9° 49'17.3"N	76° 23' 31.8"E
7	Ithipuzha	S_7	9° 46'50.6"N	76° 22' 29.6"E
8	Kariyar	S_8	9° 42' 5.8"N	76° 24' 10.9"E
9	Meenachil	S_9	9° 40' 34.6"N	76° 36' 33.9"E
10	Kumarakom	\mathbf{S}_{10}	9° 35' 47.8"N	76° 25' 46.3"E
11	Pallathuruthy	S_{11}	9° 27' 47.2"N	76° 21' 46.5"E
12	Manimala	S ₁₂	9° 25' 55.9"N	76° 40' 3.4"E
13	Pamba	S ₁₃	9° 19' 43.6"N	76° 41' 6.2"E
14	Achankoil	S_{14}	9° 14' 2.7"N	76° 38' 20.6"E

Table 5.1: Details of sampling sites selected for the study

Dynamics of trace metals, rare earth elements and nutrients in the CE



Fig 5.1: Cochin estuary and rivers discharging into the system; the red dots symbolize the station locations for the present study.

Dynamics of trace metals, rare earth elements and nutrients in the CE



Cochin inlet, (S_4) is a region where considerable exchange of materials occurs between the sea and the estuary as it is the major opening to the sea. S_4 showed dynamic nature due to the opposing effect of tidal forcing and river discharge. The depth of the channel at S_4 is 10-15 m (Shivaprasad et al., 2013). There is strong stratification at the inlet during heavy discharge periods, while it gets eroded during post and pre monsoon periods, when well mixed condition prevails (Bincy et al., 2013; Shivaprasad et al., 2013).

5.2.2. Sampling and analysis

River sampling was carried out during the period $28-30^{\text{th}}$ October 2015. For the flux measurements, the instantaneous cross sectional flow through each of the six rivers (locations S₁, S₅, S₉, S₁₂, S₁₃ and S₁₄) were measured (one time measurement) using an Acoustic Doppler Current Profiler (ADCP-Teledyne River Ray-600 KHz) with bottom-track capability. The dissolved and particulate metal contents of water samples collected from the surface and near bottom were analysed and used in the computation of metal input fluxes through each rivers. Although the metal fluxes through other 4 locations (locations S₃, S₆, S₇ and S₈) were also measured during the same period, these values were not included while computing the net fluxes (input-export) through the estuary as they were located in the upper estuary.

Since the flushing time of Cochin estuary is ~2.5 days during this period (Vinita et al., 2015a), the measurements to quantify the export fluxes through the Cochin inlet were conducted after 3 days of the river sampling (during 3-4th November 2015). A tide gauge was moored at Cochin inlet to obtain the water level variations during the flux measurements. The metal fluxes across the Cochin inlet was computed based on the time series

observation for 27 hrs using ADCP and water sample collection at 3hr intervals from the surface, 7.5m and 15m depths. The water flux obtained from ADCP was extracted as surface, mid and bottom fluxes using MATLAB. The measured metal concentration and water flux at each depth was used to compute the instantaneous cross sectional transport, which on integration over the tidal cycle, gives the net fluxes across the inlet.

Temperature and salinity at each location (14 locations) were measured using a conductivity-temperature depth profiler (CTD 19 plus, Sea-Bird Electronics). The atmospheric conditions and sampling time were noted at all locations. Water samples were collected from the surface, mid and bottom using Niskin sampler (Hydrobios-Kiel) for pH, dissolved oxygen, heavy metals and suspended particulate matter. The samples for dissolved oxygen were fixed immediately after collection and later, analyzed according to Winkler's method (Grasshoff et al., 1999). pH was measured using pH meter (Eutech pH 2700, accuracy ± 0.001). The water samples for trace metals were kept in ice, brought to the lab and filtered through a pre-weighed 0.45micron millipore filter paper (dried at 60° C, kept in desiccators and weighed) to separate suspended particulate matter and the filtrate were preserved at pH <2 using con.HNO₃ until extraction. The extraction was done with APDC (2%), DDDC (2%) and chloroform at pH 4.5 (adjusted with ammonium acetate). The filter paper containing the suspended solids was washed with 5 ml of Milli-Q water (Owens et al., 1997) to remove sea salts and dried at 60° C in an air oven to constant weight. The filter paper together with the suspended solids was weighed and the difference in weight was taken as the weight of suspended particles (mg/l). Each filter paper was separately stored in polyethylene petry dishes at -20° C until the extraction for trace metals. The extraction was done by digestion with a mixture of HNO₃+HClO₄ (5:2) in Teflon vessels on a hot

Dynamics of trace metals, rare earth elements and nutrients in the CE

171

plate (110°C) to dryness (Balaram and Rao, 2002). The residue was redissolved in 2% ultrapure HNO₃ (with 1ppm of Rhodium as internal standard) and analyzed by Inductively Coupled Plasma Optical Emission Spectrometer (ICP-OES). Certified reference material was used to validate the method.

5.3. RESULTS AND DISCUSSION

5.3.1 General Hydrography

Trace metals are transported to the ocean by rivers in dissolved and particulate forms. In estuaries, where river water and coastal or oceanic waters of widely different compositions are mixed, strong gradients in chemical properties occur (Burton, 1976). These variations may affect the distribution of dissolved and particulate components of trace metals. Therefore, hydrographical parameters are vital tools while attempting to characterize the distribution pattern of trace metals during their transport and also in the partitioning of metals in different phases. The freshwater input, currents and tidal range are useful to explain the physical behavior and transport properties in an estuary. The fresh water discharges through 6 rivers into the Cochin estuary are depicted in Fig 5.2. The water level variations at Cochin inlet on 3rd-4th November 2015 are given in Fig 5.9. As the observation was made between a spring and neap tide, the maximum tidal range observed was 0.56m (Fig. 5.9). The salinity at the upstream locations varied from 0.022 to 0.024. The salinity at upper estuary (S₃, S₆, S₇ and S₈) varied from 0.032 to 0.450 (Fig 5.3). As the water column at the Cochin inlet was stratified, there exist strong vertical gradient in salinity. The salinity at the surface, mid and bottom waters varied between 3.23 to 31.68; 13.23 to 33.48 and 31.98 to 33.78 respectively (Fig 5.9).



Fig 5.2: Fresh water discharges (m3/d) into the Cochin estuary from 6 major rivers on 28th-30th Nov 2015



Fig 5.3: Distribution of Salinity and SPM (mg/L) observed at the rivers on 28th-30th Oct-15 and at the Cochin inlet (time series average) on 3rd-4th Nov-15.

Dynamics of trace metals, rare earth elements and nutrients in the CE



5.3.2 Suspended particulate matter

Suspended particulate matter (SPM) plays a major role in the geochemical cycling of elements. The composition and variability of suspended particulate matter in the river-ocean boundary are affected by many complex geochemical processes (precipitation and flocculation, desorption-adsorption etc.) and by physical processes (river flow, tidal energy, currents) and are, therefore, often difficult to interpret (Morris et al., 1978; Turner et al., 1991, 1994).

5.3.2.1 Suspended particulate matter in the upper and lower reaches of rivers

Suspended particulate matter (SPM) in the upstream varied from 3.88 to 66.7mg/L (Fig 5.3). SPM was very low in River Periyar and very high in rivers Achankoil and Pamba. SPM levels in the rivers generally followed an order Achankoil > Pamba > Manimala > Meenachil > Muvattupuzha > Periyar. SPM in the lower reaches of rivers varied from 6.8 to 18.9mg/L and followed the order Pallathuruthy > Kariyar > Kumarakam/Ithipuzha > Moolampilly > Murinjupuzha. An increasing trend in SPM was noticed towards the south of the estuary with maximum SPM at Pallathuruthy. Sedimentation, re-suspension and erosion processes may change SPM concentrations in the water column (Michaelis, 1990), influencing the trace metal distribution in the estuary (Luoma and Phillips, 1988; Vollmer et al., 1990).

5.3.2.2 Suspended particulate matter at the Cochin inlet

Suspended particulate matter at the inlet varied widely during a tidal cycle between 5.83 to 718.5mg/L (Fig 5.13). The maximum SPM in the surface was during the lowest ebb period (21-24hr) and the minimum SPM

was during the withdrawal of flood at 33hr (next day 9am). The SPM was generally low at mid-depth (12.45 to 102.9mg/L), though it was high (102.87mg/L) during flood tide at 30-33hr. SPM was always high in the bottom (9.13 to 701.14mg/L), especially during ebb tide (highest low tide) at 12hr and 39 hr (next day 3pm). The increase in SPM can be due to the estuarine mixing (Balachandran, 2001) indicating the effect of fluvial discharge and dredging.

5.3.4 Distribution of trace metals

5.3.4.1 Concentration of trace metals in the six rivers

There was a large variability in the distribution of metals among the dissolved and particulate phases of rivers. The concentration of metals in dissolved phase followed the trend Fe > Mg > Zn > Cr > Ni and that in the particulate phase followed an order Fe > Mg > Cr > Mn > Zn > Ni. Dissolved Fe in the rivers varied from 30 to $60\mu g/L$ (Fig 5.5) with maximum in the River Achankoil and minimum in the River Manimala. Dissolved Mg in rivers varied from 6 to 30µg/L with slightly higher concentration in Muvattupuzha and a lower concentration in Pamba and Periyar (Fig 5.5). The dissolved Zn in the rivers varied from 1.5 to $4.3 \mu g/L$ with maximum in Meenachil surface $(4.3\mu g/L)$ and minimum in the Manimala $(1.5\mu g/L)$ (Fig 5.4). Dissolved Zn was not present in measurable levels in the River Pamba during the study period. Traces of dissolved Cr (2.21µg/L) and Ni (0.086µg/L) was observed only in the River Periyar (Fig 5.6) and the concentration of dissolved Co, Cu, Cd, Mn and Pb were below measurable levels in all the 6 rivers. The dissolved metals in all the 6 rivers studied were comparable with the values reported by Central Water Commission (CWC), India and were well within the permissible and acceptable limit of drinking water recommended by Bureau of

Indian Standard (BIS 10500-2012) and World Health Organization (WHO). As the toxicity of metals in rivers were not solely depends on the dissolved fractions, particulate trace metals in river water were analyzed.



Fig 5.4: Distribution of dissolved Zn, Cr and Ni (μg/L) observed at the rivers on 28th-30th Oct-15 and at the Cochin inlet (time series average) on 3rd-4th Nov-15. (a) DZn; (b) DCr; (c) DNi.



Fig 5.5: Distribution of dissolved Fe, Mg and Pb (μ g/L) observed at the rivers on 28th-30th Oct-15 and at the Cochin inlet (time series average) on 3rd-4th Nov-15. (d). DFe; (e). DMg; (f). DPb.

The concentrations of all the metals analyzed in suspended particles were several times higher than in dissolved form. Fe in particulate phase

Dynamics of trace metals, rare earth elements and nutrients in the CE

varied from 3.8% to 11.2% with maximum concentration at the bottom of Periyar and minimum at the surface of Periyar showing weathering of hinterland rocks and re-suspension are the sources of particulate Fe into the rivers. Iron oxide is a major component of laterite, the major soil occurring in low and midlands of Kerala (Dept of Mining and Geology, DMG, Kerala, 2016). Thus, the high particulate Fe in rivers could be eroded from this polymetallic ore. Particulate Mn was present in appreciable levels in the rivers, Periyar (390 & 1970µg/g), Muvattupuzha (20 & 39.5µg/g), Pamba $(512.4 \& 89.4 \mu g/g)$ and Achankoil $(436.2 \& 694.9 \mu g/g)$ with maximum concentration in Periyar (Fig 5.8). Under highly oxygenated condition Fe and Mn exist as oxides and hydroxides which are insoluble in water leads to lower soluble Fe and absence of soluble Mn in all the six rivers. The presence of carbonates and sulphates in these rivers also support Fe precipitation (CWC, 2014). Zn in particulate phase varied from non-detectable level to $179.8 \mu g/g$ at the upstream with maximum concentration in Manimala. Particulate Zn was in non-measurable levels in Muvattupuzha and Periyar rivers (Fig 5.6).

Cr in particulate phase showed moderate values in the Rivers, Periyar (130.2 & $2613\mu g/g$), Muvattupuzha (45.1 & $537.8\mu g/g$) and Pamba (58.8 $\mu g/g$). Particulate Ni (188 $\mu g/g$) was present only in the River Meenachil. However the concentrations of particulate Co, Cu, Cd and Pb were below measurable levels in all the rivers. Mg in particulate phase varied from 0.24 to 1.2% with maximum concentration in Periyar and minimum concentration in Manimala. Mg²⁺ in the river waters are supplied by weathering of ferromagnese minerals including pyroxenes and amphiboles as well as dark colored micas (biotite), which are abundant in the host lithology (Thomas et al., 2015).



Fig 5.6: Distribution of particulate Zn, Cr and Ni (μ g/g) observed at the rivers on 28th-30th Oct-15 and at the Cochin inlet (time series average) on 3rd-4th Nov-15. (a) PZn; (b) PCr; (c) PNi.

Dynamics of trace metals, rare earth elements and nutrients in the CE

•





Fig 5.7: Distribution of particulate Fe, Mg (%) and Pb (μg/g) observed at the rivers on 28th-30th Oct-15 and at the Cochin inlet (time series average) on 3rd-4th Nov-15. (d). PFe; (e) PMg; (f) PPb.





Fig 5.8 : Distribution of particulate Mn, Cu and Co (μg/g) observed at the rivers on 28th-30th Oct-15 and at the Cochin inlet (time series average) on 3rd-4th Nov-15. (g). PMn; (h) PCu; (i) PCo.

The concentration of particulate Zn and Mg observed during the present study were comparable with the concentration reported in World Rivers Average SPM values (WRSPM) and that in Post Archean Australian Shale (PAAS) indicating the natural origin of these metals in all the 6 rivers. The concentration of particulate Fe, Cr, Ni and Mn was higher than that in the PAAS and WRSPM. Higher concentration of iron, manganese, nickel and chromium in particulate matter of these rivers could have derived from Laterite, a polymetallic ore in the low lands and midlands of Kerala enriched with these metals. The high Chromium in rivers might also be derived from various anthropogenic activities in river basins (e.g., fertilizer residues from farmlands and plantations, domestic sewage from settlement clusters). All the six rivers showed wide variation in their particulate trace metal distribution depending on the nature of alluvium that is deposited and the characteristics of the catchment area through which the river flows.

5.3.4.2 Distribution of trace metals in the lower reaches of rivers (upper estuary):

The distribution of metals in dissolved phase follows the sequence Fe > Mg > Zn > Ni > Cr (Fig 5.4 to Fig 5.5). Concentration of dissolved Zn (nd-67µg/L), Mg (0.006-0.01%) and Fe (0.04-0.15%) in the lower reaches of rivers were higher in Moolampilly (northern side) and Kumarakam (southern side) regions. Dissolved Ni and Cr were present only at the surface of Moollampilly (0.4 & 2µg/L) and Kariyar (1.0 & 2.3µg/L) suggesting their source from Periyar and Muvattupuzha rivers. Pb was not observed in any of the upstream regions while its presence in Ithipuzha (0.1µg/L) confirms its addition through non-point source. The possible sources of lead to this region may be derived from the pigments used in paints and from petroleum products and sewage discharge. Dissolved Cd, Cu, Mn and Co was not in measurable levels in the lower reaches of the rivers.

The distribution of metals in particulate phase at the lower reaches followed the trend Fe > Mg > Zn > Cr > Cu > Mn. Particulate Fe (μ g/g) showed an increasing trend from upstream towards the lower reaches for all the rivers except Muvattupuzha. In Muvattupuzha, the Fe concentration (4.5%) at upstream reduced to 3.4% and 3.7% towards the lower reaches (Murinjupuzha and Ithipuzha). However higher particulate Fe (16.2%), higher than in WRSPM and PAAS observed at Kariyar, north of Kumarakom can be possibly due to the sewage disposal and land use pattern. High particulate Fe in Pallathuruthy (5.5%), Kumarakom (6.3%) and Moolampilly (8.1%) regions indicate its additional input from human activities. Particulate Mn was lower at the lower reaches all the rivers.

Particulate Cr Purapillikkavu Moolampilly in $(209.7 \mu g/g),$ $(317.8\mu g/g)$, Kariyar $(447\mu g/g)$ and Kumarakom were higher than that in WRSPM (130 μ g/g) and PAAS (110 μ g/g). Particulate Cr was high in the Muvattupuzha river while it was absent towards the lower reaches, Murinjupuzha and Ithipuzha region showing its deposition while traversing towards the lower reaches (Fig. 5.6). A similar trend was noticed in the distribution of Cr in the Periyar too. A substantial reduction in the suspended solids towards the lower reaches due to meandering of Periyar River has been reported (Paul, 1980). Meandering reduce the flow velocity and support sedimentation. The distribution of particulate trace metals in these rivers can be generally attributed to the progressive sorting of minerals to which it is associated. During deposition, the denser minerals settle quickly, while the lighter minerals are transported further downstream. It is reported that due to the high competency of river water, settling of low gravity minerals do not take place in the upstream, while they increasingly get deposited towards

downstream (Seralathan, 1979; Lewis, 1984; Kundras, 1987; Padmalal, 1997). The strong association of Fe and Cr towards the denser particles during the present observation might have resulted in their rapid settling and lower particulate Fe and Cr at the lower reaches. Bio-accumulation can also be a factor for the absence of Cr in both dissolved and particulate phases of Ithipuzha and Murinjupuzha. The lower reaches of Muvattupuzha are rich in black clam and these filter feeders are capable of removing heavy metals from the aquatic systems (Cherian, 1990; Renjitha et al., 2011). Absence of Cr in Murinjupuzha, Ithipuzha, and Pallathuruthy shows that the transport of Cr into the estuary through this region was negligible. However, high concentration of particulate Cr in Kumarakom shows the anthropogenic addition from farm lands nearby.

The trace metals Zn, Cu and Pb at the lower reaches of rivers were several times higher than in the upstream regions. Particulate Zn was high at the lower reaches of Periyar and Muvattupuzha even though it was completely absent in the upstream. The concentration of particulate Zn in Moolampilly (1353µg/g), Sand bridge (64.8µg/g), and Ithipuzha (319.3µg/g) were higher than that in PAAS (85µg/g) and WRSPM (208µg/g) pointing out the anthropogenic addition of Zn from industrial areas nearby. The particulate Cu in Purappillikkavu was greater than that in PAAS (50µg/g) and WRSPM (75.9µg/g). High concentration of particulate Cu at Purapillikkavu (282.7µg/g) and Moolampilly (60µg/g) confirm their anthropogenic sources in the estuary. The particulate Pb at Purapillikkavu (126.4µg/g) and Kumarakom (84.9µg/g) was also higher than that in PAAS (20µg/g) and WRSPM (61.1µg/g). Murinjupuzha and Pallathuruthy were the least affected regions, where all the trace metals (Ni, Cu, Cr, Cd, Cu, Pb, Co and Mn) measured were completely absent except dissolved Zn in trace levels.

Dynamics of trace metals, rare earth elements and nutrients in the CE

5.3.4.3 Distribution and variation of trace metals at the Cochin inlet

The water level variations and salinity at Cochin inlet are given in Fig 5.9. The intra-tidal variation in both dissolved and particulate Mg, Fe, Cr, Zn, Cd, Ni and Mn at the Cochin inlet for a period of 27hrs at 3 hr intervals on 3rd-4th November 2015 are depicted in Fig 5.10-5.13.





Dynamics of trace metals, rare earth elements and nutrients in the CE

185



Fig 5.10: Intra-tidal variation in the dissolved (μ g/L) and particulate (μ g/g) Zn and Cr at the Cochin inlet on 3rd Nov 12.00hrs to 4th Nov 15.00hrs, 2015.

Dissolved Co, Cu and Pb were not present in measurable levels at the Cochin inlet, while other dissolved metals like Mg (28.8-947 μ g/L), Fe (9.7-125.3 μ g/L), Cr (2.14-5.55 μ g/L), Zn (0-6.15 μ g/L) and Ni (0-0.64 μ g/L) showed appreciable concentrations. The concentration of dissolved Mn and Cd were 10 μ g/L and 3.3 μ g/L respectively during one of the sampling time while it was non-detectable during all other time periods (Fig 5.10-5.13).

The particulate metals followed a sequence Fe > Mg > Zn > Mn > Cr >Ni at the Cochin inlet. The concentration of particulate metals increased with SPM (412 & 718mg/L), but Ni and Cr showed a depletion at the bottom due to their dissolution at higher salinity. Ni was mostly bound to particles when suspended particles increased at surface. Towards the bottom, at higher salinity Ni was mostly found in dissolved form owing it's desorption from particles at higher salinity. Cr attached to particles increased with SPM in surface shows that Cr is concentrated in riverine suspended particulate matter. Similar to Ni, desorption of Cr at higher salinity leads to more dissolved fraction of Cr at the mid and bottom of the water column. Particulate Mn showed dissolution at mid depth at 36hrs. High concentrations of particulate Fe, Mn, Cr, Ni and Pb in the bottom water at 12hrs were associated with high SPM during the ebb period following the re-suspension. Cu was completely absent in particulate matter at the inlet while, particulate Pb $(1.8\mu g/g)$ was present in the bottom during flood at 12hrs. Particulate Co was present only at the surface during the ebb tide at 24hrs, indicating its riverine source.



Fig 5.11: Intra-tidal variation in the dissolved (μg/L) and particulate (%) Fe and Mg at the Cochin inlet on 3rd Nov 12.00hrs to 4th Nov 15.00hrs, 2015.

Dynamics of trace metals, rare earth elements and nutrients in the CE

187



Fig 5.12: Intra-tidal variation in the dissolved (μ g/L) and particulate (μ g/g) Mn and Ni at the Cochin inlet on 3rd Nov 12.00hrs to 4th Nov 15.00hrs, 2015.

Fe in both dissolved and particulate phase displayed similar trend in its distribution. Dissolved Fe varied inversely with salinity suggesting its riverine origin. Particulate Cr, Ni and Co showed significant mutual correlations besides having good linear correlation with SPM (Table 5.2). Correlation of these trace metals with SPM suggests that these metals are concentrated through the rivers and re-suspension. Particulate metals (Fe, Zn and Mg) also showed strong inter relationship and have no significant correlation with other parameters represent similar sorbent for their association on to particles.





Fig 5.13: Intra-tidal variation in the particulate Co and Pb (μ g/g), dissolved Cd (μ g/L) and SPM (mg/L) at the Cochin inlet on 3rd Nov 12.00hrs to 4th Nov 15.00hrs, 2015.

5.3.4.4 Discussion

Higher suspended particles in the rivers (Meenachil, Manimala, Achankoil and Pamba) discharging south of the estuary may leads to higher SPM at the Kumarakom and Pallathuruthy. The high SPM at the Cochin inlet can be explained as due to the combined effect of river flux and re-suspension (Shibu et al., 1992). Mg showed an increase in the concentration towards downstream along with SPM concentrations due to higher adsorption with increase in salinity. Under oxidizing conditions, Mn exists as oxide and

hydroxide which are highly insoluble in water account for the absence of Mn in dissolved phases and high particulate fraction in rivers. A reduction in particulate Mn towards the lower reaches was due to their removal into the sediment. Cr was mainly bound with suspended particles in the upstream region however the increase in dissolved Cr towards the inlet region implies it's desorption from particulates at higher salinity.

With increased salinity at the inlet, elements like Na, K and Ca ions compete with trace metals for adsorption sites displacing both weakly and moderately sorbed metals such a Cr, Cu, Ni and Pb and release them to water column. The dissolution of Cr and Ni was noticed at higher salinity. However, metals such as Fe, Mn and Zn are relatively stable even at higher salinity may be due to their strong bond to particles. Therefore, the distribution of metals between the particulate and the dissolved phase at the Cochin inlet predominantly depends on the availability of sorbents and results from the operation of surface forces.

High particulate metals associated with maximum turbidity were reported in other ecosystems (Gonzalez et al., 2007). Moreover, the association of plankton with metals can alter partitioning of metal between water and SPM (Fisher et al., 1984). The minimum concentration of dissolved metals at the inlet can also be attributed to high chlorophyll a (bloom) during the present observation.

The enrichment of Fe in the upstream and lower reaches, both in suspended and dissolved form and its least abundance at the inlet region suggest the flocculation of Fe bound particles towards downstream, as observed early (Sankaranarayanan and Reddy, 1973; Nair et al., 2013). The flocculation processes greatly modify the metal concentrations and their partition between dissolved and particulate phases in estuaries (Bernhard et al., 1986).

Under low fluvial discharge, the salinity increases in the estuary which increases the dissolution and desorption of metals from particulate matter (Wang et al., 2009). Increasing river discharge however, implies high suspended material input or re-suspension and low salinity as evident from the present study at the inlet. This can increase the adsorption and precipitation processes resulting in a maximum particulate metal and minimum dissolved metal concentration.

A comparison of dissolved and particulate trace metals partitioned in the study area with previous reports showed (Table 5.3, 5.4) that particulate Cu and Fe have increased over the years (Paul and Pillai, 1983; Ouseph et al., 1992; Rajamani, 1994; Nair et al., 2013). However, particulate Mn was relatively lower than reported by Paul and Pillai, 1983, though it was higher than other reports (Ouseph et al., 1992; Rajamani, 1994; Nair et al., 2013). Dissolved fractions of all the metals were comparable with the earlier studies (Ouseph et al., 1992; Rajamani, 1994; Unnikrishnan, 2000; Balachandran, 2001; Nair et al., 2013; Anas et al., 2015).



ariables	salinity	P Zn	PCu	P NI	$P C_0$	P Mn	P Cr	P Pb	P Fe	P Mg	D Zn	D NI	D Fe	D Mn	DCr	D Mg	SPM	Hd
alinity	1.00																	
P Zn	-0.21	1.00																
PCu	-0.20	0.15	1.00															
ΝI	-0.03	0.01	-0.05	1.00														
P Co	-0.10	0.01	-0.03	0.71	1.00													
P Mn	-0.25	0.01	0.24	0.16	0.12	1.00												
P Cr	-0.25	0.19	0.21	0.13	0.10	0.11	1.00											
P Pb	-0.24	-0.06	0.81	-0.05	-0.03	0.22	0.59	1.00										
P Fe	-0.46	0.28	0.06	-0.16	-0.13	-0.02	0.10	0.12	1.00									
P Mg	0.46	0.06	-0.11	0.13	0.07	-0.15	-0.18	-0.17	-0.05	1.00								
D Zn	-0.25	06.0	0.11	-0.05	0.00	-0.07	0.13	-0.05	0.38	0.00	1.00							
D Ni D Fe	0.32	0.12	-0.07	-0.09	-0.15	-0.23	-0.09	-0.15	-0.09	0.37	0.09	1.00	100					
D Mn	-0.02	-0.06	-0.04	-0.06	-0.03	-0.08	0.21	-0.05	0.25	-0.09	-0.06	-0.16	-0.14	1.00				
D Cr	0.55	-0.01	-0.17	0.13	0.00	-0.24	-0.17	-0.24	-0.29	0.69	-0.11	0.24	-0.07	0.04	1.00			
D Mg	0.29	-0.03	-0.12	0.67	0.47	0.03	-0.02	-0.14	-0.27	0.35	-0.15	-0.09	-0.23	0.01	0.48	1.00		
SPM	0.15	-0.06	-0.07	0.69	0.64	0.11	0.05	-0.08	-0.39	-0.09	-0.09	-0.07	-0.15	-0.07	0.04	0.40	1.00	
Hd	0.86	-0.10	-0.18	0 11	0.03	9C UT	VCU	20.05	117	0.64	0.15	0.49	0.22	10.07	0.71	0.42	100	1 00

Rivers	Fe (µg/L)	Cr (µg/L)	Cu (µg/L)	Ni (µg/L)	Pb (µg/L)	Z n (µg/L)	Cd(µg/L)	Mg (µg/L)	Reference
Perivar	-		1.0-10	-	а	1.0-81	0.1		Paul & Pillai., 1983
Periyar	1.1	а	10	0.2	8	40	0.2	ī	Ouseph., 1992
Periyar	0.00-220	0.00-2.02	0.00-49.07	0.00-18.77	0.00-4.06	0-37	0.00-00.0		CWC, 2014
Periyar*	24-56	0.39-1.64	0.12 - 26.08	0.0-10.46	0.14-4.26	3.0-55	0.00-0.34		CWC, 2014
Meenachil*	28-540	0.35-1.30	0.68-13.63	0.00-7.78	0.08-4.17	1.0-67	0.00-0.32		CWC, 2014
Muvattupuzha	0.00-180	0.00-1.97	0.00-8.24	0.00-12.54	0.00-4.46	0.0-21	0.00-0.55		CWC, 2014
Muvattupuzha*	20-480	0.61-3.24	0.67-46.49	0.00-14.04	0.21-3.91	2.0-60	0.00-0.38		CWC, 2014
Pamba	20-490	0.14-2.06	0.28-33.47	0.00-8.58	0.03-7.15	2.0-44	0.00-0.09		CWC, 2014
Pamba	24-530	0.33-2.54	0.42-37.53	0.00-9.08	0.03-4.21	1.0-77	0.01-0.07		CWC, 2014
Pamba*	10-510	0.23-5.11	3.02-23.32	0.00-7.29	0.01-3.97	4.0-74	0.05-0.33		CWC, 2014
Periyar	31-47	2.29	0	0.086	0	17-26	0	5.8-9.6	present study
Muvattupuzha	37-44	0	0	0	0	1.0-2.0	0	8.4-25.6	present study
Meenachil	37-39	0	0	0	0	0.6-4.0	0	9.6-9.8	present study
Manimala	27-28	0	0	0	0	0.00-1.5	0	8.7-10.4	present study
Pamba	48	0	0	0	0	0	0	8.5	present study
Achankoil	49-60	0	0	0	0	1.0-1.5	0	8.8-10.1	present study
SI8**	300	50	50	20	10	500			BIS-10500- 2012

193

Table	5.4 Com	parison of di	ssolved tra	ce metals (μ ν	g/L) observ vith previou	ed at the l is reports	ower reach	ies of rive	rrs and CE	on 28 th -30 th Oct-15
Location	Fe	Zn	Mn	Cr	Cu	Cd	Pb	Ni	Mg	Reference
KC	0.1-306	0.26-346	pu	а	<1.0	0.5-7.0	0.5-19	а	а	Rajamani, 1995
Off Cochin	24-381	19-31	a		5.2-11.9	a	2	а	1	Balachandran, 2000
Inlet	799	4.36	0.63	4.7	0.68			4.56		Anas et al., 2015
Inlet	9.6-125	nd-6.2	nd-9.5	2.0-5.5	pu	3.3	pu	co.05	29-948	Present study
Kuttanad	9.0-390	2.0-390	0.05-120	0.08-07.2	0.3-8.5	0.4-6.0	0.6-41	0.3-18.2	1.0-520	Unnikrishnan, 2000
CE(north)	<1.0	20-30	12.0-70		1.0-3.0	<0.1-4.0		<1.0		Paul and Pillai, 1983
CE	0.2-0.8	145-216	£		10.4-18.8	2.2-3.4	8.0-14.0	0.4-0.6	Ŀ	Ouseph et al., 1992
CE	0.1-64	0.5-89	pu		0.1-6.0	0.1-0.59	6.0-24	а		Rajamani, 1994
CE	100-460	4.0-47	ĸ	в	1.0-3.0	r.	r			Balachandran, 2001
CE	50-400	20-140	10.0-90		1.0-3.0	0.15-10	20-80	5.0-10		Nair et al., 2013
CE(north)	7.0-68.3	6.36-105.9	nd-0.13	0.07-0.009	nd-0.44	Ľ	T.	nd-5.51	ĸ	Anas et al., 2015
*LR & CE	39-147	nd-67.4	nd-9.7	nd-2.3	pu	nd	nd-0.13	nd-0.5	6.0-10.2	Present study
		*LR-Lower	reaches of r	ivers dischargi	ng into Cochi	n estuary: Cl	3- Cochin E	stuary: KC-	Kerala Coas	t

Table 5.5 previous n	Comparis eports (Co	son of partix yncentration	culate trace n n of Fe and N	nctals (µg Ag was rcj	/g) obscrved presented in	i at the lower percentage)	r reaches of i	rivers and	CE on 28	ահ_30ահ ()ct-15 with
Location	Fe (%)	Zn	Mn	С	C	Cd	ЪЪ	N	Mg(%)	లి	Reference
КC	0.29-2.5	11.18-954	74.2-1103.8		23.6-168.1	nd-190	100-449				Rajamani, 1994
Inlet	0.06-7.8	nd-1307.9	nd-3483.4	nd-148.1	pu	pu	nd-1.79	nd-59.2	0.02-18.1	nd-3.7	Present study
kuttanad	0.25-43.3	54-4602.5	210-2.7*10 ⁵	6.59-925	7.62-1559	0.56-191.5	8.8-2556.8	8.9-860.4			Unnikrishnan, 2000
CE (north)	2.96	1131	500		107	95		274			Paul and Pillai., 1983
CE	7.4-8.1	150-2200	I	I	65-230	50-150	60-180	10.0-80			Ouseph et al., 1992
CE	1.15-6.4	654-1748	35.78-89.47		31.9-33.8	5.0-48	250-990				Rajamani, 1994
CE	<2.0	<200	<40		12.0-40.0	5.0-25.0	<100	10.0-30.0			Nair et al., 2012
*LR & CE	2.7-18.4	nd-1353.62	nd-289.88	nd-864.4	nd-282.7	pu	nd-126.4	nd	0.21-1.2	pu	Present study

Input and export fluxes of trace metals

CHAPTER 5
5.3.5 Input-export fluxes of trace metals

5.3.5.1 Input Fluxes

Input of Zn:

The inputs of trace metals incident through the 6 major rivers to the Cochin estuary are depicted in Fig 5.14. The input fluxes of Zn through Periyar, Muvatttupuzha, Meenachil, Pamba, Manimala and Achankoil on 28th-30th November 2017 was 19.8kgd⁻¹, 22.4kgd⁻¹, 37.2 kgd⁻¹, 36.1kgd⁻¹, 37.7kgd⁻¹ and 50.2kgd⁻¹ in which the transport through Periyar, Muvattupuzha and Meenachil was entirely in dissolved form whereas the transport through Pamba, Manimala and Achankoil was mostly bound to suspended particles. The Zn concentration was high in river Manimala, though its fluxes were lower than through Achankoil. The total input of Zn through all the six rivers was 20.3.57kg⁻¹d, which account for the natural addition of Zn through geochemical processes such as weathering, dissolution and scavenging.

The effect of effluent discharges into the Periyar through the industrial zone of Eloor was well reflected in the transport of Zn into the estuary through Moolampilly (upper estuary). The Zn flux through Moolampilly region towards downstream during the study period was 182.6kgd⁻¹, which was ~10 times higher than the riverine flux. As, Moolampilly is a region prone to salinity fluctuations at the bottom in accordance with the tidal incursion and as the flux measurement at Moolampilly was a one-time cross sectional measurement using ADCP at the ebb period, the value observed during the present observation can vary a little with respect to ebb-flood variations. The transport of Zn through Moolampilly was mainly in dissolved form (87.8%).

The discharge of Zn through upper reaches of Muvattupuzha was 22.4kg/d, which was entirely in the dissolved form. When the river bifurcates

into two branches, Murinjupuzha and Ithipuzha, the partitioning of Zn modified in the branches before entering the Cochin estuary. While, the flux of Zn through Ithipuzha was mostly particulate bound, the flux through Murinjupuzha (1.45kg/d) was entirely in dissolved form. Nearly 4.8kg of Zn associated with particles and 0.4kg of Zn in dissolved form was inflowing the Cochin estuary per day through Ithipuzha. The higher association of Zn to particles in Ithipuzha branch was due to the higher suspended particles. A small branch of Muvattupuzha is inflowing the estuary through the Kariyar and the Zn fluxes through this region were 0.02kg/d (mostly in dissolved form)

Input of Cr:

The present study identifies the source of Cr into the estuary. The input fluxes of Cr through Periyar, Muvatttupuzha and Pamba, on 28th-30th November 2017 was 139.5kgd⁻¹, 95.06kgd⁻¹ and 13.9kgd⁻¹ and was mostly bound to suspended particles (Fig 5.14). Cr was completely absent in the rivers Meenachil, Manimala and Achankoil. Accordingly, the total Cr influx into the estuary was 247.5kgd⁻¹ during the study period.

The transport of Cr through Moolappilly, the upper estuary (lower reaches of Periyar) was merely 7kgd⁻¹ (54.5 % particulate), because of its removal on to the coarser sediments. The high concentration of Cr in the residual and Fe-Mn oxide fractions in the sediments of Moolampilly with clayey sand texture also support this finding (Chapter 3). Even though greater transport of particulate Cr (95.06kg/d) was noticed at the upstream of Muvattupuzha, a very less amount (0.01kg/d) was found to flow through Kariyar. Rapid settling of Cr bound to coarser suspended particles might have resulted in the absence of Cr at Murinjupuzha and Ithipuzha region. High filtration capabilities of some molluscs also significantly increase the removal

of these elements through sedimentation (Hawkins et al, 1996). Thus presence of filter feeders reported at lower reaches of Muvattupuzha can also contribute to the lower Cr levels in this region.

Input of Fe:

Even though concentration of Fe (7.4%) was higher in Periyar compared to Muvattupuzha (4.5%), the rate of transport of Fe through Muvattupuzha (8.55td⁻¹) was higher compared to the River Periyar (7.33td⁻¹) (Fig.5.14) due to higher freshwater flow. The transport of Fe (11.15td⁻¹) through river Manimala was slightly higher than Muvattupuzha in which 98.5% transported Fe was bound to particles. The river Meenachil discharged about 13.31tons of Fe per day and the river Pamba discharged about 21.22 tons of Fe per day towards the lower reaches. The greater discharge of Fe was through river Achankoil and the transport was mainly through particulate phase. About 27.44 tons of Fe was discharging into the estuary per day through river Achankoil. The variations in the transport rate can be attributed to the suspended loads and water flux. The concentrations of Fe in rivers Manimala, Muvattupuzha and Meenachil were higher, though the Fe influx was higher through Achankoil and Pamba. The lower water flux of river Manimala and lower suspended particles in Meenachil and Muvattuppuzha account for this discrepancy. It is obvious that the major source of Fe into the estuary was riverine and the major contributor was river Achankoil due to higher suspended particles and higher water flux. The contribution of rivers towards the Fe influx follows the order Achankoil > Pamba > Meenachil > Manimala > Muvattupuzha > Periyar.

As the Moolampilly region (upper estuary) accumulated more SPM, the particulate iron concentration was much lowered. Moreover, weak flow in this region also reduced the downstream transport of Fe (2.15td⁻¹). The transport of Fe through Murinjupuzha branch was 3.0td⁻¹ (mostly particulate, 88.1%) while the flux through Ithipuzha was 1.12td⁻¹ (mostly particulate, 89.6%). The reduced transport towards lower reaches was due to its low concentration as well as lean flow. Particulate Fe was high at Kariyar, but its fluxes were also low 0.0064td⁻¹ (particulate, 93.3%) due to lowered water flow.

Input of Mg:

The input of Mg through river periyar was 757.8kgd⁻¹ in which, 90.6% was as particulate (Fig 5.14). Despite its higher concentration in Periyar, the Mg flux through Muvattupuzha was slightly high (781kgd⁻¹) where 67.8% was in particulate form. Similarly, the inputs of Mg through Meenachil, Manimala, Pamba and Achankoil rivers were also mainly contributed by particulate fraction and the fluxes were 1245.5, 588.2, 1666 and 1705.1kgd⁻¹ respectively.

Input of Mn:

The transport of Mn to lower reaches was completely in particulate phase (Fig 5.14). The flux of Mn through river Periyar was 110.09kgd⁻¹. The flux of Mn through river Muvattupuzha was relatively low (6.98kgd⁻¹). The River Pamba contributed 142.73 kg of Mn per day towards the lower reaches. However, Mn was entirely absent in Rivers Meenachil and Manimala. The flux of Mn through Achankoil River was relatively high (344.97 kgd⁻¹) compared to other 5 rivers. This could be due to the high turbidity (spm:44.2-66.7mg/L) in this river. It's clear that the major source of Mn to the estuary

was rivers and especially, the River Achankoil. The contribution of rivers towards the Mn influx follows the order Achankoil > Pamba > Periyar > Muvattupuzha.

A decrease in flux was observed towards the upper estuary. The rate of transport at Moolampilly was only 0.4kgd⁻¹. The rapid deposition of Mn bound to particles while traversing towards the estuary could be the reason for such a decrease. Similarity in the reduction of Fe, Mn and Cr towards the lower reaches shows that Cr mostly bound to Fe-Mn oxyhydroxides in the suspended particles (Chapter 3), gets settled towards downstream. Mn was entirely absent in Murinjupuzha, at the same time a considerable influx was observed at the Ithipuzha towards the lower estuary. Higher suspended particles and low flow in Ithipuzha may favor the adsorption of Mn to particles than Murinjupuzha. About 2.1kg of Mn was entering the lower estuary per day through Ithipuzha and about 0.0024kg of Mn was entering through the Kariyar.

Input of Ni:

Major portion of Ni entering the estuary was through river Periyar (0.396kgd⁻¹) in dissolved form (Fig 5.14). From Moolampilly region, the flux of dissolved Ni (0.5kgd⁻¹) increased towards downstream due to the presence of industrial activities. There was also a very low input of dissolved Ni through Kariyar (0.0008kgd⁻¹), probably from the nearby small scale industries. In general, the Nickel input to the Cochin estuary was in dissolved form and mainly through anthopogenic sources.





Fig 5.14: Input flux of dissolved and particulate trace metals (kgd⁻¹) through rivers into CE on 28th -30th Oct, 2015. (a) Zn; (b) Cr; (c) Mn; (d) Ni; (e) Mg; (f) Fe. [Fe & Mg fluxes are represented in td⁻¹].

Input of Pb:

Lead was not present in any of the six rivers during the present study. About 0.1kg of Pb (dissolved) was entering the estuary per day through Ithipuzha due to anthropogenic input (pigments in paints, petroleum products, land drainage and sewage discharge).

Input of Cu:

Cu was non-detectable in the upstream of any of the rivers. However, the Cu flux of nearly 0.7 kgd⁻¹ through Moolampilly towards downstream accounts for the anthropogenic input into the region.

Discussions

The combined freshwater discharge from six major rivers into the estuary on 28th-30th October 2015 was 58.559Mm³d⁻¹. The major discharge was through Muvattupuzha and Achankoil and least discharge was through River Manimala (Fig 5.2). Fe, Mn, Mg and Cr inputs to the Cochin estuary were primarily from lithogenic sources whereas the inputs of other trace metals such as Zn, Cu, Ni and Pb has anthropogenic sources too, which may lead to trace metal pollution in the estuary and adjacent coastal region. The results point out that suspended particulates are the major carriers of trace metals in the estuary, which is in agreement with previous observations (Shibu et al., 1992). However, input of some of the trace metals like Zn and Ni through the rivers Periyar and Muvattupuzha was mainly in dissolved form. The poor adsorption sites for Zn and Ni or their greater stability in dissolved state might be the reason for such behavior. In the present study, it was observed that the partitioning of most of the trace metals between the dissolved and particulate phases was controlled by river flow, turbidity, salinity and the available sites on suspended particles. Recent studies have shown that fluxes of materials through riverine suspended particles are 20

times greater than the combined flux of dissolved materials and atmospheric dust (Gaillardet et al., 2003; Jickells et al., 2005; Walling, 2006).

From the results it is clear that trace metals such as Ni, Cu, Co, Mg and Pb are increasingly concentrated in suspended solids as they traverse from the upstream towards high saline zones. However, there are some other metals (Fe, Mn, Cr) that showed a reversing trend, when there was a rapid deposition (scavenging by Fe and Mn hydroxides and SPM) to estuarine sediments. Thus suspended particles play a vital role by acting simultaneously as a source and sink of trace metals. The results illustrate that the difference in the river geochemistry is principally controlled by its tributaries (Meyer et al., 1988), land use pattern (Townsend et al., 1983; Bucker et al., 2010) and soil type (Harmon et al., 2009; Leite et al., 2010), whereas their downstream transport was regulated by the suspended particles and estuarine processes

5.3.5.2 Residual fluxes at the Cochin inlet

The net riverine water flux per day (input) on 28th-30th October 2015, the export/ residual flux at inlet on 3rd and 4th November 2015 and the computed net water flux per tidal cycle through Cochin estuary on 3rd and 4th November 2015 are presented in Table 5.6.

Table 5.6 River input (28th-30th Oct-15), export at Cochin inlet (3rd-4th Nov-15) and net water flux (Mm³) through Cochin estuary during Oct-Nov 2015.

Water Flux						
Input $(\mathbf{Mm}^{3}\mathbf{d}^{-1})$	Export (Mm ³ /tidal cycle)	Net flux through CE (Mm ³ /tidal cycle)	Net flux relative to input flux			
58.559	67.095	6.096	9.993%			

Dynamics of trace metals, rare earth elements and nutrients in the CE



Fig 5.15: Water flux (m³s⁻¹) for each 3 hour intervals through surface (a), mid (b) and bottom (c) water at Cochin inlet on 3rd-4th November 2015

The freshwater discharge from six major rivers into the estuary was measured on 28^{th} - 30^{th} October 2015 (58.559Mm³d⁻¹). As the flushing time of ~ (2.5) days was reported in Cochin estuary (Vinita et al., 2015a) for wet season, the time series measurements were conducted at Cochin inlet on 3^{rd} and 4^{th} November 2015 between a spring and neap tide to compute the export

fluxes. The flux measurements recorded using ADCP were then partitioned into three layers using MATLAB. Variations of water flux (surface, mid and bottom) with respect to tide are depicted in Fig 5.15.

The net water flux at the Cochin inlet through surface, mid and nearbottom during 3 - 4th November 2015 was 45.76, 29.44 and -8.18Mm³/tidal cycle respectively, showing the vertical changes in the bidirectional flow (Fig 5.16). Surface and mid depth water flux was directed seaward (+ve) and the bottom flux was directed landward (-ve). The residual flow during this period (sum of surface, mid and bottom flux) was 67.095Mm³/tidal cycle indicating a net export (Table 5.6).



Fig 5.16: Residual water flux (Mm³tidal cycle⁻¹) through surface, mid and bottom water at Cochin inlet on 3rd-4th November 2015

During the observation, the advection of seawater through the bottom and its influence on the surface salinity was noticed only during the highest high water (HHW) at 27hr. For all other time period irrespective of ebb/flood tide, the surface and mid depth water flow was directed seaward, resulting in a CHAPTER 5

net seaward flow. Intertidal and seasonal changes will also influence the water exchange at the inlet, as the major controlling factors are the tides and freshwater discharge. It is to be noted that being a monsoonal estuary, the inlet exhibited a net export during most of the time. The Cochin estuary has a very good flushing (42 times/yr.) capability (Revichandran et al., 2011). The fluxes of trace metals (Zn, Cr, Fe, Mg, Ni, Mn, Pb, Cu, Co and Cd) through the Cochin inlet on 3rd-4th November 2015 are depicted in Fig 5.17-5.23. It can be seen that during this period, the major part of the trace metal fluxes across Cochin inlet was through particulate form and directed seaward.

The flux of dissolved and particulate Zn through surface, mid and bottom water at the Cochin inlet were (0.166 & 1.31)t, (0.036 & 0.007)t and (-0.052 & 0.085)t/tidal cycle respectively, indicating a net export of (0.150 &1.397)t/tidal cycle. There was import of dissolved Zn (-0.052t/tidal cycle) through the bottom, which could be due to the high dissolved Zn concentration during flooding (15-18 hr and 27-30hr). The dissolution of Zn ion is found to increase at high salinity (Chester, 2009). The flux of dissolved and particulate Cr through surface, mid and bottom water at the Cochin inlet were (0.172 & 1.46)t, (0.172& 0.0)t and (0.172 & 0.176)t/tidal cycle respectively, also indicating a net export of (0.515 & 1.636)t/tidal cycle. The adsorption of Cr increased with increasing turbidity, i.e with an increase in the concentration of suspended matter leads to increased particulate export.

The fluxes of Fe (dissolved and particulate) through surface, mid and bottom through the Cochin inlet were (2.05 & 297.77)t, (1.093 & 2.659)t and (0.127 & 31.694)t/tidal cycle respectively. A net export of (3.271 & 332.12) t/tidal cycle at the inlet is thus, attributed to the increased concentration of Fe and increased fresh water flow. Similarly, Ni fluxes (dissolved and particulate)

at Cochin inlet through surface, mid and bottom were (0.01&0.582)t, (-0.001 & 0.004)t and (-0.004 & 0.0)t/tidal cycle respectively. The Ni flux was also mainly associated with particulate material. Dissolved Ni was high in the bottom, probably due to it's desorption from particles at higher salinity. Higher Ni at bottom during flood period leads to higher dissolved flux directed landward. The net export of Ni at the inlet was (0.005 & 0.586) t/tidal cycle. Pb was observed only in particulate phase at the bottom during ebb (12hr) and indicated a net export (0.003t/ tidal cycle). Particulate Co also showed a net export of (0.029t/tidal cycle). Dissolved Cd showed a net export (0.0025t/tidal cycle), probably through remobilization from sediments during ebb tide (12 hr).

The Mn fluxes (dissolved and particulate) through surface, mid and bottom water at the Cochin inlet were (0.0 & 1.796)t, (0.102 & -0.233)t and (0.0& 0.184)t/tidal cycle respectively. Dissolved Mn was not present during the observation, except at 33hr (ebb tide) in the mid depth ($9.5\mu g/L$), which caused an export flux of 0.102t/tidal cycle. In aquatic systems, manganese solubility increases at low pH as well as under low oxidation-reduction potential (Heal, 2001). As the pH was high (7.83) at inlet at 33hr and the entire water column was well oxygenated (5.01-6.34mg/L), the presence of Mn at mid depth ($9.5\mu g/L$) could be due to some other reason. Manganese is an essential element in phytoplankton and therefore, can be a source during its decomposition (Nadaska et al., 2012). High saline waters containing high concentrations of nitrate, chlorides and sulphate can also increase the solubility of Mn and its uptake by plankton (Reimer, 1999).



Fig 5.17: Zn flux (gs⁻¹) for each 3 hour intervals and the residual flux (t/tidal cycle) through surface, mid and bottom water at Cochin inlet on 3rd-4th November 2015.



Fig 5.18: Cr flux (gs⁻¹) for each 3 hour intervals and the residual flux (t/tidal cycle) through surface, mid and bottom water at Cochin inlet on 3rd-4th November 2015

209



Fig 5.19: Fe flux (kgs⁻¹) for each 3 hour intervals and the residual flux (t/tidal cycle) through surface, mid and bottom water at Cochin inlet on 3rd-4th November 2015



Fig 5.20: Mg flux (kgs⁻¹) for each 3 hour intervals and the residual flux (t/tidal cycle) through surface, mid and bottom water at Cochin inlet on 3rd-4th November 2015.

211





Fig 5.21: Mn flux (gs⁻¹) for each 3 hour intervals and the residual flux (t/tidal cycle) through surface, mid and bottom water at Cochin inlet on 3rd-4th November 2015.



Fig 5.22: Ni flux (gs⁻¹) for each 3 hour intervals and the residual flux (t/tidal cycle) through surface, mid and bottom water at Cochin inlet on 3rd-4th November 2015.

213



Fig 5.23: Pb and Co flux (gs⁻¹) for each 3 hour intervals and the residual flux (t/tidal cycle) through surface, mid and bottom water at Cochin inlet on 3rd-4th November 2015.

Dynamics of trace metals, rare earth elements and nutrients in the CE

There was a reversal in the water flux at surface and mid depth, during flood at 27hr (-ve water flux) associated with increased turbidity (SPM). Elevated concentration of particulate Mn in the flood waters has probably resulted in an import flux at mid depth. However, the residual flux of particulate Mn showed an export of 0.102 & 1.747t/tidal cycle.

5.3.5.3 Net fluxes of trace metals

The net fluxes of trace metals through Cochin estuary is the difference in net riverine input and export at Cochin inlet.

Net flux = (net input fluxes through 6 major rivers) – (the export/residual fluxes through Cochin inlet).

The Cochin estuary showed a net export of 6.096Mm³ water (per tidal cycle) during the observation period, which tuned to $\sim 9.99\%$ of fresh water input (Table 5.6). This difference in volume exchange (input-export) was well reflected in the trace metal transport across the Cochin inlet. The input and export fluxes of trace metals during the observational period are provided in Table 5.7. The net fluxes were computed based on the difference between the sum of inputs from 6 rivers and the transport through Cochin inlet. The result showed an average export of 246.39t Fe, 232.7t Mg, 1.90t Cr, 1.34t Zn, 1.25t Mn, 0.59t Ni, 0.029t Pb and 0.003t Pb respectively from the estuary during a tidal cycle on 3rd and 4th Nov-15 (Table 5.7), of which, ~ 68-99% was transported as particulate fraction. The dissolved metal fluxes were in the order Mg > Fe > Cr > Mn > Zn and particulate metal fluxes followed an order Fe > Mg > Cr > Zn > Mn > Ni > Co > Pb. The net transport of dissolved and particulate Cr was 45.9 and 5.67 times higher than their river input. Same was the case for Mg, in which, the net fluxes were 116.6 and 34.1 times higher than their riverine input. The net flux of Ni was 10.22 times in excess than its

CHAPTER 5

riverine input. Since the particulate Ni was absent in rivers whatever exported was excess contribution. The export fluxes of Fe (dissolved/particulate) were less (0.2 and 2.73 times of river discharge) compared to other metals. Considering the role of iron in oceanic productivity, this element plays an important role in the carbon cycle (Moffett et al., 2015). The net flux of particulate Zn was 11.02 times higher than its riverine input while the net flux of dissolved Zn was only 61.4% of the river discharge. The excess transported metals are the consequence of their anthropogenic sources into the estuary.

Table 5.7 Input (28th-30th Oct-15), export /residual at inlet (3rd-4th Nov-15) and net heavy metal flux (tons) through Cochin estuary during Oct-Nov 2015

Heavy metals	Net Input (td ⁻¹)	Export/ residual flux at the Cochin inlet (t/tidal cycle)	Net Flux through CE (t/tidal cycle)	Net flux D+ P (t/tidal cycle)
D Zn	0.09	0.15	-0.06	-1.34
P Zn	0.11	1.40	-1.29	
D Cr	0.01	0.51	-0.50	-1.90
P Cr	0.24	1.64	-1.40	
D Fe	2.70	3.27	-0.57	-246.39
P Fe	86.30	332.12	-245.82	
D Mn	0.00	0.10	-0.10	-1.25
P Mn	0.60	1.75	-1.14	
D NI	0.0004	0.00	0.00	-0.59
P Ni	0.00	0.59	-0.59	
D Mg	0.61	74.32	-73.71	-232.70
P Mg	4.48	163.47	-158.99	
P Pb	0.00	0.00	0.003	-0.003
P Co	0.00	0.03	-0.03	-0.03

'D' & 'P' represent dissolved and particulate fractions of metals: -ve sign represents seaward flux.

Suspended particles are the major carriers of trace metals and some of the binding sites on them (Fe and Mn oxides) are good scavengers capable of binding other elements to accumulate in sediments. Thus, Cochin estuary is a prominent depository of trace metals and one of the most impacted estuaries in the world (Balachandran et al., 2005). However, since SPM are finer particles, it may increase the export fluxes of trace metals especially during heavy run off periods (Shynu, et al., 2012). The present study supports this finding, as there was a net export of most of the heavy metals through the Cochin inlet.

Now, what could be the fate of these input/export fluxes is a question of great concern. The ultimate sink of these trace metals could be their deposition in coastal sediments (Pichaimani et al., 2008). The changes in redox conditions, pH, dissolved oxygen depletion etc. could be very crucial in the depositional environment where they are removed from the dissolved and particulate form to the sediments and vice versa (Machado et al., 2016). Thus, sediments can acts as both sink and source of metals under changing redox conditions. In the estuary, the deposition will be high, but there is a chance of their release back to the water as bioavailable and toxic form. The accumulation of trace metals in sediments is a long-term source of contamination to higher trophic level, as they are exposed to detrital and deposit-feeding organisms (Mendil and Uluozlu, 2007). The increased trace metal levels in suspended particles have adverse effect on the filter feeders.

In the coastal environment, oxygen deficient conditions may develop occasionally, when the bioavailable fractions of trace metals increase. This will definitely affect the health of marine organisms as well. There are several reports to show the impact of the deterioration in the Cochin Estuary and its adjoining coast (Jayakumar et al., 2001). Recent reports showed exceptionally high carbon uptake rates in the near shore and inner shelf waters of Cochin which is attributed to the export fluxes from the estuary (Bhavya et al., 2017). These also evidence the anthropogenic inputs in the region modulating the coastal biogeochemistry. However, sediment trace metals are found to undergo diagenesis and rapid chemical alterations (Nath et al, 2000). Increased supplies of trace metals are deposited in coastal sediments during monsoon, while they are effectively masked by incorporation onto clay or organic matter during the remaining period (Balchandran et al., 2003). Thus, these trace metals do not seem to constitute a threat to the coastal areas other than in estuaries. However, recent studies have shown that the coastal sediments of Cochin have increased trace metal levels and seem to be under stress (Sudhananth et al., 2011; Sreekanth et al., 2015).

The co-precipitation and adsorption of metals by Fe-Mn oxides is a major pathway of metal deposition (Fairbrother et al., 2007; Lu et al., 2014). Under reducing conditions, the hydroxides of Mn and Fe are destabilized and start releasing all other adsorbed and co-precipitated metals like Co, Ni, Cr, Cd, Cu, Pb and Zn (Zwolsman et al., 1993). This is accelerated by sulphate-metal interactions that increase their concentrations in pore or surface water (Du Laing et al., 2009). At even stronger reducing conditions, sulphate is reduced to sulphides, which form complexes with other metals and reduces their mobility and toxicity (Lu et al., 2014). In high saline conditions, the ionic interactions with carbonates, oxides, oxide-hydroxides, silicates and chloride can increase the metal adsorption on to sediments and vice versa (Fairbrother et al., 2007), whereas tidal activity can increase the sediment re-suspension

(Hawkins et al., 1996) to desorb metals from particles and pore waters (Lindberg et al., 1975).

The presence of deposit-feeders (polychaetes) also influences the deposition and remobilization of metals (Machado et al., 2016). Certain bacteria are found to solubilize heavy metals in sediment (Pillidge, 1985) and concentrate as extracellular polysaccharides (Aislabie and Loutit, 1986) facilitating its transfer to food chain (Bremer and Loutit, 1986). Therefore, a detailed study regarding the mobilization of metals is necessary to evaluate their ecotoxicological risk.

5.4 SUMMARY AND CONCLUSION

The present study shows that all the 6 major rivers contribute very low concentrations of trace metals, which are within the limits of Bureau of Indian Standards (BIS). Therefore, at present, there is no toxicity of trace metals (Fe, Zn, Mn, Cr, Ni, Co, Cu and Cd) from these rivers. Weathering and resuspension are the major source of trace metals and Fe & Mg in the rivers. The present study showed that there were wide variations in the geochemical properties of all these rivers due the difference in the nature of alluvium in the catchment area.

The concentration of dissolved metals in the lower reaches of rivers (upper estuary) were below the permissible value of drinking water recommended by BIS-10500 (2012) and WHO, whereas particulate matter showed enriched heavy metals levels. Particulate Zn at Moolampilly and Ithipuzha were greater than the values reported in PAAS ($85\mu g/g$) and WRSPM ($208\mu g/g$). The concentration of Fe (dissolved and particulate) in

CHAPTER 5

Kariyar and Cu in Purappillikkavu (particulate) were also high during the study period. Particulate Cr concentrations at Moolampilly, Purappillikkavu, Kariyar and Kumarakom were also higher than that in WRSPM ($130\mu g/g$) and PAAS ($110\mu g/g$). Particulate Pb at Purappillikkavu and Kumarakom was higher than that in PAAS ($20\mu g/g$) and WRSPM ($61.1\mu g/g$), while particulate Mn was lower than that of WRSPM (0.16%) and PAAS (0.08%).

The concentrations of trace metals like Zn, Cu, Cr and Pb in suspended matter showed an increase from the rivers to the estuary indicating their addition towards downstream. For other metals (Cr, Fe, Mn), a decreasing trend was noticed for Periyar and Muvattupuzha rivers, probably due to their deposition in the upper estuary. Thus, suspended particles play a vital role as a source and sink for trace metals. Bio-accumulation could also be a reason for the absence of Cr in both dissolved and particulate phases at Ithipuzha and Murinjupuzha. The present study showed that Murinjupuzha and Pallathuruthy were the least polluted among the regions studied, where heavy metals were generally absent.

The time series measurement at Cochin inlet on 3^{rd} -4th November 2015 showed that the concentrations of dissolved Co, Cu and Pb were below measurable levels at the Cochin inlet, while dissolved Mg, Fe, Cr, Zn and Ni showed appreciable concentrations. Dissolved Mn and Cd were measured up to 10 and $3.3\mu g/L$ respectively, though it was non-detectable for most of the time. The metals Fe, Mg, Zn and Mn were mostly present in particle form, whereas dissolved Ni and Cr showed enrichment at higher salinity (bottom). The increase of heavy metals (Cr, Ni and Co) in turbid waters suggests that these metals are concentrated by suspended particulate matter. Some of the particulate heavy metals (Fe, Zn and Mg) showed strong inter relationship, while others did not show any correlation indicating the variable interaction among metals and with particles.

Increased river discharge naturally increases the suspended load, while tidal activity increases the re-suspension, both of which were evident during the present study. The enrichment of Fe in the upstream and its depletion towards the inlet could be due to the flocculation of Fe bound particles at high salinity. The non-conservative behavior of Fe could be due to its mobilization.

The observation that suspended particulates are the major carriers of trace metals to the Cochin estuary is in accordance with earlier studies. However, some metals like Zn and Ni behaved differently, as they were transported through Periyar and Muvattupuzha rivers. The poor adsorption sites of Zn and Ni or their better stability in dissolved state might be the reason for this. The present study also showed that partitioning of trace metals between the dissolved and particulate phases is controlled by river flow, nature of suspended particles, pH and salinity. The particulate concentrations of Cu and Fe during the present study were higher than the previous reports. However, particulate Mn was relatively lower than reported by Paul and Pillai, 1983, though it was higher than other reports. Dissolved fractions of all the metals were comparable with the earlier studies.

The average freshwater flow through the six rivers during 28-30th October 2015 was 58.559Mm³d⁻¹, resulting in the net river input of 203.57, 604, 247.5, 0.4kgd⁻¹of Zn, Mn, Cr, Ni and, 89.0, 5.09td⁻¹of Fe and Mg respectively. The major contributor of Zn, Fe, Mn and Mg to the estuary was the River Achankoil following a high turbid condition. The major contibutor

CHAPTER 5

of Cr and Ni to the estuary was River Periyar. The results shows that the difference in the river geochemistry is principally controlled by its tributaries, land use pattern and soil type, whereas their downstream transport was regulated by the suspended particles and estuarine processes.

The exchange of water through the Cochin inlet (surface, mid and bottom) during 3- 4th November 2015 was 45.76, 29.44and -8.18Mm³/tidal cycle respectively. The residual flux of 67.095Mm3/tidal cycle at the inlet was indicative of a net export from the estuary. The net water flux through the estuary (Input-Export) was 6.095Mm³/tidal cycle directed seaward. It is estimated that about 9.99% of freshwater (6.095Mm³) is getting exported in excess through Cochin inlet in a tidal cycle on 3- 4th Nov -15. This was also reflected in the trace metal transport across the inlet (246.39t Fe, 232.7t Mg, 1.90tCr, 1.34t Zn, 1.25tMn, 0.59t Ni, 0.029 t Co and 0.0031t Pb) per tidal cycle. Approximately 68 to 99% of this flux was associated with particles. The export fluxes can be attributed to the anthropogenic activities and internal processes within in the estuary

From the present study, it can be concluded that suspended particulate matter plays a major role in the geochemical cycling of elements in the Cochin estuary. The processes that affect the SPM dynamics leave significant impressions on the heavy metal distribution in the estuary. Moreover, the interactions of phytoplankton and filter feeders with particulate and dissolved materials also control the distribution of trace metals in the Cochin estuary. Scavenging by suspended particles has resulted in the removal of these trace metals on to estuarine sediments. Otherwise, the export fluxes would have been much larger than that observed today. The fate of these export fluxes is a matter of great concern as they are ultimately deposited along the coastal environment. The sediment texture, hydrological regime, salinity and organic matter were responsible for the mobility and release of these metals. Even though the fractionation study (chapter 3) indicates the fate of metals in the coastal sediments, which can be used for natural water management system; there are still uncertainties in the regulating mechanisms for metal mobility in coastal sediments. Therefore future studies should include the physical, chemical and biological mediated metal remobilization and microbial mediated mineralization in the estuarine and coastal marine environment.

****\$\$****

Chapter **6**

SUMMARY AND CONCLUSION

Estuarine and coastal marine areas are highly complex and dynamic environments. Different kinds of physical, chemical, geological and biological interactions in the estuary during the mixing of freshwater and seawater can significantly regulate the transport and fate of materials (nutrients, trace metals and rare earth elements) from the estuary. Nutrients are the raw materials for the food chain, as they are essential to the primary production, while heavy elements are important due to their toxic effects, as well as their importance as micronutrients for many organisms. Rare earth elements are important tracers to understand the evolution of the earth's continental crust. They could also provide information regarding the depositional conditions of an environment. Sediments can represent sources and sinks of these materials in estuaries and the factors determining these pathways will largely be determined by quantifying the input, export and depositional processes in sediments.

The present study was mainly focused on the Cochin estuarine system, the largest tropical estuarine system along the south west of India which has significant influence on the biogeochemistry of the adjacent Arabian Sea. Although there are several studies delineating the environmental changes in the Cochin estuary caused by anthropogenic activities, there are only a few studies to quantify the material transport across the estuary and its fate on the adjacent coastal marine area. The major objective of the study was to understand the inter-tidal, intra-tidal, spatial and seasonal variations in the nutrient transports in the estuary and compute their respective fluxes to the adjacent coastal areas. To the best of our knowledge, this is the first attempt to quantify the fluxes of nutrients across the Cochin estuary. The magnitude of removal and addition for nutrients in the estuary are seasonally variable. Therefore, a series of time series measurements were carried out at different locations in the Cochin estuary for 3 different seasons to compute the input, export and net fluxes of nutrients.

Environmental impact of trace metal pollution in sediments depends not only on the total concentration of the metals, but the dominant state in which, each metal is associated in the sediment crystal or lattice. Therefore, fractionation studies were carried out to identify their mobility and availability in the sediments, which influences their deposition and availability to planktons and other organisms. Rare earth elements distribution and their chemical fractionation were also studied to access the geochemical variability and depositional condition affecting their fate and transport in the estuarine and coastal environment. The temporal and spatial variations of heavy metals in the rivers and estuary help to compute their fate, transport and transformations during estuarine mixing. These studies could provide a background of the processes that control the heavy metal cycling in the water, suspended particles and sediments of Cochin estuary and thereby its fluxes from rivers through estuary to the coastal region.

It is observed that rivers are the major source of the nutrients, nitrate and silicate in the estuary, while phosphate and ammonia are also added through non-point sources and re-suspension during estuarine mixing. Another observation is that the Cochin estuary maintains an oxidized water environment during all the three seasons. However, under-saturation in the dissolved oxygen levels recorded at the Cochin inlet during summer monsoon and post monsoon period was associated with coastal upwelling and incursion of hypoxic waters. These hypoxic waters were found to reach well inside the estuary, a finding reported for the first time. Moreover, there were significant inter-tidal, intra-tidal and seasonal variations in all the nutrients and dissolved oxygen during the study period. These variations are typical of an estuary when nutrient-rich fresh water mixes with nutrient deplete seawater on a varying degree depending upon the high/low (food/ebb flow), spring/neap (high amplitude/low amplitude) and seasons (variable fresh water flow).

The present study showed that despite a moderate utilization and sinking of nutrients, dissolved nutrients are exported from the Cochin estuary, especially during peak summer monsoon. Cochin estuary is a source of all nutrients during monsoon and a sink for SiO₄ during post-monsoon (spring and neap phase) and pre-monsoon (spring phase) period. The removal (15-20%) of SiO₄ during post-monsoon was coincided by a net addition of N and PO₄. During pre-monsoon spring 88.9% of NO₃ and 27.6% of SiO₄ were removed from the estuary due to biogeochemical processes. It has been estimated that approximately 148.6tons of NH₄ was retained in the estuary as an additional input from coastal region during pre-monsoon.

Thus, the Cochin estuary was a small source of nutrients to the adjacent coastal region during the study period. The present estimates of nutrient fluxes from Cochin estuary were higher than some estuaries (Chaliyar and Tapi estuary), but comparable with that of Hoogly and Meghna estuaries. The significance of this finding is that the export fluxes from the Cochin estuary could be a major factor sustaining the spectacular monsoon fishery along the south west coast of India.

The present study identified increased anthropogenic input of trace metals into the estuary and its impact on the coastal region. It was observed that the binding phases played a major role in the mobility of trace metals in the sediments. The exchangeable and bicarbonate phases were the important non-residual fractions for some metals (Zn, Cd and Mn). The enrichment of labile fractions of Cd and Mn with increase in their total metal content indicates that this would exacerbate the potential toxicity of these metals in the estuarine and coastal environment. The organic phase was the preferred phase for Pb, Cr and Ni in the estuarine sediments and Cr in coastal sediments. Inorganic precipitation though Fe-Mn oxyhydroxide was assumed to be the important depositional pathway for Zn, Ni, Pb, Cd, Mn and V in both estuarine and coastal sediments. The residual phase was the stable phase for all the metals (except Cd at some stations) in the estuarine and coastal sediments. Mn and Zn in residual fraction were higher than the value reported for adjacent source rock and was even higher than the standard sediment quality guidelines. The enrichment of these metals in the residual fraction could be due to the biogenic/diagenic transformation of non residual metals to stable residual forms under favourable conditions and thus making the pollutants less available to biota

A comparison with other fractionation studies in Cochin estuary showed that, an increase in the exchangeable Cd and Zn in the estuarine sediments prevails over the years. The concentrations of Mn, Cr, Pb, Ni and Cu in exchangeable/carbonate fractions have decreased over the years. The fractionation study explains the fate of metals under different environmental conditions, which can be used for natural water management system. The present study is a first time attempt to carry out the fractionation of rare earths, thorium and uranium in the sediments of Cochin estuary and adjacent coastal region to understand the fate, transport and depositional characteristics of these elements. Rare earth element (REE) geochemistry of sediments from Cochin estuary and adjacent coastal stations showed that the light rare earth elements (LREE) are the more abundant compared to middle rare earth elements (MREE) and heavy rare earth elements (HREE). Cerium was markedly high in sediments of estuarine and coastal stations as it was the most abundant rare earth in earth crust. Lutetium was the least abundant rare earth present in the estuarine and coastal sediments.

Enrichment of rare earth elements (REE) in the north estuary (S_1 , S_2) and a gradual decrease towards the central estuary (S_3 , S_4 , S_7 , S_8) and to a lowest level near the Cochin inlet ($S_5 \& S_6$) are some of the notable observations. The dynamic nature of the inlet may support swiping of finer particles leading to impoverishment of elements in the sediments. The coastal sediments showed high concentration of rare earths and the values were similar to the central estuary. This generally indicates the mobility of elements from riverine sources to downstream.

Shale normalized values of rare earths showed felsic source. A deviation in shale normalized pattern observed at northern estuary can be due to additional supply of REE into the estuary in addition to detritus. High thorium content in theses station too supports the external source. Positive Ce anomaly observed in the study region supports the terrigeneous origin. A slight negative Ce anomaly at northern station could be due to biological productivity. No anomaly or slightly positive Eu anomaly in both estuarine and coastal sediments supports its feldspar origin. Redox constraints like U/Th

ratio and authigenic U supports the oxygenated conditions in both estuarine and coastal sediments.

The fractionation studies were carried out to understand the mechanism behind the deposition of REE, and actinides (Thorium and Uranium) in the Cochin estuary and adjacent coastal region. It was observed the residual phase was preferred by LREE, while organic phase was the major site for HREE in clayey sediments with high organic matter and residual phase was the major site for HREE in sandy sediments. The high thermodynamic stability of LREE in residual phase could be a reason for its enrichment in sediments. The stability of HREE in sediments depends upon the nature of organic compound to which it is associated. Sc showed deposition similar to LREEs and Y showed deposition similar to HREE due to their identical chemical properties. Thorium in the sediment was stabilized in the residual phases. The organic matter generally stabilized the Uranium in estuarine sediments, coastal sediments (with high organic matter) preferentially contained Uranium in residual phase. The bio-mineralization seems to be responsible for its accumulation in residual phase in the coastal environment. Thorium was mainly associated with residual phase in both estuarine and coastal sediments. Fe/Mn oxyhydroxides and exchangeable/carbonate phases of rare earths was relatively low compared to that in residual and organic phases. The present study thus highlights the major role of particle size and organic matter in controlling the deposition of REE in the estuarine and coastal environments.

The present study showed that suspended particles, which act both as a source as well as sink of trace metals, play a vital role in the distribution of trace metals in the estuary. The average river flow through the six rivers during 28th-30th October 2015 was 58.6Mm³d⁻¹, resulting in net river input of 203.57kg Zn, 604kg Mn, 247.5kg Cr, 0.4kg Ni, 89t Fe and 5.09t Mg per day. The major contributor of Zn, Mn, Fe and Mg was river Achankoil following a turbid condition and the major contributor of Ni and Cr to the estuary was river Periyar. The concentration of Zn, Cu, Cr and Pb in suspended particles showed an increase from rivers to the estuary indicating their addition towards downstream. For other metals (Cr, Fe and Mn), a decreasing trend was noticed for Periyar and Muvattupuzha rivers, probably due to their deposition in the upper estuary. Bioaccumulation could also be a factor for the absence of Cr in the lower reaches of Muvattupuzha. At the Cochin inlet, the metals Fe, Mg, Zn and Mn were mostly present in particle form, whereas Ni and Cr showed dissolution at higher salinity.

Some of the metals (Fe, Zn, Mn) showed strong interrelationship, while some others did not show any correlation indicating the variable interaction among metals with particles. The enrichment of Fe in the upstream and its depletion towards the inlet could be due to the flocculation of Fe bound particles at higher salinity.

The observed estuarine flow through the Cochin inlet during 3rd- 4th November 2015 was 67.095Mm³/tidal cycle (surface: 45.76Mm³/tidal cycle; mid: 29.44Mm³/tidal cycle; bottom: -8.18Mm³/tidal cycle). Consequently the net water flux through the estuary (input-export) was 6.095Mm³/tidal cycle directed seaward. This means that about 9.99% of the freshwater entering the estuary (through 6 major rivers) was getting exported in excess through the Cochin inlet in a tidal cycle on 3rd- 4th November-15. This difference in volume exchange was well reflected in the trace metal transport through Cochin estuary. The net metal flux (input-export) through the Cochin estuary

on 3rd-4th November-15 was 246.39t Fe, 232.7tMg, 1.90t Cr, 1.34t Zn, 1.25t Mn, 0.59t Ni, 0.029t Co and 0.003t Pb in a tidal cycle and directed seaward. Approximately 68 to 99% of this flux was associated with particles. Similar to the results obtained from the nutrient flux studies, there was also a net export for most of the trace metals through the Cochin inlet as a consequence of anthropogenic sources, which ultimately could get deposited in coastal sediments. The sedimentation due to flow restrictions and flocculation processes in the estuary reduces the export of trace metals into the coastal sediments. Otherwise the export fluxes for all the trace metals would have been even larger than that observed for today.

The present study shows that the high run off through the rivers are capable of maintaining a better flushing of the estuary and regulate a balance in the nutrient and heavy metal levels to safe guard the estuary from intense eutrophication and pollution. The Cochin estuary seems to be a source of nutrients to the adjacent coastal areas even during dry season. The eutrophication in coastal waters to a certain extent increase the fishery production however as a consequence of monsoon upwelling, when the hypoxia intensify over the western India shelf, the pelagic and benthic fishes tend to the surface layers to avoid oxygen deficient waters. Thus controlling nutrient input from estuary is of important for achieving a healthy coastal environment.

Moreover, the fate of the export fluxes of trace metals is a matter of great concern as the ultimate sink could be their deposition in coastal sediments. The sediment texture, hydrological regime, salinity and organic matter were responsible for the mobility and release of these metals. Even though the fractionation study indicates the fate of metals in the coastal sediments, which can be used for natural water management system; there are
still uncertainties in the regulating mechanisms for metal mobility in coastal sediments. Therefore future studies should include the physical, chemical and biological mediated metal remobilization and microbial mediated mineralization in the estuarine and coastal marine environment for proper estuarine and coastal management.

****\$\$****

- Abhilash, K. R., Raveendran, T. V., Limnamol, V. P., Deepak, M. P., 2012. Sediment oxygen demand in Cochin Backwater, a tropical estuarine system in the south west coast of India. Mar. Environ. Res. 79, 160-166.
- Aislabie, J., Loutit, M.W., 1986. Accumulation of Cr (III) by bacteria isolated from polluted sediment. Mar. Environ. Res. 20, 221-232.
- Akhil, P. S., Sujatha, C. H., 2014. Spatial budgetary evaluation of organo-chlorine contaminants in the sediments of Cochin Estuary, India. Mar. Pollut. Bull. 78, 246-251.
- Allen, P., Condie, K. C., Narayana, B. L., 1985. The geochemistry of prograde and retrograde charnockite gneiss reactions in southern India. Geochim. Cosmochim. Acta. 49, 323-336.
- Alongi, D. M., 1998. Coastal Ecosystem Processes. CRC Press, New York.
- Anas, A., Jasmin, C., Sheeba, V. A., Gireeshkumar, T. R., Nair, S., 2015. Heavy Metals Pollution Influence the Community Structure of Cyanobacteria in Nutrient Rich Tropical Estuary, Oceanography. 3, 137.
- Anirudhan, T. S., 1988. Study on the nutrient chemistry of a tropical estuary. Ph.D. Thesis. Cochin University of Science and Technology, Cochin, India.
- Anju, A. K., Dipu, S., Sobha, V., 2011. Seasonal variation of heavy metals in Cochin estuary and adjoining Periyar and Muvattupuzha Rivers, Kerala, India. Glob. J. Environ. Res. 5, 15-20.
- APHA, 2000. In: Eaton, A. D., Clesceri, L. C., Greenberg, A. E. (Eds.), Standard Methods for the Examination of Water and Wastewater (21st ed.). Washington, DC: American Public Health Association.
- Arya, P. M., Jyothibabu, R., Jagadeesan, L., Lallu, K. R., Karnan, C., 2016. Summer monsoon onset-induced changes of autotrophic pico - and nanoplankton in the largest monsoonal estuary along the west coast of India. Environ Monit Assess. 188, 93.

- Aston, S. R., 1980. Nutrients, dissolved gases and general biogeochemistry in estuaries. In: Olausson E, Cato I (Eds.), Chemistry and Biogeochemistry of Estuaries. John Wiley & Sons, New York, pp. 233-262.
- Azis, A. P. K., Nair, N. B., 1981. Certain aspects of pollution in the aquatic ecosystems of the south west coast of India. Proceeding of Seminar on Status of Environmental Studies in India. 345-356.
- Balachandran, K. K., 2001. Chemical oceanographic studies of the coastal waters of Cochin, India: Ph.D. thesis, Cochin University Science and Technology, Cochin. pp. 250.
- Balachandran, K. K., Jayalakshmy, K. V., Laluraj, C. M., Nair, M., Joseph, T., Sheeba, P., 2008b. Step-up multiple regression model to compute chlorophyll *a* in the coastal waters of Cochin, southwest coast of India. Environ Monit Assess. 139, 217-226.
- Balachandran, K. K., Joseph, T. Nair, M., Sankaranarayanan, V. N., Kesavadas, V., Sheeba, P., 2003. Geochemistry of surficial sediments along the central southwest coast of India-seasonal changes in regional distribution. J. Coastal Res. 19, 664-683.
- Balachandran, K. K., Laluraj, C. M., Jyothibabu, R., Madhu, N. V., Muraleedharan, K. R., Vijay John Gerson., 2008a. Hydrography and biogeochemistry of the northwestern Bay of Bengal and the northeastern Arabian Sea during winter monsoon. J. Mar. Syst. 73, 76-86.
- Balachandran, K. K., Laluraj, C. M., Martin, G. D., Srinivas, K., Venugopal, P., 2006. Environmental analysis of heavy metal deposition in a flow-restricted tropical estuary and its sdjacent Shelf. Environ. Forensics. 7, 345-351.
- Balachandran, K. K., Laluraj, C. M., Nair, M., Joseph, T., Sheeba, P., Venugopal, P., 2005. Heavy metal accumulation in a flow restricted tropical estuary. Estuar Coastal Shelf Sci. 65, 361-370.
- Balachandran, K. K., Reddy, G.S., Revichandran, C., Srinivas, K., Vijayan, P. R., Thottam, T. J., 2008c. Modelling of tidal hydrodynamics for a tropical

ecosystem with implications for pollutant dispersion (Cochin estuary, South India). Ocean Dyn. 58, 76-88.

- Balachandran, K. K., Sankaranarayanan, V. N., Joseph, T., Nair, M., 1999. Non concervative controls on distribution of dissolved silicate in Cochin backwater. Proceedings of the Fourth Indian Fisheries Forum, 77-79.
- Balachandran, K. K., Thresiamma, J., Maheswari, N., Nair, K. K. C., Joseph, P. S., 2002. The complex estuarine formation of six rivers (Cochin backwater system on west coast of India) Sources and distribution of trace metals and nutrients. In Assessment of material fluxes to the coastal zone in south east Asia and their impacts, APN/SASCOM/LOICS Regional workshop. Negombo, Sri Lanka, 8-11 December, pp. 103-113.
- Balaram, V. and Rao, T. G., 2002. Rapid multi-element determination in geological samples by microwave acid digestion and ICP-MS. Proc. 3rd ISMAS (BC) Bangalore. pp. 22-27.
- Balashov, Yu. A., Ronov, A. B., Migdisov, A. A., Turanskaya, N. V., 1964. The effect of climate and facies environment on the fractionation of the rare earths during sedimentation. Geochim. Int. 5, 951-969.
- Balchand, A. N., 2001. Environmental pollutants and related health hazards: Implications and regulatory procedures. Environmental Awareness. 24, 31-37.
- Balchand, A. N., Rasheed, K., 2000. Assessment of short term environmental impacts on dredging in a tropical estuary. Terra et Aqua.79, 16-25.
- Balls, P. W., 1992. Nutrient behavior in two contrasting Scottish estuaries, the Forth and the Tay. Oceanologica Acta, 15, 261-277.
- Bapardekar, M. V., De Sousa, S. N., Zingde, M. D., 2004. Biogeochemical budgets for Tapi estuary. In Final Report for APN Project: An Assessment of Nutrient, sediment and carbon fluxes to the coastal zone in South Asia and their relationship to human activities, Sri Lanka National Committee of IGBP; Sri Lanka Association for the Advancement of Science. pp. 359.

- Bendschneider, K., Robinson, R., 1952. A new spectrophotometric method for the determination of nitrite in sea water. J. Mar. Res. 11, 87-96.
- Benjamin, M. M., Honeyman, B.D., 2000. Trace Metals. In: Jacobson, M. C., Charlson, R. J., Rodhe, H., Orians, G. H. (eds). Earth System Science - from Biogeochemical Cycles to Global Change, Academic Press, New York. pp. 377-411.
- Bernhard, M., Brinckman, F. E., Irgolic, K. J., 1986. Why speciation. In: Bernhard, M., Brinckman, F. E., Sadler, P.J. (Eds.), The Importance of chemical speciation in environmental processes. Springer-Verlag, Berlin. pp. 7-14.
- Beukema, J. J., 1991. Changes in composition of bottom fauna of a tidal-flat area during a period of eutrophication. Mar. Biol. 111, 293-301.
- Bhavya, P. S., Kumar, S., Gupta, G. V. M., Sudheesh, V., Sudharma, K. V., Varrier, D. S., 2015. Nitrogen uptake dynamics in a tropical eutrophic estuary (Cochin, India) and adjacent coastal waters. Estuar. Coast. 39, 54-67.
- Bhavya, P. S., Sanjeev, K., Gupta, G. V. M., Sudheesh, V., 2017. Carbon uptake rates in the Cochin estuary and adjoining coastal Arabian Sea. Estuar. Coast. 40, 447-456.
- Bianchi, T.S., 2007. Biogeochemistry of Estuaries. Oxford University Press, New York.
- Bianchi, T.S., Pennock, J. R., Twilley, R. R. (Eds), 1999. Biogeochemistry of Gulf of Mexico estuaries: implications for management. In: Biogeochemistry of Gulf of Mexico Estuaries, John Wiley, New York. pp. 407-421.
- Bincy, J., Revichandran, C., Naveen Kumar, K. R., 2013. Salt intrusion study in Cochin estuary-using empirical model. Indian J Geo Mar Sci. 42, 304-313.
- Bindu, K. R., Deepulal, P. M., Gireeshkumar, T. R., Chandramohanakumar, N., 2015.
 Evaluation of heavy metal enrichment in Cochin estuary and its adjacent coast: Multi-variate statistical approach. Environ Monit Assess. 187, 519.
- Bremer, P.J., Loutit, M W., 1986. Bacterial polysaccharide as a vehicle for entry of Cr (III) to food chain. Mar. Environ. Res. 20, 235-248

- Bricker, S. B., Clement, C. G., Pirhalla, D. E., Orlando, S. P., Farrow, D. G. G., 1999. National Estuarine Eutrophication Assessment: Effects of Nutrient Enrichment in the Nation's Estuaries. National Ocean Service, NOAA, Silver Springs, MD. pp. 1-47.
- Brown, S., Masra, O., Sathaye, J., 2000. Project-based activities. In: Watson, R. T.,
 Noble, I. R., Bolin, B., Ravindrananth, N. H., Verrado, D. J., Dokken, D. J.
 (Eds.), Land use, land use change and forestry, Intergovernmental Panel on
 Climate Change (IPCC). Cambridge University Press, New York. pp. 285-338.
- Bruland, K.W., Lohan, M.C., 2003. Controls of trace metals in seawater. In: Turekian, K. K., Holland, H. D. (Eds.), Treatise on Geochemistry. Elsevier Ltd., pp. 23-47.
- Bucker, A., Crespo, P., Frede, H. G., Vache, K., Cisneros, F., Breuer, L., 2010. Identifying controls on water chemistry of tropical cloud forest catchments: combining descriptive approaches and multivariate analysis. Aquat. Geochem. 16, 127-149.
- Burton, J. D., Liss, P. S., 1976. (Eds.). Estuarine Chemistry, Academic Press, London, pp. 1-36.
- Cammack, R., Joannou, C. L., Cui, X. Y., Torres Martinez, C., Maraj, S. R., Hughes, M. N., 1999. Nitrite and nitrosyl compounds in food preservation. Biochim. Biophys. Acta Bioenerg. 1411, 475-488.
- Canadian Council of Ministers of the Environment, 1999. Canadian sediment quality guidelines for the protection of aquatic life: Summary tables. In: Canadian Environmental Quality Guidelines. Winnipeg, Manitoba.
- Carey, J. H., 1990. The influence of a salt wedge and tidal flow dynamics on contaminant path ways in the Fraser river estuary, British Columbia. In: Michaelis, W. (eds.), Estuarine Water Quality Management. Springer-Verlag, Berlin. pp 203-208.
- Chakraborty, P., Chakraborty, S., Vudamala, K. Sarkar, A., Nath, B. N., 2016. Partitioning of metals in different binding phases of tropical estuarine

sediments: importance of metal chemistry, Environ. Sci. Pollut. Res. 23, 3450-3462.

- Chambers, R. M., Fourqurean, J. W., Hollibaugh, J. T., Vink, S. W., 1995. Importance of terrestrially derived particulate phosphorous to phosphorous dynamics in a west coast estuary, Estuar Coast. 18, 518-526.
- Chandramohan, P. V., 1989. Salient features of density current at Cochin. Proc. 3rd National Conference on Dock & Harbour Engineering, Surathkal. pp. 607-612.
- Cherian, P. T., 1990. Impact of Idukki hydroelectric project on the population of Villorita Cyprinoides Var. Cochinensis (Hanley) inhabiting the estuarine waters of Muvattupuzha river in Kerala. Rec. Zool. Surv. India, 87, 215-222.
- Church, T.M., 1986. Biogeochemical factors influencing the residence time of micronutrients in a large tidal estuary, Delaware Bay, Mar. Chem. 18, 393-406.
- CWC, 2014. Status of Trace and Toxic Metals in Indian Rivers. River Data Directorate Planning & Development Organisation, Ministry of Water Resources, Central Water Commission, New Delhi, p. 27.
- Dame, R. F., Allen, D. M., 1996. Between estuaries and the sea. J. Exp. Mar. Biol. Ecol. 200, 169-185.
- Dame, R. F., Chliders, D., Koepfler, E., 1992. A geohydrologic continuum theory for the spatial and temporal evolution of marsh - estuarine ecosystems. Neth. J. Sea Res. 30, 63-72.
- Dang, D. H., Lenoble, V., Durrieu, G., Mullot, J. U., Mounier, S., Garnier, C., 2014. Sedimentary dynamics of coastal organic matter: an assessment of the pore water size/reactivity model by spectroscopic techniques. Estuar. Coast. Shelf Sci. 151, 100-111.
- Deepulal, P. M., Gireeshkumar, T. R., Sujatha, C.H., 2012. Behaviour of REE's in a tropical estuary and adjacent continental shelf Southwest coast of India: Evidence from anomalies. J. Earth Syst. Sci. 121, 1215-1227.
- Degens, E. T., Ittekot, V., 1984. A new look at clay-organic interactions. In: Knetsch,F.G. (Ed.), Nord-Sud profil, Zentraleuropa-Mittelmeerraum Afrika,

Mitteilungen aus dem Geologish-Paleontologischen, 56. Institute der Universitat Hamburg. Heft, pp. 229-248.

- Dehairs, F., Rao, R. G., Chandramohan, P., Raman, A., Marguiller, S., Hellings, L., 2000. Tracing mangrove carbon in suspended matter and aquatic fauna of the Gautami-Godavari Delta, Bay of Bengal, India. Hydrobiology. 431, 225-241.
- Desouza, M. J. B. D., Nair, S., Chandramohan, D., 2000. Phosphate solubilizing bacteria around Indian peninsula. Indian J Mar Sci. 29, 48-51.
- Diaz, R.J., Rosenberg, R., 1995. Marine benthic hypoxia: a review of its ecological effects and the behavioral responses of benthic macrofauna. Oceanogr. Mar. Biol. Ann. Rev. 33, 245-303.
- Dineshkumar, P. K., Sankaranarayanan, V. N., Saraladevi, K., 1994. Cochin backwaters: An introduction to the system, prior studies, Historical trends and future implications", Indian J. Environ. Prot. 14, 98-102.
- DMG, 2016. District survey report of minor minerals, Ernakulam State, Department of Mining and Geology, Govt of Kerala. p. 23-24.
- Doval, M. D., Alvarez-Salgado, X. A., Perez, F. F., 1997. Dissolved organic matter in a temperate embayment affected by coastal upwelling. Marine Ecol. Progress Ser. 157, 21-37.
- Du Laing, G., Rinklebe, J., Vandecasteele, B., Meers, E., Tack, F. M. G., 2009. Trace metal behaviour in estuarine and riverine floodplain soils and sediments: A review. Sci. Total Environ. 407, 3972-3985
- Dupre, B., Gaillardet, J., Rousseau, D., Allegre, C. J., 1996. Major and trace elements of river-borne material: The Congo basin. Geochim. Cosmochim. Acta. 60, 1301-1321.
- Dyer, K. R., 1991. Circulation and mixing in stratified estuaries. Mar. Chem. 32, 111-120.
- El Wakeel, S. K., Riley, J. P., 1957. The determination of organic carbon in marine muds. ICES J. Mar. Sci. 22, 180-183.

- Elderfield, H., 1978. Chemical variability in estuaries. In: Bio-geochemistry of Estuarine Sediments. Unesco, Paris. pp. 171-178.
- Elderfield, H., Sholkovitz, E. R., 1987. Rare earth elements in the porewaters of reducing near shore sediments. Earth Planet. Sci. Lett. 82, 280-288.
- Elliotta, M., McLusky, D. S., 2002. The need for definitions in understanding estuaries. Estuar. Coast. Shelf Sci. 55, 815-827.
- European Commission, 2012. Report from the Commission to the European Parliament and the Council on the Implementation of the Water Frame work Directive (2000/60/EC). Brussels, 14.11.2012 COM (2010). 670.
- Fairbrother, A., Wenstel, R., Sappington, K., Wood, W., 2007. Framework for metals risk assessment. Ecotoxicol. Environ. Saf. 68, 145-227.
- Falco, S., Niencheski, L. F., Rodilla, M., Romero, I., Gonzalez del Rio, J., Sierra, J.P., Mosso, C., 2010. Nutrient flux and budget in the Ebro estuary.Estuar. Coast. Shelf Sci. 87, 92-102.
- Fasham, M. J. R. (Ed.). 2003. Ocean biogeochemistry: the role of the ocean carbon cycle in global change,. Berlin: Springer. pp. 53-97.
- Fisher, N. S., Bohe, M., Teyssie. J. L., 1984. Accumulation and toxicity of Cd, Zn, Ag and Hg in four marine phytoplankters. Mar. Ecol. Prog. Ser. 18, 201-213.
- Folk, R. L., 1980. Petrology of sedimentary rocks. Austin, Texas: Hemphill. 26-27.
- Forstner, U., 1981. Metal transfer between solid and aqueous phase. In: Forstner, U., Wittmann, G.T.W. (Eds.), Metal Pollution in the Aquatic Environment. Springer verlag, Berlin-Heidelberg - New York, pp. 197 - 270.
- Gaillardet, J., Viers, J., Dupre, B., 2003. Trace elements in rivers. In: Holland, H. D., Turekian, K. K. (Eds.), Treatise on Geochemistry: Surface and Ground Water, Weathering and Soils, vol. 5. Elsevier B. V., pp. 225-272.
- Gao, Y., Lesven, L., Gillan, D., Sabbe, K., Billon, G., De Galan, S., Elskens, M., Baeyens, W., Leermakers, M., 2009. Geochemical behavior of trace elements in sub-tidal marine sediments of the Belgian coast. Mar. Chem. 117, 88-96.

- Gattuso, J. P., Frankignoulle, M., Wollast, R. 1998. Carbon and carbonate metabolism in coastal aquatic ecosystems. Annu. Rev. Ecol. Syst. 29, 405-434.
- George, R., Martin, G. D., Nair, S. M., Thomas, S. P., Jacob, S., 2016. Geochemical assessment of trace metal pollution in sediments of the Cochin backwaters. Environ. Forensics. 17, 156 - 171.
- Gireeshkumar, T. R., Deepulal, P. M., Chandramohanakumar, N., 2013a. Distribution and sources of sedimentary organic matter in a tropical estuary, south west coast of India (Cochin estuary): a baseline study. Mar. Pollut. Bull. 66, 239 -245.
- Gireeshkumar, T. R., Deepulal, P. M., Chandramohanakumar, N., 2013b. Phosphorous speciation in surface sediments of the Cochin estuary. Environ Monit Asses, 185. 2535-2552.
- Goldstein, S. J., Jacobsen, S. B., 1988. Rare earth elements in river waters. Earth Planet. Sci. Lett. 89, 35-47.
- Gonzalez, J. L., Thouvenin, B., Dange, C., 2007. Role des particules sur le comportement et la speciation demetaux traces: example du cadmium. La Houille Blanche. 4, 56-62.
- Gopalan, U. K., Doyil, T. V., Udayavarma, P., Krishanankutty, M., 1983. The shrinking of Cochin back water. J. Mar. Biol. Assoc. India, 25, 131-141.
- Gopinathan, C. P., Nair, P. V. R., Nair, A. K. K., 1984. Quantitative ecology of phytoplankton in Cochin backwater. Indian J. Fish. 31, 325-346.
- Graf, J. L. 1977. Rare earth elements as hydrothermal tracers during the formation of massive sulphide deposits in volcanic rocks; Econ. Geol. 72 527–548.
- Grasshoff, K., Ehrhardt, M., & Kremling, K. (Eds.). 1999. Methods of seawater analysis. Weinheim: Verlag Chemie. Grasshoff, K., Ehrhardt, M., Kremling, K., et al., 1983. Methods of seawater analysis (2nded.). Deerfield Beach: Weinheim.
- Grim, R. E. 1968. Clay mineralogy. New York: McGraw Hill. pp. 468.

- Gupta, G. V. M., Thottathil, S. D., Balachandran, K. K, Madhu, N. V, Nair, S., 2009. CO₂ super saturation and net heterotrophy in a tropical estuary (Cochin, India): influence of anthropogenic effect. Ecosystems. 12:1145–1157.
- Habeebrehman, H., Prabhakaran, M. P., Jacob, J., Sabu, P., Jayalakshmi, K. J., Achuthankutty, C. T., 2008. Variability in biological responses influenced by upwelling events in the Eastern Arabian Sea. J. Mar. Syst. 74, 545-560.
- Hannigan, R., Dorval, E., Jones, C., 2010. The rare earth element chemistry of estuarine surface sediments in the Chesapeake Bay. Chem. Geol. 272, 20-30.
- Haridas, P., Madhupratap, M., Rao, T.S.S., 1973. Salinity, temperature, oxygen and zooplankton biomass of the backwaters from Cochin to Alleppey. Indian J. Mar. Sci. 2, 94-102.
- Harikumar, P. S., Nasir, U. P., Rahman, M. P. M., 2009. Distribution of heavy metals in the core sediments of a tropical wetland system. Int. J. Environ. Sci. Te. 6, 225-232.
- Harmon, R. S., Lyons, W. B., Long, D. T., Ogden, F. L., Mitasova, H., Gardner, C.B., Welch, K. A., Witherow, R. A., 2009. Geochemistry of four tropical montane watersheds, Central Panama. Appl. Geochem. 24, 624-640.
- Hawkins, A. J. S., Smith, R.F. M., Bayne, B. L., Héral, M., 1996. Novel observations underlying the fast growth of suspension-feeding shellfish in turbid environments: Mytilus edulis. Mar. Ecol. Prog. Ser. 131,179-190.
- Hobbie, J. E. (Ed.). 2000. Estuarine science: the key to progress in coastal ecological research. In: Estuarine Science, A Synthetic Approach to Research and Practice, Island Press, Washington DC. pp. 1-11.
- Hoch, M. P., Kirchman, D. L., 1995. Ammonium uptake by heterotrophic bacteria in the Delaware estuary and adjacent coastal waters. Limnol.Oceanogr. 40, 886-897.
- Holmes, C.W. 1986. Trace metal seasonal variation in Texas marine sediments. Mar. Chem., 29: 13-27.

- Howarth, R. W., Jaworski, N., Swaney, D., Townsend, A., Billen, G., 2000. Some approaches for assessing human influences on fluxes of nitrogen and organic carbon to estuaries; In: Hobbie, J. E. (Ed.), Estuarine science: A synthetic approach to research and practice Island Press, Washington DC, pp. 17-42.
- Howarth, R. W., Sharpley, A., Walker, D., 2002. Sources of nutrient pollution to coastal waters in the United States: Implications for achieving coastal water quality goals. Estuaries. 25, 656-676.
- Jain, C. K., 2004. Metal fractionation study on bed sediments of River Yamuna, India. Water Res, 38, 569-578.
- Jassby, A. D., Kimmerer, W. J., Monismith, S. G., Armor, C., Cloern, J. E., Powel, T. M., Schubel, J. R., Vendlinski, T. J., 1995. Isohaline position as a habitat indicator for estuarine populations. Ecol. Appl. 5, 272-289.
- Jayakumar, D. A., Naqvi, S. W. A., Narvekar, P. V., George, M. D., 2001. Methane in coastal and offshore waters of the Arabian Sea. Mar. Chem. 74, 1-13.
- Jickells, T. D., Dorling, S., Deuser, W. G., Church, T. M., Arimoto, R., Prospero, J. M., 1998. Air-borne dust fluxes to a deep water sediment trap in the Sargasso Sea. Global Biogeochem. Cycles. 12, 311-320.
- Jiya, J., Rajesh, G., Anas, A., Lokabharathi, P.A., Shanta Nair., 2011. Heavy metal pollution exerts reduction/adaptation in the diversity and enzyme expression profile of heterotrophic bacteria in Cochin estuary, India. Environ. Pollut. 159, 2775-2780.
- Joseph M. L., 1984, The Declining trend of Biodiversity and Fish Production in consequence of pollution in the Lower Reaches of Periyar River, Status report on river Periyar, Cochin University of Science and Technology, Cochin, India, pp. 51-82
- Joseph, S., Ouseph, P. P., 2010. Assessment of nutrients using multivariate statistical techniques in estuarine systems and its management implications: a case study from Cochin Estuary, India. Water Environ J. 24, 126-132.

- Justic, D., Rabalais, N. N., Turner, R. E., Dortch, Q., 1995. Changes in nutrient structure of river-dominated coastal waters: stoichiometric nutrient balance and its consequences. Estuar. Coast. Shelf. Sci. 40, 339-356.
- Jyothibabu, R., Madhu, N. V., Jayalakshmi, K. V., Balachandran, K. K., Shiyas, C. A., Martin, G. D., Nair, K. K. C., 2006. Impact of fresh water influx on micro zooplankton mediated food web in a tropical estuary (Cochin backwaters), India. Estuar. Coast. Shelf. Sci. 69, 505-515.
- Kartha, K. N. R., Karunakaran Nair, P., 1980. Grow more prawns. Krishi Vigyan Patrika: Mariculture, series. 5. Kochi: CMFRI.
- Kennish, M. J., 1991. Ecology of Estuaries: Anthropogenic Effects. CRC Press, Florida.
- Kennish, M. J., 2002. Sediment contaminant concentrations in estuarine coastal marine environment: Potential for remobilization by boats and personal water crafts, J Coastal Res. 37, 151-178.
- Klucakova, M., Pavlikova, M., 2017. Lignitic humic acids as environmentally friendly adsorbent for heavy metals. J. Chem; doi:10.1155/2017/7169019.
- Koroleff, F., 1983. Determination of silicon. In: Grasshoff, K., Ehrhardt, M., Kremling, K. (Eds.), Methods of sea water analysis, Verlag chemie, Weinheim. pp. 174-183.
- Koski -Vahala, J., Hartikainen, H., Tallberg, P., 2001. Phosphorus mobile zation from various sediment pools in response to increased pH and silicate concentration. J. Environ. Qual. 30, 546-552.
- Krumbein, W. C., Pettijohn, F. J., 1938. Manual of sedimentary petrography, New York: Appleton Century Crofts Inc. pp.549.
- KSPCB, 1982. Kerala State Pollution Control Board, Environmental Status Report on Greater Cochin, Kerala (with special reference to water pollution). pp. 32.
- Kumar, R., Kumar, G. 2010. Spatial variability and contamination of heavy metals in the inter-tidal systems of a tropical environment, 4, 691-700.

- Kundrass, H. R., 1987. Sedimentary models to estimate the heavy mineral potential of shelf sediments, In: Teleki, P. G. et al. (eds.), Marine minerals, Reidel Publishing Comp. pp. 39-56.
- Kunjikrishna Pillai, V. 1991. Studies on the hydrobiology and pollution of the Vembanad Lake and adjacent waters. Ph.D. Thesis. Cochin University of Science and Technology, Cochin. pp.148.
- Kurup, B. M. 1992. Appraisal of aquatic ecosystem of the 'EUS'struck regions of Kuttanad (Kerala). Fish. Chimes. 12, 27-33.
- Kurup, M. B., 1983. Studies on the systematics and biology of the fishes of the Vembanad Lake. Ph.D. Thesis. Cochin University of Science and Technology, Cochin.
- Laluraj, C. M., 2009. Geochemistry of transition, non-transition and rare earth elements in the surficial sediments of Continental Shelf of Kerala and an annex to the Cochin estuarine system. Ph.D. thesis, Cochin University Science and Technology, Cochin.
- Leite, M. G. P., Fujaco, M. A. G., Nalini Jr., H. A., Castro, P. T. A., 2010. Influence of geology in the geochemistry signature of Itacolomi State Park waters, Minas Gerais- Brazil. Environ. Earth Sci. 60, 1723-1730.
- Leonard, L.A., Luther, M.E. 1995. Flow hydrodynamics in tidal marsh canopies. Limnol.Oceanogr. 40, 1474-1484.
- Lewis, D.W., 1984. Practical sedimentology. Hutchinson Ross Publishing Company, Pennsylvania, p. 227.
- Lindberg, S. E., Andre, A. W., Harris, R.C., 1975. Geochemistry of mercury in the estuarine environment. In: Cronin, L.E. (Ed.) Estuarine Research. vol. I. Academic Press, New York, pp. 64-107.
- Liu, Y. G., Miah, M. R. U., Schmitt, R. A., 1988. Cerium: A chemical tracer for paleo-oceanic redox conditions. Geochim. Cosmochim. Acta. 52. 1361-1371.

- Longphuirt, S. N., Boyle, S.O., Stengel, D. B., 2015. Environmental response of an Irish estuary to changing land management practices. Sci. Total Environ. 521, 388-399.
- Louis, Y., Garnier, C., Lenoble, V., Omanovic, D., Mounier, S., Pizeta, I., 2009. Characterisation and modelling of marine dissolved organic matter interactions with major and trace cations. Mar. Environ. Res. 67, 100-107.
- Lu, X. Q., Zhang, Y., Liu, H., Xing, M., Shao, X., Zhao, F., Li, X., Liu, Q., Yu, D., Yuan, X., Yuan, M., 2014. Influence of early diagenesis on the vertical distribution of metal forms in sediments of Bohai Bay, China. Mar. Pollut. Bull. 88, 155-161.
- Luoma, S. N., 1983. Bioavailability of trace metals to aquatic organisms: A review. Sci. Total Environ. 28, 1-22.
- Luoma, S. N., 1989. Can we determine the biological availability of sediment-bound trace elements? Hydrobiologia. 176/177, 379-396.
- Luoma, S. N., Phillips, D. J. H., 1988. Distribution variability and impacts of trace elements in San Francisco Bay. Mar. Pollute Bull. 19, 413-425.
- Luther III, G. W., Rozan, T. F., Witter, A., Lewis, B., 2001. Metal-organic complexation in the marine environment. Geochem. Trans. 2, 65.
- Machado, A. A. D., Spencer, K., Kloas, W., Toffolon, M., Zarfl, C., 2016. Metal fate and effects in estuaries: A review and conceptual model for better understanding of toxicity. Sci Total Environ. 541, 268-281.
- Mackas, D. L., Harrison, P. J., 1997. Nitrogenous nutrient sources and sinks in the Juan de Fuca Straight / Straightof Georgia / Puget Sound estuarine system: assessing the potential for eutrophication. Estuar. Coast. Shelf. Sci. 4, 1-21.
- Mackas, D.L., Harrison, P. J., 1997. Nitrogenous nutrient sources and sinks in the Juan de Fuca Straight / Straight of Georgia / Puget Sound estuarine system: Assessing the potential for eutrophication. Estuarine Coastal Shelf Sci. 4, 1-21.

- Madhu, N. V, Reny, P. D., Paul, M., Ullas, N., Resmi, P., 2011. Occurrence of red tide caused by Karenia mikimotoi (toxic dinoflagellate) in the south west coast of India, 40, 821-825.
- Madhu, N. V., Jyothibabu, R., Balachandran, K. K., 2010. Monsoon-induced changes in the size-fractionated phytoplankton biomass and production rate in the estuarine and coastal waters of south west coast of India. Environmental Monitoring Assessment, 166, 521-528.
- Madhu, N. V., Jyothibabu, R., Balachandran, K. K., Honey, U. K., Martin, G. D., Vijay, J. G., Shiyas, C. A., Gupta, G. V. M., Achuthankutty, C.T., 2007.
 Monsoonal impact on planktonic standing stock and abundance in a tropical estuary (Cochin backwaters India). Estuar. Coast. Shelf Sci., 73, 54-64.
- Madhupratap, M., 1977. Studies on the ecology of zooplankton of Cochin backwaters (a tropical estuary). Ph.D. Thesis, Cochin University of Science and Technology, Cochin, India.
- Madhupratap, M., 1987. Status and strategy of zooplankton of tropical Indian estuaries: A review. Bull. Plankton Soc. Japan. 34, 65-81.
- Maheswari, N., Balachandran, K. K., Sankaranarayanan, V. N., Thressiamma, J., 1997. Heavy metals in fishes from coastal waters of Cochin, south-west coast of India. Indian J. Mar. Sci. 26, 98-100.
- Maheswari, N., Jayalakshmy, K. V., Balachandran, K. K., Joseph, T., 2006. Bioaccumulation of toxic metals by fish in a semi-enclosed tropical ecosystem. Environ. Forensics. 7, 197-206.
- Mallik, T. K., Vasudevan, V., Verghese, P.A., Machado, T., 1987. The black sand placer deposits of Kerala beach, southwest India: Marine Geology, 77, 129-150.
- Manju, P. Nair., 2014. Geochemical metal fractionation profile of the core sediments in the Cochin estuarine system. Ph.D thesis. Cochin University of Science and Technology, Cochin. pp. 49-63.

- Manju, P. Nair., Sujatha C.H., 2012. Biogeochemical quality assessment of the sediments in Kerala coast. Int J Environ Sci. 3, 707-719.
- Manju, P. Nair., Sujatha, C. H., 2013. Biennial metal discrepancy in the surface sediment of Cochin Estuarine System. Int J Recent Sci Res. 4, 1365-1369.
- Manoj, M. C., Biswajeet, T., Vandana., P., 2016. Rare earth element distribution in tropical coastal wetland sediments: a case study from Vembanad estuary, south west India. Arab J Geosci. 9,197.
- Martin, G. D., George, R., Shaiju, P., Muraleedharan, K. R., Nair, S. M., Chandramohanakumar, N., 2012. Toxic metals enrichment in the surficial sediments of a eutrophic tropical estuary (Cochin backwaters, southwest coast of India). Sci World J. doi:10.1100/2012/972839.
- Martin, G. D., Jyothibabu, R., Madhu, N. V., Balachandran, K. K., Nair, M., Muraleedharan, K. R., Arun, P. K., Haridevi, C. K., Revichandran, C., 2013. Impact of eutrophication on the occurrence of Trichodesmium in the Cochin backwaters, the largest estuary along the west coast of India. Environ. Monit. Assess. 185, 1237-1253.
- Martin, G. D., Muraleedharan, K. R., Vijay, J. G., Rejomon, G., Madhu, N. V., Shivaprasad, A., Haridevi, C. K., Nair, M., Balachandran, K. K., Revichandran, C., Jayalakshmy, K. V., Chandramohanakumar, N., 2010. Formation of anoxia and denitrification in the bottom waters of a tropical estuary, southwest coast of India. Biogeosciences Discuss. 7, 1751-1782.
- Martin, G. D., Nisha P. A., Balachandran K. K., Madhu. N.V., Nair, M., Shaiju, P., Joseph, T., Srinivas, K., Gupta G. V. M., 2011. Eutrophication induced changes in benthic community structure of a flow-restricted tropical estuary (Cochin backwaters), India. Environ. Monit. Assess. 176, 427-438.
- Martin, G. D., Vijay, J. G., Lalu raj, C. M., Madhu, N. V., Joseph, T., Nair, M., Gupta, G. V. M., Balachandran, K. K., 2008. Fresh water influence on nutrient stoichiometry in a tropical estuary, south west coast of India. Appl Ecol Env Res. 5, 57-64.

- Martin, J. M., Whitfield, M., 1983. The significance of river input of chemical elements to the ocean. In Wongs, C. S., Boyle, E., Druland, K. W., Burton, J. D., Goldberg, E. D. (Eds.), Trace metals in sea water. New York: Plenum. pp. 265-296.
- Mckee, L. J., Eyre, B. D., Hossain, S. 2000. Transport and retention of nitrogen and phosphorus in the sub-tropical Richmond River estuary, Australia - a budget approach. Biogeochemistry, 50, 241-278.
- McLennan, S. M., 1989. Rare earth elements in sedimentary rocks: influence of provenance and sedimentary processes. In: Lipin, B. R., McKay, G.A. (Eds.), Geochemistry and Mineralogy of the Rare Earth Elements. Min. Soc. Am. Rev. Mineral. 21, pp. 169-200.
- McManus, L. T., Kremer, H. H., Marshell, C. J. I., 2001. SARC/WOTRO/LOICZ: Biogeochemical and human dimensions of coastal functioning and change in Southeast Asia. Final report of the SARC/WOTRO/LOICZ project 1996-1999. LOICZ, Reports and Studies, vol. 17, LOICZ, Texel, The Netherlands, pp ii-277.
- Mendil, D., Uluozlu, O. D., 2007. Determination of trace metals in sediment and five fish species from lakes in Tokat, Turkey. Food Chem. 101, 739-745.
- Menon, N. N., Balchand, A. N., Menon, N. R., 2000. Hydrobiology of the Cochin backwater system - a review. Hydrobiologia, 430, 149-183.
- Menon, N. R., 1986. Combined toxicity of silver and copper on the sub-lethal response of Perna viridis. Proc. Natn. Sem. on Mussel Watch. University of Cochin. pp. 13-14.
- Michaelis, W., 1990. The bilex concept research supporting public water quality surveillance in estuaries. In: Michaelis, W. (Ed.), Estuarine water Quality Management, Springer-Verlag. Berlin. pp. 79-88.
- Middelburg, J. J., Nieuwenhuize, J., 2000. Nitrogen uptake by heterotrophic bacteria and phytoplankton in the nitrate-rich Thames River. Mar. Ecol. Prog. Ser. 203, 13-21.

- Millennium Ecosystem Assessment, 2005. Ecosystems and Human Well-being: Wetlands and Water Synthesis. World Resources Institute, Washington, DC. USA.
- Millward, G.E., Turner, A., 1995. Trace metals in estuaries. In: Salbu, B., Steinnes, E. (Eds.), Trace Elements in Natural Waters, CRC Press, Boca Raton, FL. pp. 223-245.
- Miranda, J., Balachandran, K. K., Ramesh, R., Wafar, M., 2008. Nitrification in Kochi backwaters. Estuarine, Coastal and Shelf Science, 78, 291-300.
- Moffett, J. W., Vedamati, J., Goepfert, T. J., Pratihary, A., Mangesh, G., Naqvi, S. W. A., 2015. Biogeochemistry of iron in the Arabian Sea. Limnol Oceanogr. 60, 1671-1688.
- Mohan, M., Augustine, T., Jayasooryan, K K., Shylesh, C. M. S., Ramaswamy, E. V., 2012. Fractionation of selected metals in the sediments of Cochin estuary and Periyar River, south-west coast of India. Environmentalist, 32, 383-393.
- Montani, S., Magni, P., Shimamoto, M., Abe, N., Okutani, K., 1998. The effect of a tidal cycle on the dynamics of nutrients in a tidal estuary in the Seto Inland Sea. Japan. Journal of Oceanography, 54, 65-76.
- Moore, W. S., 2003. Sources and fluxes of submarine groundwater discharge delineated by radium isotopes. Biogeochemistry. 66, 75-93.
- Morel, F. M. M., Dzombak, D. A., Price, N. M., 1991. Heterogeneous reactions in coastal waters. In: Mantoura. R. F. C., Martin, J. M., Wollast, R. (eds.). Ocean Margin Processes in Global Change. John Wiley, New York. pp.165-180.
- Morris, A. W., Mantoura, R. F. C., Bale, A. J., Howland, R. J. M., 1978. Very low salinity regions of estuaries: important sites for chemical and biological reactions. Nature, 274, 678 - 680.
- Morris, A.W., 1990. Kinetic and equilibrium approaches to estuarine chemistry. Sci. Total Environ. 97/98, 253-266.
- Morris, R. J., Mc Cartney, M. J., Bone, Q., 1982. The distribution of sterols and stanols in the tunicate Cionaintestinalis. J. Mar. Biol. Ass. U.K. 62, 117-123.

- Mukhopadhyay, S. K., Biswas, H., De, T. K., Jana, T. K., 2006. Fluxes of nutrients from the tropical River Hooghly at the land ocean boundary of Sundarbans, NE Coast of Bay of Bengal, India. J. Mar. Syst. 62, 9-21.
- Muller, P. J., Suess, E., Ungerer, A. E., 1986. Amino acids and amino sugars of surface particulate and sediment trap material from waters of the Scotia Sea. Deep Sea Res. Part 1. 33, 829-838.
- Murphy, K., Dymond, J., 1984. Rare earth element fluxes and geochemical budget in the eastern equatorial Pacific. Nature. 307, 444-447.
- Murray, R. W. and M. Leinen. 1993. Chemical transport to the sea floor of the equatorial pacific Ocean across a latitude transect at 135°W: Tracing sedimentary major, trace, and rare earth element fluxes at the equator and the Intertopical Convergence Zone. Geochim.Cosmochim.Acta 57.4141-4163.
- Nadaska, G., Lesny, J., Michalik, I., 2012. Environmental aspect of manganese chemistry. Hungarian Journal of Sciences, ENV-100702-A, 1-16
- Naik, H., 2000. Budgets for Periyar estuary, Kerala. Presented at 'Regional Training Workshop on Biogeochemical Budgeting and Socio-Economic modeling for Coastal Scientist'. APN/SASCOM/LOICZ, 18-22 September, Colombo.
- Nair, C. K, Balchand, A. N, Jacob Chacko. 1993. Sediment characteristics in relation to changing hydrography of Cochin estuary. Indian J Mar Sci. 22, 33-36.
- Nair, C. K., 1992. Chemical Partitioning of trace metals in sediments of a tropical estuary. Ph.D. thesis, Cochin Univ. Science and Technology.
- Nair, C. K., Balchand, A. N., Nambisan, N. P. K., 1991. Heavy metal speciation in sediments of Cochin estuary determined using chemical extraction techniques. Sci. Total Environ. 102, 113-128.
- Nair, K. K. C., Sankaranarayanan, V. N., Gopalakrishnan, T. C., Balasubramanian, T., Lalithambikadevi, C. B., Aravindakshan, P. N., 1988. Environmental conditions of some paddy-cum-prawn culture fields of Cochin backwaters, south-west coast of India. Indian J. Mar. Sci. 17, 24-30.

- Nair, P. V. R., Joseph, K. J., Balachandran, V. K., Pillai, V.K., 1975. A study on the primary production in the Vembanad Lake. Bull. Dept. Mar. Sci. Univ. Cochin. 7, 161-170.
- Nair, S. M., Balchand, A. N., Prashob, K. V., Shaiju, P., 2013. Mobilization of trace metals in a tropical turbid estuary influenced by a monsoon season. Environ Earth Sci 70, 807-819.
- Naqvi, S. W. A., Jayakumar, D. A., 2000. Ocean biogeochemistry and atmospheric composition: Significance of Arabian Sea. Current Science, 78, 289-299.
- Naqvi, S. W. A., Naik, H., Jayakumar, D. A., Shailaja, M. S., Narvekar, P. V., 2006. Seasonal oxygen deficiency over the western continental shelf of India. In: Neretin, L. N. (Ed.), Past and present water column anoxia. pp. 195-224.
- Nath, B. N., Bau, M., Rao, B. R., Rao, Ch. M., 1997. Trace and rare earth elemental variation in Arabian Sea sediments through a transect across the oxygen minimum zone. Geochim. Cosmochim. Acta. 61. 2375-2388.
- Nath, B. N., Kunzendorf, H., Plugeer, W. L., 2000. Influence of provenance, weathering and sedimentary processes on the elemental ratios of the fine grained fraction of bed load sediments from the Vembanad lake and the adjoining continental shelf, southwest coast of India. J. Sedimen. Res. 70, 1081.
- Nriagu, J. O., Pacyna, J. M., 1988. Quantitative assessment of worldwide contamination of air, water and soils by trace metals. Nature. 333, 134-139.
- Ouseph, P. P., 1992. Dissolved and particulate trace metals in the Cochin Estuary. Mar. Pollut. Bull. 24, 86-192.
- Owens, R. E., Balls, P. W., Price, N. B., 1997. Physico-chemical processes and their effects on the composition of suspended particulate material in estuaries: Implications for monitoring and modeling. Mar. Pollut. Bull, 34, 51-60.
- Padmalal, D., Maya, K., Seralathan, P., 1997. Geochemistry of Cu, Co, Ni, Zn, Cd and Cr in the surficial sediments of a tropical estuary, southwest coast of India: a granulometric approach. Environ. Geol. 31, 85-93.

- Page, H. M., Petty, R. L., Meade, D. E., 1995. Influence of watershed runoff on nutrient dynamics in a southern California salt marsh. Estuar. Coast. Shelf Sci. 41, 163-180.
- Palmer, M. R., Elderfield, H., 1986. Rare earth elements and neodymium isotopes in ferromanganese oxide coatings of Cenozoic foraminifera from the Atlantic Ocean. Geochim. Cosmochim. Acta 50, 409-417.
- Parker, C. A., O'Reilly, J. E., 1991. Oxygen depletion in Long Island Sound: a historical perspective. Estuaries. 14, 248-265
- Parvathi, A., Jasna, V., Jina, S., Jayalakshmy, K. V., Lallu, K. R., Madhu, N. V., Muraleedharan, K. R., Naveen Kumar, K. R., Balachandran, K. K., 2015. Effects of hydrography on the distribution of bacteria and virus in Cochin estuary, India. Ecol. Res. 30, 85-92.
- Pattan, J. N., Pearce, N. J. G., Mislankar, P. G., 2005. Constraints in using Cerium anomaly of bulk sediments as an indicator of paleo bottom water redox environment: A case study from the Central Indian Ocean Basin. Chem. Geol. 221, No. 3-4. 260-278.
- Pattan, J. N., Pearce, N. J. G., Mislankar, P.G., 2005. Constraints in using Ceriumanomaly of bulk sediments as an indicator of paleo bottom water redox environment: A case study from the Central Indian Ocean Basin. Chem. Geol. 221, 260-278.
- Paul, P. C., Pillai, K. C., 1983. Trace metals in tropical river environment distribution. Water Air Soil Pollut. 19, 63-73.
- Pennock, J. R., Sharp, J. H., Schroeder, W. W., 1994. What controls the expression of estuarine eutrophication? Case studies of nutrient enrichment in the Delaware Bay and Mobile Bay estuaries, USA. In: Dyer, K. R., Orth, R. J. (Eds.), Changes in fluxes in estuaries: implications from science to management. Fredensborg: Olsen and Olsen. pp. 139-146.

- Pereira-Filho, J., Schettini, C. A. F., Rorig, L., Siegle, E., 2001. Intratidal variation and net transport of dissolved inorganic nutrients, POC and chlorophyll *a* in the Camboriu River Estuary, Brazil. Estuarine Coastal Shelf Sci. 53, 249-257.
- Petersen, W., Wallmann, K., Li, P. L., Schroeder, F., Knauth, H. D., 1995. Exchange of trace elements of the sediment-water interface during early diagenic processes. Aust. J. Mar. Freshwater Res. 46, 19-26.
- Pichaimani, S. V., Jonathan, M. P., Srinivasalu, S., Rao, R. N., Mohan, S. P., 2008. Enrichment of trace metals in surface sediments from northern part of Point Calimere, SE coast of India. Environ Geol. 55, 1811-1819.
- Pillai, N. G. K., 1977. Distribution and seasonal abundance of macrobenthos of the Cochin backwaters. Ind. J. Mar. Sci. 6: 1-5.
- Pillidge, C. J., 1985. Bacterial mobilization of chromium (III) in polluted marine sediment. Ph.D thesis. University of Otago Library.
- Piper, D. Z., 1974. Rare earth elements in the sedimentary cycle: A summary. Chem. Geol. 14, 285-304.
- Piper, D. Z., Bau, M., 2013. Normalized Rare Earth Elements in Water, Sediments, and Wine: Identifying Sources and Environmental Redox Conditions. Am J Analyt Chem. 4, 69-83.
- Prasad, M. B. K., Ramanathan, A. L., Shrivastav, S. K., Anshumali, R.S., 2006. Metal fractionation studies in surfacial and core sediments in the Achankovil river basin in India. Environ Monit Assess. 121, 77-102.
- Prasanthan, V., Udayakumar, P., Ouseph, P. P., 2011. Influence of abiotic environmental factors on the abundance and distribution of Vibrio species in coastal waters of Kerala, India, Indian. J. Geo. Mar. Sci. 40, 587-592.
- Pratihary, A. K., Naqvi, S.W. A., Narvenkar, G., Kurian, S., Naik, H., Naik, R., Manjunatha, B. R., 2014. Benthic mineralization and nutrient exchange over the inner continental shelf of western India. Biogeosciences. 11, 2771-2791.
- Priju, C.P., Narayana, A.C., 2007. Heavy and trace metals in Vembanad Lake sediments. Int. J. Environ. Res. 1, 280-289.

- Priyakant, Verma, N. K., Rao, L. I. M., Singh, A. N., 2003. Surface approximation of point data using different interpolation techniques-a GIS approach; <u>http://gisdevelopment.net/</u> technology/survey/techgp0009pf.htm.
- Prosser, J. I., Nicol, G. W., 2008. Pelative contributions of Archaea and bacteria to aerobic ammonia oxidation in the environment. Environ Microbiol. 10, 2931-2941.
- Qasim, S. Z. 2003. Cochin backwaters and Vembanad. In: Qasim, S. Z. (Ed.), Indian Estuaries; ISBN: 81-7764-369-X. Allied Publishers Pvt. Ltd, Mumbai. pp. 305-382.
- Qasim, S. Z., Madhupratap, M. 1979. Changing ecology of the Cochin backwaters. In: Contributions to marine sciences. Univ. Cochin, 137-142.
- Qasim, S.Z. and Gopinathan, C.K., 1969. Tidal cycles and the environmental features of Cochin backwater (a tropical estuary). In: Proc. Indian Acad. Sci. (Bangalore, India), pp. 1183-1169.
- Rabalais, N. N., Gilbert, D., 2009. Distribution and consequences of hypoxia. In: Urban Jr, E. R., Sundby, B., Malanotte-Rizzoli, P., Melillo, J. M. (Eds.), Watersheds, Bays, and Bounded Seas Washington DC: Island Press. pp. 209-225.
- Rajagopal, M. D., 1974. On assimilation and regeneration of phosphorus in two different environment. Mahasagar, Bulletin, National Institute of Oceanography, 7, 143–149.
- Rajamani Amma, V., 1994. The distribution and partition of some of the trace metals in sediments and waters on the coastal environment, Ph.D Thesis, Cochin University of Science and Technology.
- Rajaneesh, K. M., Mitbavkar, S., Anil, A. C., Sawant, S. S., 2015. Synechococcus as an indicator of trophic status in the Cochin backwaters, west coast of India. Ecological Indicators. 55, 118-130.

- Ramamirtham, C. P., Muthusamy, S., 1986. Estuarine oceanography of the Vembanad Lake Part II: the region between Cochin and Azhikode. Indian J Fish, 33, 218-224.
- Ramasamy, E. V., Jayasooryan, K K., Chandran, M. S. S., Mohan, M., 2017. Total and methyl mercury in the water, sediment, and fishes of Vembanad, a tropical backwater system in India. Environ Monit Assess.189, 130.
- Ramasamy, E. V., Toms, A., Shylesh, C. M. S., Jayasooryan, K. K., Mahesh, M., 2012. Mercury fractionation in the sediments of Vembanad wetland, west coast of India, Environ. Geochem. Health. 34, 575-586.
- Ranjitha, R., Sujatha C. H., 2011, Quantization of specific trace metals in bivalve, Villorita cyprinoides Var Cochinensis in the Cochin Estuary, Indian J Geo Mar Sci. 40, 424-429.
- Rao, V. P., Rao B. R., 1995. Provenance and distribution of clay minerals in the sediments of the western continental shelf and slope of India. Continental Shelf Res. 15, 1757-1791.
- Rao, V. P., Shynu, R., Singh, S. K., Naqvi, S. W. A., Kessarkar, P. M., 2015. Mineralogy and Sr & Nd isotopes of SPM and sediment from the Mandovi and Zuari estuaries: Influence of weathering and anthropogenic contribution. Estuarine Coastal Shelf Sc. 156, 103-115.
- Ratheesh Kumar, C. S., Joseph, M. M., Gireesh Kumar, T. R., Renjith, K. R., Manju, M. N., Chandramohanakumar, N., 2010. Spatial variability and contamination of heavy metals in the inter-tidal systems of a tropical environment. Int J Environ Res. 4, 691-700.
- Reddy, N. P. C., Durga Prasad Rao, N. V. N., Dora, Y. L., 1992, Clay mineralogy of inner shelf sediments off Cochin, west coast of India. Indian J Mar Sci. 21, 152-154.
- Renjith, K. R., Joseph, M. M., Ghosh, P. K., Rahman, H., Ratheeshkumar, C. S., Chandramohanakumar, N., 2013. Biogeochemical facsimile of the organic

matter quality and trophic status of a micro-tidal tropical estuary. Environ Earth Sci. 70, 729-742.

- Renjith, K. R., Sudheesh, V., Shaji, A., George, E., Joseph, M. M., Kumar, C. S. R., Chandramohanakumar, N., 2016. Geochemical distribution, bioavailability and potential toxicity of some trace metals in a complex micro-tidal estuary, southwest India. Toxicol. Environ. Chem. 98, 36-52.
- Revichandran, C., Srinivas, K., Muraleedharan, K. R., Rafeeq, M., Shivaprasad, A., Vijayakumar, K., Jayalakshmy, K. V., 2011. Environmental set-up and tidal propagation in a tropical estuary with dual connection to the sea (SW Coast of India). Environ. Earth Sci. 66, 1031-1042.
- Riley, J. P., Chester, R., 1971. Introduction to Marine Chemistry. Academic Press, London, pp. 465.
- Robin, R. S., Pradipta, R. M., Vishnu Vardhan, K., Abhilash, K. R., Paneerselvam., Charan Kumar, B., Balasubramanian, T., 2012. Assessment of hydrogeochemical characteristic in an urbanized estuary using environmental techniques. Geosciences, 2, 81-92.
- Romankevich, E. A., 1984. Geochemistry of Organic Matter in the Ocean. Springer-Verlag, Berlin, Heidelberg, pp. 334.
- Rullkotter, J., 2002. Organic matter: the driving force for early diagenesis. In: Schulz, H.D., Zabel, M. (Eds.), Marine Geochemistry. Springer, Berlin, Heidelberg, pp. 125-168.
- Salas, P. M., Sujatha, C. H., Kumar, C. S. R., Cheriyan, E., 2017. Heavy metal distribution and contamination status in the sedimentary environment of Cochin estuary. Mar. Pollut. Bull. 119, 191-203.
- Salomons, W., Forstner, U., 1984. Metals in the Hydrocycle. Berlin, Germany: Springer-Verlag. pp. 348.
- Sankaranarayanan, V. N., Panampunnayil, S. U., 1979. Studies on organic carbon, nitrogen & phosphorus in sediments of the Cochin Backwater. Indian J. Mar. Sci. 8, 27-30.

- Sankaranarayanan, V. N., Qasim, S. Z., 1969. Nutrients of the Cochin backwater in relation to environmental characteristics. Mar. Bio. 3, 236-247.
- Sankaranarayanan, V. N., Reddy, C. V. G., 1973. Cu content in the inshore and estuarine waters along the central west coast of India. Curr. Sci. 42, 223-224.
- Sankaranarayanan, V. N., Udayavarma, P., Balachandran, K. K., Pylee, A., Joseph, T., 1986. Estuarine characteristics of the lower reaches of the River Periyar. Indian J. Mar. Sci. 15, 166-170.
- Santschi, P. H., Lenhart, J. J., Honeyman, B. D., 1997. Hetrerogenus processes affecting trace contaminant distribution in estuaries: the role of narural organic matter. Mar. Chem. 58, 99-125.
- Saraladevi, K., Sankaranarayana, V. N., Venugopal, P., 1991. Distribution of nutrients in the Periyar river estuary. Indian J. Mar. Sci. 20, 49-54.
- Saraladevi, K., Venugopal, P., Remani, K. N., Lalitha, S., Unnithan, R. V., 1979. Hydrographic features and water quality of Cochin backwaters in relation to industrial pollution. Ind. J. Mar. Sci. 8, 141-145.
- Saraladevi, K., Venugopal, P., Remani, K. N., Zacharias, D., Unnithan, R. V., 1983. Nutrients in some estuaries of Kerala. Mahasagar, 16, 161-173.
- Sarkar, A., Chakraborty, P., Nath, B. N., 2016. Distribution and nature of sedimentary organic matter in a tropical estuary: An indicator of human intervention on environment. Mar. Pollut. Bull. 102, 176-186.
- Scholkovitz, E. R., Boyle, E., Price, N. B., 1978. The removal of dissolved humic acids and iron during estuarine mixing. Earth Planet. Sci. Lett. 40, 130-136.
- Scholkovitz, E. R., Copland, D., 1981. The coagulation, solubility and adsorption properties of Fe, Mn, Cu, Ni, Cd, Co and humic acids in river water. Geochim. Cosmochim. Acta, 45, 181-189.
- Scholz, F., Neumann, T., 2007. Trace element diagenesis in pyrite-rich sediments of the Achterwasser lagoon, SW Baltic Sea. Mar. Chem. 107, 516–532.

- Seitzinger, S. P., Kroeze, C., Bouwman, A. E., Caraco, N., Den-tener, F., Styles, R.V. 2002. Global patterns of dissolved inorganic and particulate nitrogen inputs to coastal systems, Estuaries, 25, 640-655.
- Seralathan, P., 1979. Studies on texture, mineralogy and geochemistry of the modern deltaic sediments of Cauvery river, India. Ph.D.Thesis, Andhra University, Waltair.
- Shajan, K. P., 2001. Geochemistry of bottom sediments from a river-estuary-shelf mixing zone on the tropical southwest coast of India. Bull. Geol. Surv. Japan. 52, 371-382.
- Sheeba, P., 2000. Distribution of benthic infauna in the Cochin estuary in relation to environmental parameters. Ph.D. Thesis. Cochin University of Science and Technology, Cochin, India. pp. 241.
- Shen, Z., Hongbing, J. I., Weijin, Y., Shuiwang, D., 2003. Composition and flux of nutrients transport to the Changjiang Estuary. J. Geog. Sci. 13, 3-12.
- Shepard, F. P., 1954. Nomenclature based on sand-silt-clay ratios. J. Sediment. Petrol. 24, 151-158.
- Shibu, M. P., 1992. Trace metal speciation in the Cochin estuary. Ph.D. Thesis, Cochin University of Science and Technology, Cochin. pp.18.
- Shibu, M. P., Balchand, A. N., Nambisan, N. P. K., 1990. Trace metal speciation in a tropical estuary: significance of environmental factors. Sci. Total Environ. 97/98, 267-287.
- Shivaprasad, A., Vinita, J., Revichandran, C., Reny, P.D., Deepak, M. P., Muraleedharan, K. R., Naveen Kumar, K. R., 2013. Seasonal stratification and property distributions in a tropical estuary (Cochin estuary, west coast, India). Hydrol. Earth Syst. Sci. 17, 187-199.
- Shoji, D., Balachandran, K. K., Gupta, G. V. M., Madhu, N. V., Shanta, N., 2008. Influence of allochthonous input on autotrophic-heterotrophic switch-over in shallow waters of a tropical estuary (Cochin Estuary), India. Estuar. Coast. Shelf Sci. 78, 551-562.

- Sholkovitz, E. R., 1976. Flocculation of dissolved organic and inorganic matter during the mixing of river and seawater. Geochim. Cosmochim. Acta. 40, 831-845.
- Shynu, R., Rao, V. P., Kessarkar, P. M., Rao, T. G., 2012. Temporal and spatial variability of trace metals in suspended matter of the Mandovi estuary, central west coast of India. Environ. Earth Sci. 65, 725-739.
- Shynu, R., Rao, V. P., Parthiban, G., Balakrishnan, S., Narvekar, Tanuja, R., Kessarkar, P. M., 2013. REE in suspended particulate matter and sediment of the Zuari estuary and adjacent shelf, western India: Influence of mining and estuarine turbidity. Mar. Geol. 346, 326-342.
- Siby, V., 2004. Geochemistry of rare earth elements and trace metals along the western continental shelf of India. Ph.D thesis, Cochin University of Science and Technology, Cochin. pp. 238.
- Simpson, J. H., Vennell, R., Souza, A. J., 2001. The salt fluxes in a tidally-energetic estuary. Estuar. Coast. Shelf Sci. 52, 131-142.
- Sooria, P. M., Jyothibabu, R., Anjusha, A., Vineetha, G., Vinita, J., Lallu, K. R., Paul, M., Jagadeesan, L., 2015. Plankton food web and its seasonal dynamics in a large monsoonal estuary (Cochin backwaters, India)-significance of mesohaline region. Environ Monit Assess.187, 427.
- Spencer, K. L., Dewhurst, R. E., Penna, P., 2006. Potential impacts of water injection dredging on water quality and ecotoxicity in Lime house Basin, River Thames, SE England. Chemosphere 63, 509-521.
- Sprin, R. V., 1965. Rare earth distributions in the marine environment, Ph.D.Thesis ,Massachustts Institution of Technology, Cambridge, Mass, 165.
- Sreedhar, S., Devasya, R. P., Naregundi, K., Young, C. C., Bhagwath, A., 2014. Phosphate solubilizing uranium tolerant bacteria associated with monazite sand of a natural background radiation site in south-west coast of India. Ann Microbiol. 64, 1683- 1689. https://doi.org/10.1007/s13213-014-0812-4.

- Sreekanth, A., Mrudulrag, S. K., Cheriyan, E., Sujatha, C.H., 2015. Trace metal enrichment and organic matter sources in the surface sediments of Arabian Sea along southwest India (Kerala coast). Mar. Pollut. Bull. 101, 938-946.
- Srinivas, K., 2000. Seasonal and Interannual Variability of Sea Level and Associated Surface Meteorological Parameters at Cochin. Ph.D thesis. Cochin University of Science and Technology, pp. 212.
- Srinivas, K., Revichandran, C., Maheswaran, P. A., Mohammed Ashraf, T. T., & Nuncio, M. 2003. Propagation of tides in the Cochin estuarine system, southwest coast of India. Indian J Mar Sci. 32, 14-24.
- Srinivas, K., Revichandran, C., Mohamed Asharaf, T. T., Thottam, T. J., Maheswaran, P. A., Nuncio, M., 2004. Sea level and currents in the upper reaches of the Cochin estuarine system during October 2000. Indian J Mar Sci. 33, 155-163.
- Statham, P. J., 2011. Nutrients in estuaries an overview and the potential impacts of climate change. Sci. Total. Environ. 434, 213-227.
- Stephenes. D. W., Gillespie. D. M., 1976. Phytoplankton production in the Great Salt Lake, Utah, and a laboratory study of algal response to enrichment. Limnol Oceanogr. 21, 74-87.
- Sudhanandh, V. S., Udayakumar, P., Ouseph, P. P., Amaldev, S., Babu, K. N., 2011. Dispersion and accumulation trend of heavy metals in coastal and estuarine sediments and its textural characteristics, a case study in India. J Hum Ecol, 36, 85-90.
- Sujatha, C. H., Benny, N., Raveendran, R., Fanimol, C. L., Samantha, N. K., 2009. Nutrient dynamics in the two lakes of kerala, India. Ind. J. Mar. Sci, 38, 451-456.
- Sujatha, C. H., Nair, S. M., Chacko, J., 1999. Determination and distribution of endosulfan and malathion in an Indian estuary. Water Res. 33,109-114.

- Sundaray, S. K., Nayak, B. B., Lin, S., Bhatta, D., 2011. Geochemical speciation and risk assessment of heavy metals in the river estuarine sediments - A case study: Mahanadi basin, India. J. Hazard. Mater.186, 1837-1846.
- Sundelin, B., Eriksson, A. K. W., 2001. Mobility and bioavailability of trace metals in sulfidic coastal sediments. Environ. Toxicol. Chem, 20, 748-756.
- Szefer, P., 2002. Metals, Metalloids and Radionuclides in the Baltic Sea Ecosystem. Elsevier Science B.V, Amsterdam. pp. 752.
- Tallberg, P., 2000. Silicon and its impacts on phosphorus in eutrophic freshwater lakes. Doctoral dissertation. University of Helsinki.
- Taylor S. R., McLennan S.M. 1985. The Continental Crust, Its composition and evolution. Palo Alto, CA: Blackwell. pp. 1-328.
- Tessier A., Campbell P.G.C., Bisson M., 1979, Sequential extraction procedure for the speciation of particulate trace metals, Anal. Chem., 51, 844-851.
- Thomas, J. B., Bodnar, R. J., Shimizu, N., Chesner, C. A., 2003. Melt inclusions in zircon. Rev Mineral Geochem. 53, 63-87.
- Thomas, J., Joseph, S., Thrivikramji, K. P., 2015. Hydrochemical variations of a tropical mountain river system in a rain shadow region of the southern Western Ghats, Kerala, India. Appl. Geochem.
- Thomson, K. T., 2002. Economic and social issues of biodiversity loss in Cochin backwaters. Technical report: Cochin University of Science and Technology, Cochin, India. pp 51-82.
- Townsend, C. R., Hildrew, A. G., Francis, J., 1983. Community structure in some southern English streams: the influence of physicochemical factors. Freshw. Biol. 13, 521-544.
- Treguer, P., Nelson, D., Van Bennekon, A., DeMaster, D., Leynaert, A., Queguiner, B., 1995. The silica balance in the world ocean: A re-estimate. Science. 268, 375-379.
- Turner, A., Millward, G. E., 2002. Suspended particles: their role in estuarine biogeochemicalcycles. Estuar. Coast Shelf Sci. 55, 857-883.

- Turner, A., Millward, G. E., Morris, A.W., 1991. Particulate metals in five major North Sea estuaries. Estuar Coast Shelf Sci. 32, 325-346.
- Turner, A., Millward, G. E., Tyler, A. O., 1994. The distribution and chemical composition of particles in a macrotidal estuary. Estuar. Coast Shelf Sci. 38, 1-17.
- Turner, A., Millward, G.E., Le Roux, S. M., 2001. Sediment-water partitioning of inorganic mercury in estuaries. Environ. Sci. Technol. 35, 4648-4654.
- Udayakumar, P, Jose, J. J., Krishnan, K. A., Kumar, C. S. R., Manju, M. N., 2014. Heavy metal accumulation in the surficial sediments along south-west coast of India. Environ Earth Sci. 72, 1887-1900.
- Udayavarma, P., Pylee, A., Rama Raju, V. S., 1981. Tidal influence on the seasonal variation in current and salinity around Willington Island. Mahasagar, 4, 225-237.
- Uddin, M. M., Zafar, M., Chowdhury, S. R., 2014. Water, salt and nutrient flux through the lower Meghna River estuary, Bangladesh. Indian J Geo Mar Sci. 43, 280-283.
- Uncles, R. J., Jordan, M. B., 1979. Residual fluxes of water and salt at two stations in the Severn Estuary. Estuarine and Coastal Marine Science, 9, 287-302.
- Unnikrishnan, P., 2000. Phase transitions of trace metals in the aquatic environment of Kuttanad, Kerala, Ph.D Thesis, Cochin University of Science and Technology, Cochin. pp. 70-111.
- Unnithan, R. V., Vijayan, M., Radhakrishnan, E. V., Ramani, K. N., 1977. Incidence of fish mortality from industrial pollution in Cochin backwaters. Indian J. Mar. Sci. 6, 81-83.
- Vale, C., 1990. Temporal variations of particulate metals in the Tagus river estuary. Sci,. Total Environ., 97/98: 137- 154
- Vander Weijden, C. H., Arnoldus, M. J. H. L., Meurs, C. J., 1977. Desorption of metals from suspended material in the Rhine estuary. Neth. J. Sea. Res. 11, 130-145.

- Vineetha, G., Jyothibabu, R., Madhu, N. V., Kusum, K. K., Sooria, P. M., Shivaprasad, A., Reny, P. D., Deepak, M. P., 2015. Tidal Influence on the diel vertical migration pattern of zooplankton in a tropical monsoonal estuary. Wetlands. 35, 597-610.
- Vinita, J., Shivaprasad, A., Manoj, N. T., Revichandran, C., Naveenkumar, K. R., Jineesh, V. K., 2015b. Spatial tidal asymmetry of Cochin estuary, west coast. India. J Coast Conserv. 19, 537-551.
- Vinita, J., Shivaprasad, A., Revichandran, C., Manoj, N. T., Muraleedharan, K. R., Binzy, J., 2015a. Salinity response to seasonal runoff in a complex estuarine system (Cochin Estuary, West Coast of India), J Coastal Res, 31, 869-878.
- Vipindas, P. V., Anas, A., Jasmin, C., Lallu, K. R., Fausia, K. H., Balachandran, K. K., Muraleedharan, K. R., Nair, S., 2015. Bacterial domination over Archaea in ammonia oxidation in a monsoon-driven tropical estuary. Microb. Ecol. 69, 544-553.
- Vollmer, M., Hudec, B., Knauth, H. D., Schwedhelm, E., 1990. Seasonal variations of mineralogical and heavy metal contamination of suspended matter in the river Elbe estuary. In. Michaelis, W. (Ed.), Estuarine Water Quality Management. Springer-Verlag, Berlin. pp. 153-158.
- Walsh, J. J., 1991. Arctic carbon sinks: present and future. Global Biogeochem. Cycles. 3, 399-411.
- Walsh, J.J., 1988. On the nature of continental shelves. Academic Press, San Diego. pp. 520.
- Wang, C., Wang, X., Wang, B., Zhang, C., Shi, X., Zhu, C., 2009. Level and fate of heavy metals in the Changjiang estuary and its adjacent waters. Oceanology. 49, 64-72.
- Wasserman, J. C., Oliveira, F. B. L., Bidarra, M., 1998. Cu and Fe associated with humic acids in sediments of a tropical coastal lagoon. Org. Geochem. 28, 813-822.
- Wen, L., Shiller, A., Santschi, P. H., Gill, G., 1999. Trace element behavior in Gulf of Mexico estuaries. In: Bianchi, T. S., Pennock, J. R., Twilley, R. R. (Eds.),

Biogeochemistry of Gulf of Mexico estuaries: New York, John Wiley, pp. 303-346.

- Whitehouse, U. G., Jeffrey, L. M., Debrecht, J. D., 1960. Differential settling tendencies of clay minerals in saline waters. In: Seventh National Conference on Clays and Clay Minerals. pp. 1-79.
- WHO, 2001. Environmental Health Criteria 224: Arsenic compounds 2nd edition.World Health Organisation, Geneva.
- Wignall, P. B., Myers, K. J., 1988. Interpreting the benthic oxygen levels in mudrocks, a new approach. Geology. 16, 452-455.
- Willen, E., 1991. Planktonic diatoms an ecological review. Algol Stud. 62, 69-106.
- Wood, C.M., Farrell, A.P., Brauner, C.J., 2012. Homeostasis and Toxicology of Nonessential Metals. Elsevier, San Diego, USA.
- Woods, L., 1990. Groundwater Quality in the Nore River Bain, Republic of Ireland. Dissertation. University College, London.
- Xavier, J. K., Joseph, T., Paimpillil, J. S., 1993. Fluxes of Nitrogen in Chaliyar River Estuary, India. International Journal of Ecology & Environmental Sciences, 31, 223-229.
- Zhang, J., Gilbert, D., Gooday, A. J., Levin, L., Naqvi, S. W. A., Middelburg, J. J., Scranton, M., Ekau, W., 2010. Natural and human-induced hypoxia and consequences for coastal areas: synthesis and future development. Biogeosciences. 1443-1467.
- Zhang, L., Liao, Q., Shao, S., Zhang, N., Shen, Q., Liu, C., 2015. Heavy metal pollution, fractionation, and potential ecological risks in sediments from lake Chaohu (Eastern China) and the surrounding rivers, Int. J. Environ. Res. Public Health. 12, 14115-14131.
- Zwolsman, J. J. G., Berger, G. W., Van Eck, G. T. M., 1993. Sediment accumulation rates, historical input, post depositional mobility and retention of major element and trace metals in salt marsh sediment of the Scheldt estuary. SW Netherlands. Mar. Chem. 44, 73-94.

****\$\$****

Dynamics of trace metals, rare earth elements and nutrients in the CE

Appendix 1a:

Calculation of cumulative area for nutient flux measurement using current meter records:

We can define the area by specifying the geographical position, width and depth of each locations (2 upstream and 2 inlet) of Cochin estuary. For that we can divide the area of each location into small different cross sections

The co ordinates of the area of one section is represented below



Calculation of section length:

Section length is calculated using the equation of a straight line,

$$\sqrt{(x1-x2)^2+(y1-y2)^2}$$

Where, $x_1 \& x_2$ are the longitudes and $y_1 \& y_2$ are the latitudes of the area of interest.

Cumulative length =
$$\sum_{a}$$
 sectionlength

Cross Section area = Section length* section depth Cumulative Area = $\sum_{a}^{h} sectionarea$

The coordinates can be specified as degree decimals which we can to convert to kilometeres as 10,000 kilometers are there per 90 degree.

Cumulative area (in meteres)= (Cumulative area in degree decimals) *(111.11)* (1000).

Calculated cumulative area of the regions for computing nutrient fluxes. Varapuzha = $510m^2(max. depth= 3.5m)$

Thanneermukkam = $2200m^2(max. depth= 4.87m)$

(NB:The area of thanneermukkam bund is crossed out during calculation). Cochin Barmouth= 5348.8m²(max. depth= 15m)

Munambam inlet = $203.63m^2(max depth=2m)$

(NB: The drop or rise in water level were not considered during area computation)
Appendix 16:

Water level variations at Cochin inlet during three different seasons (2009-2010).

The selection of spring and neap phase observations are made based on the monthly variations in the water level at the Cochin inlet obtained from Indian Tide Tables. These are again checked during the simultaneous tidal measurements made at this location during the entire observational period, which is furnished below for three different seasons. (a) Post-monsoon spring; (b) Post-monsoon neap; (c) Pre-monsoon spring; (d) Pre-monsoon neap; (e) Peak summer monsoon spring; (f) Peak summer monsoon neap.







Appendix 2a

	Water	quality pa	rameters	s (Meta	l fraction	ation stu	dies - Oct	ober 201	1)
Stn	рН	Salinity	DO (ml/l)	BOD ml/l	NO₃ (μM)	NO₂ (μM)	SiO₄ (μM)	ΡΟ₄ (μΜ)	NH₄ (μM)
S_1	7.3	4.68	3.17	1.64	11.77	0.83	32.87	0.47	17.91
S_2	7.36	5.44	2.96	2.24	11.12	0.71	29.62	0.56	4.96
S_3	7.59	9.13	3.31	2.07	3.53	0.67	20.85	1.01	12.37
S_4	7.81	9.84	2.98	1.77	6.89	0.70	17.93	1.53	14.17
S_5	7.81	10.17	3.11	2.22	5.82	0.56	15.44	0.98	15.02
S_6	8.28	25.02	3.17	2.65	1.13	0.23	2.08	0.76	12.24
S ₇	7.65	10.09	3.55	1.79	3.88	0.73	16.29	1.40	15.84
S_8	7.55	11.66	2.92	1.97	3.59	0.45	13.96	1.00	12.63
S_9	8.41	28.36	4.03	3.16	2.41	0.23	1.95	0.36	10.29
S_{10}	8.43	23.42	3.86	3.24	0.69	0.11	1.11	0.61	11.09
S_{11}	8.33	26.28	4.34	3.57	3.03	0.01	0.14	0.29	10.40
S ₁₂	8.41	30.58	4.14	3.48	1.86	0.05	6.35	0.43	6.06

Appendix

Appendix 2b





X-Ray Diffractograms of the estuarine sediments S1 & S2, during October 2011.





X-Ray Diffractograms of the estuarine sediments S3 & S4 during October 2011.





X-Ray Diffractograms of the estuarine sediments S_5 & S_6 , during October 2011.







Appendix





X-Ray Diffractograms of the estuarine sediments S_9 & S_{10} , during October 2011.





X-Ray Diffractograms of the estuarine sediments $S_{11}\ \&\ S_{12},$ during October 2011.



TGd																				1.00	014
Gd4																			1.00	0.77	TAT 2
Gd3																		1.00	0.26	0.75	STX
Gd2																	1.00	0.76	0.66	0.81	
Gd1																1.00	-0.02	0.32	0.04	0.21	
TEu															1.00	0.11	0.72	0.49	0.87	0.91	
Eu4														1.00	0.51	-0.15	0.24	-0.30	0.73	0.19	
Eu3													1.00	-0.45	0.41	0.38	0.66	0.95	0.15	0.69	
Eu2												1.00	0.69	0.12	0.64	-0.07	0.97	0.74	0.51	0.73	
Eu1											1.00	-0.09	0.38	-0.16	0.09	1.00	-0.04	0.31	0.02	0.19	
TSm										1.00	0.17	0.66	0.58	0.32	0.92	0.19	0.78	0.67	0.85	0.98	> 0.05
Sm4									1.00	0.80	-0.01	0.21	0.11	0.43	0.80	0.00	0.34	0.16	0.79	0.74	alpha
Sm3								1.00	0.04	0.61	0.30	0.75	0.82	-0.09	0.44	0.32	0.80	0.92	0.33	0.66	level
Sm2							1.00	0.74	0.34	0.74	-0.17	0.94	0.54	0.33	0.69	-0.14	0.99	0.66	0.69	0.75	icance
Sm1						1.00	-0.12	0.34	0.03	0.22	1.00	-0.06	0.39	-0.12	0.13	1.00	0.00	0.34	0.08	0.23	ı signij
oc					1.00	-0.01	0.29	0.17	0.33	0.44	-0.01	0.46	0.57	-0.27	0.39	-0.04	0.37	0.48	0.16	0.52	with c
SPM				1.00	0.46	-0.49	-0.24	-0.44	-0.05	-0.28	-0.48	-0.13	-0.07	-0.36	-0.28	-0.50	-0.25	-0.22	-0.31	-0.26	from 0
K%			1.00	0.17	-0.37	-0.64	0.04	-0.14	-0.11	-0.22	-0.65	-0.09	-0.36	-0.05	-0.29	-0.63	-0.07	-0.23	-0.14	-0.25	ferent
1%		1.00	0.56	0.29	-0.04	-0.15	-0.41	-0.22	-0.05	-0.21	-0.14	-0.46	-0.24	-0.27	-0.34	-0.15	-0.42	-0.13	-0.31	-0.25	are dif
S%	1.00	-0.79	-0.95	-0.23	0.29	0.53	0.12	0.19	0.10	0.24	0.53	0.24	0.36	0.13	0.34	0.52	0.20	0.22	0.22	0.28	pold o
Variables	S%	I%	K%	SPM	OC	Sml	Sm2	Sm3	Sm4	TSm	Eu1	Eu2	Eu3	Eu4	TEu	Gd1	Gd2	Gd3	Gd4	TGd	Values in

bles	S%	1%	K%	SPM	OC	Tb1	Tb2	Tb 3	Tb4	TTb	Dy1	Dy2	Dy3	Dy4	TDy	Hol	Ho2	Ho3	Ho4	THo
	1.00																			
	-0.79	1.00																		
Ŷ	-0.95	0.56	1.00																	
X	-0.23	0.29	0.17	1.00																
٢)	0.29	-0.04	-0.37	0.46	1.00															
-	0.52	-0.15	-0.62	-0.51	-0.05	1.00														
5	0.23	-0.42	-0.10	-0.23	0.39	0.01	1.00													
3	0.42	-0.32	-0.40	-0.04	0.59	0.31	0.68	1.00												
4	0.26	-0.29	-0.21	-0.26	0.26	0.00	0.61	0.25	1.00											
٩	0.31	-0.27	-0.29	-0.19	0.58	0.20	0.79	0.78	0.74	1.00										
,1	0.52	-0.15	-0.62	-0.52	-0.06	1.00	0.01	0.31	0.00	0.20	1.00									
Ç	0.22	-0.41	-0.09	-0.16	0.45	-0.04	0.99	0.68	0.56	0.78	-0.04	1.00								
ŝ	0.30	-0.17	-0.32	-0.25	0.51	0.41	0.74	0.91	0.22	0.74	0.41	0.74	1.00							
4	0.21	-0.28	-0.14	-0.11	0.21	-0.09	0.49	0.13	0.97	0.64	-0.09	0.46	0.04	1.00						
ý	0.32	-0.25	-0.31	-0.14	0.62	0.21	0.76	0.81	0.69	1.00	0.21	0.76	0.75	09.0	1.00					
-	0.52	-0.15	-0.62	-0.54	-0.05	0.99	0.06	0.32	0.08	0.24	0.99	0.00	0.43	-0.01	0.25	1.00				
5	0.06	-0.34	0.09	0.00	0.45	-0.32	0.94	0.57	0.54	0.69	-0.32	0.96	0.59	0.47	0.67	-0.28	1.00			
3	0.37	-0.16	-0.41	-0.23	0.51	0.47	0.63	0.92	0.14	0.70	0.47	0.63	0.98	-0.03	0.73	0.47	0.47	1.00		
4	0.29	-0.36	-0.21	-0.20	0.16	0.03	0.43	0.10	0.93	0.61	0.03	0.38	0.00	0.97	0.57	0.10	0.36	-0.07	1.00	
lo	0.31	-0.23	-0.30	-0.15	0.61	0.24	0.74	0.82	0.64	0.99	0.23	0.74	0.77	0.55	1.00	0.26	0.64	0.75	0.53	1.00
ues in	pold	are di	Herent	from (D with	a sion	ficanc	01000	alnha	0.0 < 10.0	5						TSTX	4T201	4	

TYb																				1.00	2014
Yb4																			1.00	0.47	STAT :
Yb3																		1.00	-0.15	0.80	STX
Yb2																	1.00	0.53	0.37	0.75	
Ybl																1.00	0.04	0.31	0.08	0.33	
Tm															1.00	0.48	0.72	0.75	0.50	0.98	
Dm4														1.00	0.64	0.23	0.52	0.23	09.0	0.58	
Cm3 7													1.00	0.20	0.87	0.55	0.61	0.82	0.21	0.86	
m2 T												1.00	0.61	0.50	0.71	0.01	1.00	0.54	0.37	0.75	
ml											1.00	00.0	0.55	0.21	0.47	1.00	0.03	0.30	3.08	0.32	
Er										00.	.29	.76	.82	.62	.98	.29	.75 (.78	.50 (00.	0.05
4 T									00.	.60	.08	50 (.13 (.98	58 (.10	.52 (.21 (57 (.54 1	oha >
3 Eı								00	07 1	78 0	52 0	58 0	97 0	14 0	83 0	52 0	57 0	81 0	16 0	83 0	vel al _l
Er Er							00	42 1	38 0.	63 0	32 0.	94 0.	43 0.	33 0.	52 0.	32 0.	92 0	48 0.	24 0	62 0.	nce le
Ē						9	1.	52 0.	l6 0.	32 0.	.0- 0)5 0 .	57 0.	.0 0.	51 0.	.0-	.0 0	32 0.	0.0	35 0.	gnifica
Er1					0	4 1.(6 -0.2	8 0.5	3 0.1	0.0	2.0 5	8 0.(0.5	0.0	2 0.5	6 0.9	6 0.(6 0.3	2 0.1	1 0.3	h a sig
OC					1.0	-0.0	0.4	0.4	0.2	0.6	0.0-	0.3	0.5	0.1	0.5	-0.0	0.3	0.8	-0.2	0.6	0 witi
SPM				1.00	0.46	-0.53	0.01	-0.22	-0.27	-0.18	-0.53	-0.24	-0.22	-0.39	-0.31	-0.53	-0.28	0.08	-0.35	-0.19	from
K%			1.00	0.17	-0.37	-0.63	0.08	-0.46	-0.25	-0.32	-0.61	-0.07	-0.39	-0.29	-0.41	-0.62	-0.08	-0.51	0.15	-0.34	ferent
1%		1.00	0.56	0.29	-0.04	-0.15	-0.34	-0.23	-0.33	-0.23	-0.15	-0.41	-0.10	-0.36	-0.27	-0.15	-0.40	-0.11	-0.23	-0.25	ure di <u>f</u>
S%	1.00	-0.79	-0.95	-0.23	0.29	0.52	0.07	0.43	0.31	0.33	0.51	0.20	0.32	0.35	0.41	0.51	0.21	0.42	-0.03	0.35	bold ϵ
/ariables	S%	1%	K%	SPM	OC	Er1	Er2	Er3	Er4	TEr	Tm1	Tm2	Tm3	Tm4	TTm	Ybl	Yb2	Yb3	Yb4	TYb	/alues in

'ariables	S%	I%	K%	SPM	8	Lu1	Lu2	Lu3	Lu4	TLu	Th1	Th2	Th3	Th4	TTh	10	U2	U3	U4	ΤU
S%	1.00																			
I%	-0.79	1.00																		
K%	-0.95	0.56	1.00																	
SPM	-0.23	0.29	0.17	1.00																
OC	0.29	-0.04	-0.37	0.46	1.00															
Lu1	0.51	-0.15	-0.61	-0.53	-0.07	1.00														
Lu2	0.20	-0.37	-0.09	-0.37	0.31	0.10	1.00													
Lu3	0.30	-0.14	-0.33	-0.05	0.57	0.30	0.65	1.00												
Lu4	0.26	-0.38	-0.16	-0.28	0.18	0.05	0.44	-0.03	1.00											
TLu	0.33	-0.25	-0.32	-0.24	0.58	0.33	0.78	0.81	0.48	1.00										
Th1	-0.36	0.31	0.33	-0.14	-0.01	0.06	0.16	0.36	-0.52	0.05	1.00									
Th2	0.07	-0.03	-0.07	-0.33	0.20	0.43	0.73	0.50	0.44	0.64	0.27	1.00								
Th3	0.23	-0.53	-0.05	-0.42	-0.14	0.16	0.74	0.41	0.11	0.38	0.08	0.35	1.00							
Th4	0.00	-0.11	0.06	-0.40	0.05	0.12	0.74	0.36	0.59	0.58	0.16	0.86	0.32	1.00						
TTh	0.02	-0.15	0.05	-0.43	0.05	0.14	0.78	0.39	0.58	0.60	0.17	0.87	0.39	1.00	1.00					
IJ	0.11	-0.15	-0.07	-0.68	-0.29	0.45	0.53	0.16	0.51	0.37	0.11	0.78	0.33	0.87	0.87	1.00				
U2	0.04	-0.19	0.04	-0.70	-0.35	0.40	0.64	0.20	0.45	0.38	0.16	0.78	0.54	0.85	0.87	96.0	1.00			
U3	0.25	-0.24	-0.22	-0.65	-0.07	0.59	0.76	0.59	0.28	0.65	0.28	0.84	0.61	0.78	0.81	0.84	0.89	1.00		
U4	0.04	0.01	-0.06	0.23	0.73	-0.15	0.43	0.64	0.35	0.79	0.03	0.29	-0.07	0.35	0.34	-0.03	-0.07	0.15	1.00	
TU	0.22	-0.18	-0.20	-0.49	0.20	0.47	0.81	0.72	0.42	0.85	0.24	0.86	0.48	0.83	0.85	0.76	0.77	0.93	0.49	1.00
'alues in	i pold i	are dif	ferent	from 0	with a	i sienii	ficance	a level	alpha	> 0.0	10							S'IX	TAT 21	014